

Strategic Landscape and Ecology Management Plan (LEMP)

for

Langarth Garden Village,

Cornwall

12 October 2021

Prepared for: Arcadis on behalf of Cornwall Council

Prepared by: Cornwall Environmental Consultants (CEC Ltd) Five Acres Allet Truro TR4 9DJ

Ref: CEC3264-01h





five acres, allet, truro, cornwall, tr4 9dj t: 01872 245510 f: 01872 262071 e: enquiries@cecenvironment.co.uk www.cecenvironment.co.uk

Revision	Prepared by	Reviewed by	Approved by
DRAFT	Jenny Stuart MSc, CEnv, MCIEEM	Natasha Collings-Costello MEnv Sci, CEnv, CSci, MCIEEM	Natasha Collings-Costello MEnv Sci, CEnv, CSci, MCIEEM

Document Control

Disclosure

The information, opinion and advice which we have prepared and provided is true and has been prepared and provided in accordance with the CIEEM's Code of Professional Conduct and the British Standard for Biodiversity – Code of Practice for Planning and Development (2013). We confirm that the opinions expressed are our true and professional bona fide opinions.

ISO Accreditations





cornwall environmental consultants ltd

five acres, allet, truro, cornwall, tr4 9dj t: 01872 245510 f: 01872 262071 e: enquiries@cecenvironment.co.uk www.cecenvironment.co.uk

I

Contents

١.	Introduction	3
1.1.	Background	3
1.2.	Scope and Planning	4
١.3.	Aims and Objectives	5
I.4.	Sources of Information	5
1.5.	Management Responsibility	6
2.	Proposed Development Plan	8
2.1.	Description	8
2.2.	Phasing	8
3.	Ecological Mitigation and Enhancement	П
3.1.	Pre-construction Recommendations	П
3.2.	Construction Phase Recommendations	П
3.3.	Operational Phase Mitigation Measures & Enhancements	14
3.4.	Monitoring	16
4.	Green Infrastructure	18
4.1.	Open Green Space	18
4.2.	SuDS and Drainage	19
4.3.	Built development plots	19
4.4.	Access	21
4.5.	Lighting	21
5.	Ecological Trends and Constraints to Management	22
5.1.	Invasive Non-Native Species	22
5.2.	Pests and Diseases	22
5.3.	Natural Succession	22
5.4.	Protected Species	23
6.	Management objectives and principles for Landscape and Ecological Features	24
6.1.	Management objectives	24
6.2.	Headline maintenance and management operations for strategic GI	24
6.3.	Headline maintenance and management requirements for Reserved Matters applications	25
7.	General Maintenance Operations	26
7.1.	Strategic Green Infrastructure Management	26
7.2.	Amenity Grass (Seeded)	27
7.3.	New Areas of Wildflower Meadow/ Wetland Wildflowers (Seeded)	27
7.4.	Existing Hedges in Woody Condition	29
7.5.	Existing Hedges to be Strengthened	29
7.6.	New Cornish Hedge	30
7.7.	Native Shrub and Woodland Edge Planting	32
7.8.	Woodland Planting	33
7.9.	Tree Planting (Larger Stock)	35



CEC3264-01h Langarth Garden Village

Strategic LEMP

Report DRAFT- 12 October 2021

7.10.	Amenity and Ornamental Planting	36
7.11.	Permanent Ponds (in SuDS Basins)	37
7.12.	Bat, beebird boxes/ bricks	39
8.	Work Programme, Management Responsibilities and Review	40
8.1.	Indicative Ten-Year Work Programme	40
8.2.	Reporting and Review	41

List of Figures

Figure 1 Aerial image of Langarth Garden Village Site	3
Figure 2: Phases of the development	9
Figure 3 Design Code parameters on garden provision within different residen	tial grains
	19
Figure 4 Design Code parameters on boundary treatments within different	residential
grains	

List of Tables

Table 1: Phasing of the c	development
---------------------------	-------------



I. Introduction

I.I. Background

This report, prepared by Cornwall Environmental Consultants Ltd (CEC), was commissioned in April 2021 by Arcadis, on behalf of Cornwall Council, to provide a strategic Landscape and Ecological Management Plan (LEMP) for the Langarth Garden Village hybrid planning application. The scope of this document, which will be updated as separate elements of the project come forward for reserved matters applications, is set out in detail in section 1.2.

Section 1.3 sets out the aims and objectives of the LEMP, while key documents used in preparation of the LEMP are listed in section 1.4. Section 1.5 explains the ongoing management responsibilities for the site. Section 2 summarises the Langarth Garden Village and sets out the phased delivery of the project. Section 3 details the ecological mitigation and enhancement to be delivered. Section 4 outlines the key green infrastructure included within the project, with an emphasis on the strategic green infrastructure. Section 5 discusses ecological trends, and constraints to future management of the Site. Sections 6 and 7 cover the objectives and principles for management, and the operations that will need to be employed to achieve the objectives. Section 8 sets out an indicative 10-year work programme, explains the responsibilities for management, and review.

Appendix A comprises a Framework Biodiversity Net Gain Strategy, which summarises the measures that have been, or need to be, undertaken to avoid and minimise impacts to biodiversity, and sets out the preconstruction and post-construction biodiversity value of the Site in terms of Net Gain.



Figure I Aerial image of Langarth Garden Village Site



Landscape and Ecological Management Plans (LEMPs) are typically required for most major development projects, including projects like Langarth Garden Village, for several reasons:

- To provide the applicant with assurance that as individual projects within the Garden Village are brought forward to Reserved Matters applications and are built out, that the applicant's original vision for a high-quality Garden Village, with Green Infrastructure at its heart is realised.
- To provide the Local Planning Authority, consultees, and key stakeholders with assurance that the scheme can and will be delivered as set out within the outline application, and that it will be managed appropriately in the future.
- To provide evidence for the Building with Nature assessment of the project.
- To bring together key material from separate documents prepared as part of the outline planning application in one place, to provide a clear statement of the actions that will need to be undertaken as part of detailed design, Reserved Matters application, construction and occupation/ operation of the project in the future.

This report follows recommendations made in BS 42020:2013 Biodiversity: Code of Practice for Planning and Development BS 8683:2021 Process for designing and implementing Biodiversity Net Gain.

I.2. Scope and Planning

The Langarth Garden Village scheme is a complex project, which will be built out in several phases over the next 20 -30 years (more detail on the phasing is set out within section 2). A hybrid planning application Ref PA20/09631 (comprising an outline application for the Langarth Garden Village, and a full application for construction of the Northern Access Road running through the Garden Village) is currently being considered by Cornwall Council.

The scope of this strategic LEMP is to set out the parameters for the implementation and management of the ecological mitigation and enhancement measures that are required to deliver the high quality, connected green infrastructure across the site, in a manner which will achieve the particular aim of the project to deliver at least 20% Biodiversity Net Gain within the wider site, and meet other legislative and policy requirements for landscape and ecology.

This first issue of the strategic LEMP is being written in parallel with the LEMP for the Northern Access Road, and the intention is that individual Reserved Matters Applications will draw on this strategic LEMP in preparing their own detailed LEMPs for each phase of the project.

As individual phases are progressed, a project or phase specific LEMP will need to be prepared, drawing on the information within this strategic LEMP, to set out the measures that each project will undertake to implement and manage provision of ecological mitigation and enhancement measures.

This LEMP should be read alongside the relevant ES chapters, relevant parameter plans, the Design and Access Statement (specifically Appendix 2, the Landscape Strategy), and the Design Code.



I.3. Aims and Objectives

The overall aim of the LEMP is to ensure that ecological mitigation requirements are fulfilled, the calculated Biodiversity Net Gain measurement is reached, and the site is managed appropriately to maintain those values in perpetuity.

The National Planning Policy Framework and Cornwall Council Local Planning Policy, both require schemes to achieve a net gain for biodiversity, although neither specify a means for achieving or measuring net gain. The UK Government announced in the 2019 Spring Statement that Biodiversity Net Gain will become mandatory in England. Whilst the timetable for implementation of this requirement nationally is not yet known, Cornwall Council have applied a requirement for major development schemes to achieve a minimum 10% Biodiversity Net Gain, from March 2020.

Biodiversity Net Gain has been measured in accordance with current guidance, and using the DEFRA Biodiversity Metric 2.0, as required by Cornwall Council. The metric calculated in November 2020 to accompany the Environmental Statement, a gain in 28.77% habitat area units (+202.74 habitat units) and a gain of 12.54% in hedgerow units (+56.83 hedgerow units). This metric calculation is currently being updated to account for revisions to the application design and will be amended in due course. The habitats will need to be monitored during the operational phase, to ensure that these habitat units are provided and the habitats are managed appropriately.

Key Performance Indicators (KPI)

The client has set a number of KPIs to measure the environmental performance of the project. The following KPIs (forming part of the Benefits Realisation Strategy for LGV) with relevance to biodiversity and landscape are currently proposed, subject to final approval by the Strategic Board:

- Biodiversity (with a metric linked to achieving the required Biodiversity Net Gain, and a metric to allow for resident participation in species counts on site)
- Trees Established (with metrics linked to canopy cover, woodland area and number of trees within developed areas)
- Cornish hedges (with metrics linked to length and type of Cornish hedge i.e. tree or grass covered for example, and condition of Cornish hedges and their buffers)
- Green and Blue Infrastructure (with metrics linked to BwN accreditations and quality /usability of green infrastructure as perceived by residents)

I.4. Sources of Information

This strategic LEMP has been prepared by CEC, whose ecologists have been closely involved in the planning and design of the Langarth Garden Village scheme.

This report draws on several sources of information on mitigation measures and parameters for the soft landscape design, notably:

• The Environmental Statement LAN_02.2-ARC-XX-XX-RP-Z-0002 Submitted November 2020



- The Design Code 2020_11_04 1665-ACM-XX-XX-RP-TC-00001-V3 DAS
- The Design and Access Statement (specifically the Landscape Strategy)
- Parameter plans
 - LAN_02.1_AHR-MP-ZZ-SC-A-92-007 Phasing Schedule (Revision P09);
 - LAN_02.1_AHR-MP-ZZ-SC-A-92-010 Land Use (Revision P12);
 - LAN_02.1-AHR-MP-ZZ-DR-A-92-002 Proposed Land Use (Revision P15);
 - LAN_02.1-AHR-MP-ZZ-DR-A-92-009 Proposed Landscape (Revision P14);
 - o LAN_02.1-AHR-MP-ZZ-DR-A-93-005 Phasing (Revision P13);
 - LAN_02.1-AHR-MP-ZZ-DR-A-93-008 SuDS Strategy (Revision P18);
 - o LAN_02.1-AHR-MP-ZZ-DR-A-93-010 Density Parameter Plan (P14);
 - o LAN_02.1-AHR-MP-ZZ-DR-A-93-011 Building Heights Parameter Plan (Revision P17);
 - LAN_02.1-AHR-MP-ZZ-DR-A-93-012 Movement and Access Parameter Plan (Revision P14); and
 - o LAN_02.I-AHR-MP-ZZ-DR-A-93-026 Wider Movement Strategy (Revision P05);
 - o LAN_02.1-AHR-MP-ZZ-DR-A-92-012 Illustrative masterplan (Revision P03).
- Outline CEMP Appended to the Environmental Statement LAN_02.2-ARC-XX-XX-RP-Z-0002
- Preliminary Arboricultural Method Statement Appendix 13.7 of the Environmental Statement LAN_02.2-ARC-XX-XX-RP-Z-0002

As the phased proposals for Langarth Garden Village come forwards the most up the date revisions of the documents above should be used.

As most of the planning application is an outline application, there is limited design information on ecological mitigation and planting design, and therefore detailed information on management requirements. The intention of this strategic LEMP is to set out the parameters for ongoing management and monitoring, and that individual LEMPs will be produced as part of each Reserved Matters application, which will set out the detailed management and monitoring measures appropriate to the scheme.

I.5. Management Responsibility

It is currently proposed that the unadopted green and blue infrastructure will be managed through a Langarth Stewardship Organisation (LSO) envisaged to be in place from 2023 onwards, subject to approval of a Full Business Case and Business Plan in 2022. Until such time that the LSO is in place, responsibility for the oversight of implementation of LEMP requirements will sit with Cornwall Council in its role as Applicant for the LGV, and ensuring through its role as Planning Authority that individual RMAs do meet the required standards.



The LSO, once in place, would contract with appropriate companies to deliver management and maintenance responsibilities for unadopted green and blue infrastructure to the required standards, with the strategic LEMP and individual LEMPs for RMAs forming part of the specification. This is envisaged to include the Strategic Green Infrastructure not associated with adopted highways, above ground SUDS / Blue Infrastructure not associated with adopted highways, the GBI provisions within development parcels and the SANG. Mechanisms and provisions are required to define a process how developers of individual RMAs will hand over unadopted assets into the responsibility of the LSO – these should be defined in the S106 or in contractual provisions with developers. Upon completion of the scheme a single LEMP will be produced to pull together the strategic LEMP with the individual phase LEMPs. This will allow management of the site to be considered on a whole-scheme basis.

Adopted Green and Blue Infrastructure, including that associated with the NAR, would be maintained by Cornwall Council through its normal processes for adopted assets. The strategic LEMP and individual RMA LEMPs that cover adopted assets should form key documents to be provided to Cornwall Council Infrastructure Adoptions Team to take account of when considering Technical Approvals, maintenance specifications and commuted sum calculations for adopted elements. The LSO, once in place, will work with the Infrastructure Adoptions Team in future to ensure consistent management and maintenance standards between adopted and unadopted elements.

The LSO is proposed to be financed through a number of sources, including S106 contributions, resident charges and income generated on site, ultimately becoming self-funding in time. Cornwall Council would likely fund any shortfalls during early phases, and a cost model for the LSO is under development to be able to assess financial requirements in more detail.

With the LSO Business Case and Cost Model still under development, this Management Section will have to be updated once the Full Business Case and Business Plan including its cost model have been approved towards the end of 2022, so that management and monitoring required under this LEMP are fully integrated into the proposed management structure for the wider site and its wider assets.



2. Proposed Development Plan

2.1. Description

The scheme is set out in a hybrid planning application (Ref PA20/09631), which includes:

- Full planning application for construction of the Northern Access Road and associated junction arrangements, infrastructure, earthworks and retaining and boundary features
- Outline planning application with all matters reserved to create a mixed use, landscape-led community comprising a phased development of up to 3550 dwellings plus 200 extra care units and 50 units of student/ health worker accommodation, including affordable housing; along with related facilities including retail, office space, leisure facilities, schools and green infrastructure, as well as internal roads, paths and cycle lanes, services, etc.

2.2. Phasing

The construction of Langarth Garden Village will be phased over approximately 25 years. Table I summarises each phase and corresponding timescales. Figure 2 on the following page illustrates the locations of each phase. Individual phases of development will be completed as a series of separate reserved matters applications.

Phase	Summary	Timescales
I	Northern Access Road, Strategic Green Infrastructure at Governs (part), Development plots (residential and school)	2020-2025
2	Development plots (residential, extra care and Park & Ride Extension) and further Strategic Green Infrastructure at Governs	2025-2030
3	Development plots (residential) and Strategic Green Infrastructure west of Governs	2030-2035
4	Development plots (residential & mixed use)	2035-2040
5	Development plots (residential & school)	2040-2045

Table 1: Phasing of the development



Figure 2: Phases of the development



CEC3264-01h Langarth Garden Village Strategic LEMP Report DRAFT– 12 October 2021



Information Classification: CONTROLLED

Proposed Phasing хакотонист 5545 2019.00106.000 1:5300 3 мбр лист 15300 LAN 02.1-AHR-MP-ZZ-DR-A-83-005 P13 D5 CA The provide provide the result of the second structure dependence on the second structure of the second structure of the second structure of the second structure of the second structure and the second structure of the seco

1.97.3031 SW 9035 3.17.4038 8W 903

3. Ecological Mitigation and Enhancement

3.1. Pre-construction Recommendations

Ecological surveys will need to be updated periodically, and for each phase of development or individual Reserved Matters application, as these are progressed.

In particular it should be noted that it was not possible to carry out a bat survey of every potential bat roost feature (buildings and trees) within the Site, due to a combination of landowner constraints and the Covid-19 pandemic. As the development will be phased over a period of many years, existing survey information will need updating prior to future applications. Therefore, each Reserved Matters application will need to include a visual assessment of all potential bat roost features affected by the proposal, and, where required, further surveys (such as bat emergence and remote monitoring surveys) will need to be completed to inform the application.

An ecologist will need to review each part of the project to be progressed as a Reserved Matters application and take into account the ecological surveys undertaken to date, in order to advise on the precise scope and methodology required for any updated or additional surveys.

A number of invasive non-native plant species are present within the site, including Japanese knotweed, montbretia, variegated yellow archangel, rhododendron, Himalayan balsam, cotoneaster sp., three-cornered garlic and parrot's feather. A detailed Japanese knotweed survey and management plan has been prepared for the Northern Access Road.

As the bulk of the development will be progressed in phases over a long time period, a detailed survey and management plan for the control and eradication of non-native invasive species will be produced. Implementation of the plan will commence as early as possible within the implementation of each phase of the project. Where possible whole site treatment will be commenced, as this should reduce potential future delays to construction, and limit the further spread of the species.

3.2. Construction Phase Recommendations

The outline CEMP prepared for the scheme summarises the key construction phase mitigation measures as summarised below that will need to be incorporated into the final CEMP for each separate project as they are brought forward (Section 6.7.1 of the outline CEMP).

General measures

- All personnel involved in the construction works will be briefed on the ecological mitigation strategy and relevant wildlife legislation through tool-box talks, etc;
- Risk of pollution incidents will be reduced by adhering to current Environment Agency Guidelines;
- Zones for the piling of soil or storage of materials associated with the development will be clearly defined;
- Fuel, oil and other chemicals will be stored in appropriate impervious containers. Leaking and empty



containers will be removed from the Site immediately;

- Any trenches/pits created during the construction process will either be covered over at night or will include a means of escape to prevent animals becoming trapped in them; and
- Any new or temporary lighting scheme implemented on the Site will follow Bat Conservation Trust advice on wildlife sensitive lighting.

Non-native invasive species (NNIS)

Notwithstanding the advice above to start implementing control and eradication measures for invasives species from early in the project, it is likely that, at least in the early stages, there may still be viable stands of invasive plant species present within individual construction sites. Several invasive plant species are listed on Schedule 9 of the Wildlife and Countryside Act 1981, and it is an offence to cause these species to spread outside the site.

Suitable working practices included within a method statement, written by an appropriate specialist, will be followed to ensure that the NNISs are not spread. The Contractor will engage a competent person for managing invasive plants during site setup, clearance and construction. This will be compliant to best practice guidance such as from Construction Industry Research and Information Association (CIRIA).

Special Area for Conservation (SAC)

A drainage scheme will be designed and installed, for each phase of development (and will be in place from an early stage of construction) to provide controlled drainage for surface water run-off and prevent pollution during construction. This will prevent any possible pollution incidents to the streams which could then impact on the downstream Fal and Helford Special Area for Conservation (SAC). SACs are legally protected under the Habitats Regulations 2017, and projects such as built development must demonstrate that they will not impact on the SAC, or the interest features for which it is designated.

Hedges

With the exception of the 5.4km (c. 18%) of hedgerows identified for removal in the Environmental Statement and Biodiversity Net Gain Calculations, over 80% of existing hedges will be retained. Hedges are a Habitat of Principal Importance, and as such, development projects should ensure that there is a net gain in hedge habitat.

Retained hedges will be enhanced with buffer planting and included within green corridors, and new hedges will also be created as part of the Garden Village proposals, as identified within the landscape parameter plan.

Badgers

There are numerous badger setts on site, and a number of these (mostly outlier setts) will be impacted by the development. Badger setts are protected against damage and destruction under the Protection of Badgers Act 1996. Any badger setts that need to be closed must be closed under appropriate licence from Natural England if potential disturbance to badger setts is likely to be significant, take advice from NE



regarding need for disturbance licensing (ES chap 11, table 11.6). The main sett at the western end of the Maiden Green area requires closure and will need an artificial sett to be built (at the location already determined) before the licence to close this sett can be obtained. The need for badger licences for impacts to other setts will need to be determined by an ecologist as each separate Reserved Matters application is progressed.

Bats

A number of bat roosts, and possible bat roosts are present within the Site. Bats and their roosts are protected under the Habitats Regulations 2017. Any loss or disturbance to bat roosts can only be undertaken once EPS licence(s) have been obtained from Natural England. Mitigation will depend on the impact and is likely to include the provision of alternative roosts and timing the works carefully to avoid and minimise disturbance impacts (ES chap 11, table 11.6). The need for EPS licences for destruction and/ or damage of bat roosts will need to be determined by an ecologist as each separate Reserved Matters application is progressed.

Woodlark and wintering birds

Woodlark use some of the arable fields on site in the winter months for foraging, and other species such as snipe overwinter in areas of marshy grassland. Woodlark is a species of conservation concern at international, national and regional levels.

- An on-site mitigation strategy for woodlark is in place, and this will provide suitable habitat on-site for as long as possible through the phased development of the site, and a small area of permanent woodlark habitat will also be provided (see woodlark on-site mitigation and compensation strategy).
- Suitable off-site arable habitat is to be identified and used for compensation of loss of arable habitat
 required by woodlark. This will be secured through a Section 106 agreement, with the commitment to
 prepare and implement an off-site compensation strategy for wintering woodlark habitat. This
 mitigation will also be suitable for the loss of wintering habitat for other wintering birds, with the
 exception of wintering snipe; and
- Areas of marshy grassland along the valley floor within Langarth Garden Village should be fenced off to restrict access and allow snipe to forage and rest undisturbed.

Breeding birds

As would be expected on a site of this size and type, there are numerous birds that breed within the Site. The active nests of all species of birds are protected against damage and destruction under the Wildlife and Countryside Act 1981.

Clearance of vegetation and demolition of buildings should (in the case of the latter, where this is not impacted by advice for minimising impacts to bats) be undertaken between the months of October and February to avoid the risk of disturbing, damaging or destroying active bird's nests. As well as woody vegetation, this will apply to clearance and topsoil stripping in fields that have potential to support ground-nesting skylark.



Where this timing is not possible, ecologists will need to check the features for birds' nests ahead of clearance, and if any active nests these must be left undisturbed, with a suitable buffer, until the young have fledged. The likelihood of finding active nests is greatest between April and July at the peak of the breeding season, and this period should definitely be avoided except for very small areas of vegetation clearance which will only be cleared under an ecologist's supervision.

Invertebrates

Bee bricks will be included within south facing walls of new buildings to provide new nesting habitat for solitary bees and other insects, at a ratio of one bee brick per two dwellings.

Trees

Adequate protective fencing (as outlined in Appendix F in the Preliminary Arboricultural Method Statement) will be installed around all retained trees where practical and will be erected before any materials and machinery are brought onto the site.

Site operations involving plant with booms jibs and counterweights will be planned in advance to prevent contact with retained trees. All operations involving such plant in close proximity to trees and hedges will be conducted under the supervision of a banksman to ensure that adequate clearance from the retained trees is maintained.

All pruning and contracting works will be carried out by a competent qualified contractor in accordance with BS 3998:2010: Tree Work Recommendations.

The exact locations of Site compounds and storage areas, and the routing of services for the Proposed Development, are not currently available. Adequate allowance for, and the planning of, such items will be made to avoid encroachment with the RPA of or prevent direct contact with all retained trees on Site.

While a preliminary Arboricultural Method Statement (AMS) has been provided, when further detail is known as to the construction process a bespoke Arboricultural Method Statement will also be required to protect the trees which are to be retained.

3.3. Operational Phase Mitigation Measures & Enhancements

Mitigation during operation

This LEMP outlines the management and enhancement actions to promote the landscape and ecological biodiversity value of the Site, and these will need to be detailed in the LEMP for each Reserved Matters Application as it comes forward. Although the RM LEMPs will be separate documents the management actions should be based on an integrated and complementary approach where possible. This will include management for new and retained habitats including woodlands, hedges, parks, wetlands and grasslands (ES para 11.7.8).

The Site includes provision for Suitable Alternative Natural Green Space (SANG), as mitigation for increased recreational impacts to the Penhale Dunes SAC. General management operations for the habitat elements



that will form the SANG are outlined below in sections 6.2 & 7 and will need to be specified in greater detail within the reserved matters application for delivery of the SANG and other early delivery of strategic Green Infrastructure.

New and retained marshy grassland or other wetland habitats need to be managed well to ensure they achieve the desired ecological condition, and that the loss of marshy grassland, which is also important for invertebrate species, and potential increase in human disturbance to such habitats is appropriately mitigated (requirement from ES chap 11 - see table 11.6). There will also be a need to ensure that some areas of marshy grassland on the valley floor have restricted access during the operational phase to allow continued use by overwintering snipe (ES para 11.7.13).

Street lighting design will need to follow best practice and avoid lighting known bat roost sites, woodlands, hedges and other areas of natural habitats. This will also benefit species groups other than bats, including invertebrates (ES chap 11- table 11.6).

To ensure that connectivity is maintained within residential areas, garden fences will need to be made permeable to hedgehogs by having small holes $(13 \text{ cm} \times 13 \text{ cm})$ cut into the base of fences. (ES chap 11- para 11.7.11 & table 11.6).

Include new bird nesting opportunities within the development to mitigate for lost nesting opportunities (ES chap 11, table 11.6). The number, type, and location for bird nest boxes will need to be identified by an ecologist as each Reserved Matters application is progressed.

Creation and management of habitat for wintering woodlark, following woodlark strategy (ES chap 11, table 11.6), and refer to woodlark mitigation strategy document (in prep). There will also be a need for areas to be set aside for ground nesting birds such as skylark and meadow pipit with limits on public access (and free-running dogs), at least during the breeding season (ES para 11.7.12).

Ecological Enhancement & Biodiversity Net Gain delivery

The Biodiversity Net Gain calculations have been based on the proposals within the Landscape Parameter Plan. It is therefore crucial that each Reserved Matters application that comes forward incorporates the soft landscape elements set out within the parameter plan and follows the requirements of the Design Code to secure the delivery of Biodiversity Net Gain. Each Reserved Matters Application will need to provide a statement to confirm that the design meets the requirements of the Parameters Plan and Design Code to provide assurance to the Local Planning Authority that the committed Biodiversity Net Gain will be delivered across the whole Site.

Newly created habitats need to be appropriately managed to maximise the increase in plant diversity and habitat condition that can be achieved within the site (ES chap 11 – table 11.6). There will be extensive areas of vegetated, and often semi-natural, habitats retained, enhanced and created within the Site. This includes strategic Green Infrastructure features of new woodland, planted on steep slopes near to the existing block of woodland at Governs (which is a County Wildlife Site and designated as Ancient Woodland), new and enhanced hedges, new and enhanced species-rich grassland and a network of SUDS, which will include features such as swales and permanently wet ponds and swales.



The existing woodland running along the valley bottom that forms the northern site boundary will be enhanced through sensitive landscaping of adjacent habitat, providing a broad band of semi-natural habitat that will buffer the existing woodland, both within and beyond the site boundary, strengthening this existing green corridor.

New woodland to be planted to the west of Governs Wood will increase the overall woodland habitat area present, and being located close to existing woodland will strengthen this important habitat, which is also part of a wider County Wildlife Site that stretches north-west from Governs Wood up the valley to Treworder Woods, thereby contributing to ecological networks in the wider landscape outside the site boundary.

The existing woodland in the Treliske Valley will also be strengthened through habitat enhancement and appropriate management.

Significant hedges are being retained, and these will be enhanced through planting up gaps and habitat enhancement/ sensitive management of buffer zones to strengthen these features which link into the key woodland and valley features discussed above forming a robust network of green corridors and habitat.

The precise management requirements of these new and enhanced habitats cannot be determined at this stage, as this will depend on the detailed design, which is yet to be undertaken. However, broad management activities for typical habitat types are set out in section 7.

In line with "Cornwall Planning for Biodiversity Guide", bat and bird bricks would be incorporated into buildings, at a minimum ratio of one per residential dwelling, and one bee brick in every other dwelling. The details of the locations of boxes would be developed alongside detailed designs for each stage of development. Bird bricks built into houses would be suitable for some species such as tits, sparrows and robins and also swifts, swallows and martins. The requirement for provision of bat, bird and bee bricks within buildings is also set out within the Design Code (section 13.06). Based on the requirements of the Cornwall Planning for Biodiversity Guide, up to 25% of the bat and bird boxes could be located in trees, these would need to be constructed of a long lasting material such as woodcrete, rather than wood (ES para 11.7.10).

There is no specific requirement to maintain built-in bat and bird bricks, as these are not usually easily accessible. If bird and bat boxes are placed in trees, they should be inspected annually, to ensure that they are still in place and in good condition. Bird and bat boxes should be checked between October and February to avoid disturbing breeding birds (bat box inspections should be undertaken by a suitably licensed ecologist).

3.4. Monitoring

Monitoring is essential to ensure that mitigation and enhancement actions have been undertaken as planned, and also to ensure that management regimes are appropriate. Monitoring allows the identification of any features that have not established successfully, which enables a review to be undertaken of the detailed management prescriptions and for any remediation to be undertaken. Some monitoring activities are also necessary to provide reports back to consenting authorities (e.g. the LPA on BNG and Natural England for



any bat licences).

Biodiversity Net Gain

The habitats will need to be monitored to ensure that they achieve the required +12.54% gain in hedgerow (at +56.83 hedgerow units) and the required gain in 28.77% of habitat area units (+202.74 habitat units). This metric calculation is currently being updated to account for revisions to the application design and will be amended in due course. These targets will need to be achieved and maintained for the required period of 30 years.

These habitats will need to be monitored by an ecologist at regular intervals to ensure that the overall habitat target of an increase is reached and maintained for the required period of 30 years.

Biodiversity Net Gain should be measured and monitored in line with the monitoring provisions established through the Langarth Stewardship Organisation (LSO). Monitoring is envisaged to commence at occupation of 250 houses with establishing full actual baseline data through a review of the BNG assessment, and then repeated every 5 years. This repeat assessment will comprise a survey of the site by a suitably experienced ecologist to ensure that the habitats and their target conditions are as specified in the original Biodiversity Net Gain metric completed in 2020. It will also need to include a new completed Biodiversity Net Gain metric (DEFRA version 2.0) for each assessment, to ensure that the required gains are met.

Key Performance Indicators

The following KPIs (forming part of the Benefits Realisation Strategy for LGV) with relevance to biodiversity and landscape are currently proposed, subject to final approval by the Strategic Board:

- Biodiversity (with a metric linked to achieving the required Biodiversity Net Gain, and a metric to allow for resident participation in species counts on site)
- Trees Established (with metrics linked to canopy cover, woodland area and number of trees within developed areas)
- Cornish hedges (with metrics linked to length and type of Cornish hedge i.e. tree or grass covered for example, and condition of Cornish hedges and their buffers)
- Green and Blue Infrastructure (with metrics linked to BwN accreditations and quality /usability of green infrastructure as perceived by residents)

Licence monitoring for bats

At the current time it is unknown whether any European Protected Species Mitigation licences will be required to permit the destruction or disturbance of existing bat roosts. Given that there are several known roosts around the site, it is likely that one or several licences will be required as individual phases of the development are brought forward to construction. It will be a requirement of these licences that post-construction monitoring of some form will be required periodically for a period of time after works are completed. The precise methodology, frequency and duration of this monitoring work will be determined as part of each licence application.



4. Green Infrastructure

The overarching strategy for the scheme layout has been to retain and integrate existing Green Infrastructure features where possible, and to use these as a framework for the development. The Green Infrastructure within the site is multi-functional, providing habitats for wildlife, high quality outdoor spaces for people and sustainable drainage.

Much of the enhanced and new Green Infrastructure features will be delivered as strategic Green Infrastructure early in the phasing of the development, and this will ensure that a significant proportion of the Green Infrastructure will be available to provide benefits for people, wildlife and water as early as possible. In particular, most of the strategic Green Infrastructure at Governs will be provided in the first phase of development, and the Green Infrastructure corridor along the northern boundary will largely be provided by the end of the second phase.

The Green Infrastructure features summarised below are described in detail within the Landscape Strategy and illustrated on the Landscape Parameter Plan.

4.1. Open Green Space

The open space provision within the Site can be split into six main types (as set out within section 3.3 of the Landscape Strategy):

- Park, amenity and civic space a number of parks and amenity spaces will be provided throughout the Garden Village. The key parks will be located close to the Northern Access Road and will be served by non-vehicular access, making them easily accessible hubs for people to use. Most of the parks will also incorporate SuDS features such as infiltration basins, informal sports pitches and habitats that will have ecological value.
- 2. Natural space Natural spaces will be designed with wildlife at the forefront and will also largely be suitable for use by people, providing opportunities for people to enjoy nature and to encourage active and healthy lifestyles.
- Outdoor sport a range of outdoor sport facilities are included within the green infrastructure, with a number of informal pitches of different sizes located at different locations throughout the site.
- 4. Children's play areas play areas are incorporated into accessible public spaces, ensuring that they are distributed across the site.
- 5. Teen provision a BMX track and skate park are included in the scheme design, along with a multiuse games area (MUGA).
- 6. Allotments two main allotment sites are incorporated into the green infrastructure design, and smaller scale micro-allotments are also provided within development parcels.



4.2. SuDS and Drainage

The scheme incorporates a comprehensive network of SuDs features, such as swales, infiltration basins and wet ponds to slow the flow of surface water run-off to the wider drainage system. As well as reducing flood risk (both on and off site), the SuDS features will also help to manage water quality and will provide opportunities for wildlife and people.

A series of swales running along contours collect surface water run-off from development parcels, and these connect with swales running perpendicular to the contours, which convey water to the system of infiltration basins and wet ponds. The perpendicular dams incorporate a system of check dams and wet features.

Wet ponds have been designed for places that have good opportunities for placemaking, so are located in prominent and visible locations.

Detailed design of these features will progress as Reserved Matters applications are brought forward, and the design will need to maximise opportunities for wildlife and amenity, as well as providing the required drainage function. Examples of factors to take into account in the design will be the use of naturalistic shapes with varied depth profiles, appropriate planting, design to maximise opportunities for the public to view and learn about wetland habitats, and ensuring that a proportion of the features will retain water throughout the year.

4.3. Built development plots

Parameters have been set for the proportions of landcover within the individual built development plots, based on the proposed plot densities within each residential grain (as defined within the Design Code). Detailed design for these plots will be progressed as each phase comes through for Reserved Matters, and scheme design will be guided by the Design Code that sets out the plot densities and subsequent proportions of landcover within each plot.

	G	arden Provision within Resi	dential Grains	
Residential Grain	Density	Communal Gardens / Micro Allotments (%)	Private Gardens (%)	Total Gardens (%)*
Settlement Edge - Clu	sters			
	Up to 35	35%	15%	50%
	Up to 40	25%	15%	40%
	Up to 50	20%	15%	35%
Settlement Edge - Hills	side	7°		
	Up to 35	30%	20%	50%
	Up to 40	20%	20%	40%
	Up to 50	15%	20%	35%
Village				
	Up to 40	25%	15%	40%
	Up to 50	25%	10%	35%
	Up to 60	25%	5%	30%
	Up to 140	30%	0%	30%
Jrban				
	Up to 40	30%	10%	40%
	Up to 50	30%	5%	35%
	Up to 60	30%	0%	30%
	Up to 140	30%	0%	30%

Figure 3 Design Code parameters on garden provision within different residential grains



Boundary Treatment within Neighbourhoods				
Neighbourhood	Density	Boundary	Treatment	
A - The Brake				
	Up to 35	A		
	Up to 40	E	3	
	Up to 50	C	>	
	Up to 140	c	>	
B - West Langarth				
	Up to 35	Ą	Ϋ́	
	Up to 40	A		
	Up to 50	В	С	
	Up to 60	C	;	
C - Langarth				
	Up to 40	A	В	
	Up to 50	E	3	
	Up to 60	C	;	
	Up to 140	c	>	
D - Governs				
	Up to 35	A		
	Up to 40	A	В	
E - Willow Green				
	Up to 50	В	С	
	Up to 60	C	>	
F - Penventinnie				
	Up to 35	A	Ŷ	
	Up to 40	A	В	
	Up to 50	В	С	
	Up to 60	C	;	
	Up to 140	c	>	

Figure 4 Design Code parameters on boundary treatments within different residential grains



For the purposes of the Biodiversity Metric at planning application stage, it has been assumed for residential plots that:

- Communal gardens/ micro allotments will meet the UK Habitats Classification of allotments
- Private gardens are assumed to be 50% UK Habitats Classification: urban- unvegetated garden and 50% urban-vegetated garden
- All the plots with a housing density of up to 35 per ha will have boundaries fully planted or Cornish hedges, while some of the plots with densities up to 40 per ha will have the same. These have been included in the metric as "Native hedgerow Associated with bank or ditch" and "Native hedgerow", with the new length of hedge split 50:50 between the two.
- As the management of the hedges will largely be in private hands, the condition has been assumed to be poor.
- For non-residential plots, it is assumed that 10% of landcover will be urban introduced shrub.

4.4. Access

The movement and access strategy is set out within the Design and Access Statement (section 7.04). The LEMP draws mainly on the non-vehicular movement and access provision. A bus route that will follow the route of NAR is proposed, with regular stops, and there will also be E-bike and E-car sharing hubs at strategic locations, to decrease the need for private car ownership and to promote car sharing.

There is a network of primary and secondary cycleways and footways throughout the site, which make use of the site's topography, creating safe and attractive routes to promote healthy lifestyles and wellbeing. Existing Public Rights of Way are incorporated into the network, and existing quiet lanes and tracks will be used as walking and cycling routes. These routes also link to existing footpaths and quiet lanes outside the Langarth Garden Village site which provide connectivity for pedestrians and cyclists towards the centre of Truro, as well as general connectivity into casual routes in the wider landscape.

There will be opportunities to cross the A390 – both existing and new.

4.5. Lighting

Lighting will be designed as each Reserved Matters application is brought forward, and it will need to be designed sensitively, as set out as a requirement for the ecological mitigation in the Ecology chapter of the Environmental Statement (para 11.7.9). Retained woodland, hedges and green corridors should be maintained as dark as possible following the latest guidance. The lighting design will need to ensure that the key bat habitats of woodland and hedges are kept dark. Lighting design should be developed in consultation with an ecologist, who will need to review the modelling of light spill in order to determine the correct lighting levels for bats and other nocturnal animals. Lighting plans for the NAR show lux levels and lighting directed to the carriageway, footways and cycleways in accordance with British Standard.



5. Ecological Trends and Constraints to Management

5.1. Invasive Non-Native Species

A number of non-native invasive plant species were identified on site (listed in section 3.1), though a detailed survey for invasives was not undertaken across the whole scheme. Given the long timescales associated with this project, detailed surveys for invasive species will need to be carried out as each phase of development is taken forward, and suitable control measures included within the individual phase/ project CEMPs.

It should be noted that Schedule 9 of the Wildlife and Countryside Act 1981 is reviewed periodically, and therefore the list of species to which the legislation applies can therefore change over time, and new species may be added.

5.2. Pests and Diseases

There is some ash within the site, and arboricultural advice will be required to determine whether any trees are diseased, and if so whether affected trees need to be felled.

It is also possible that new pests and diseases could arise during the implementation of the Langarth Garden Village, which could affect existing and newly established planting. As individual phases of development are brought forward, detailed planting plans will need to take into account any existing or emerging issues around plant disease and avoid susceptible species where appropriate. New planting schemes should be diverse and resilient to climate change to prepare for the impact of novel pests and diseases.

Without suitable protection and monitoring, the establishment of the new trees, shrubs and grassland and the process of natural regeneration is likely to be limited by rabbit, deer and grey squirrel.

5.3. Natural Succession

Habitats constantly change with lower plants colonising bare substrate, succeeded by vascular plants, scrub and then trees. Habitats also change when additional animal species start colonising the site. Without management, the process of natural succession is likely to lead to:

- Wildflower, wetland and amenity grassland will revert to rough grassland with an absence of wildflowers and ultimately will scrub over, which could then develop into woodland.
- Hedgerow trees and shrubs will develop into mature, leggy specimens causing a reduction in dense woody cover for wildlife.
- Ornamental shrub planting will become overgrown, woody and leggy, blocking light from entering the ground beneath it. Depending on their proximity to the recommended locations of bee bricks, they could also cover these nesting features for invertebrates, rendering them unusable.
- Woodland buffers will become overgrown and turning into an established woodland.
- The SuDS and surrounding wetland wildflower seeding will become dried out, silted up and dominated by grasses rather than wildflowers and eventually become rough grassland.



The proposed site management will need to consider the process of natural succession in maintaining and enhancing habitats, particularly in the amenity areas and the southern fields of open space / green infrastructure which will require more intensive management to maintain the design objectives and vision for each of the areas.

5.4. Protected Species

Maintenance works carried out by the landscape contractors will have to consider protected species when managing the site to ensure that these species are not illegally hurt, killed, or their habitats or resting places disturbed. Maintenance will be carried out via a precautionary approach, using appropriate methods and timed during suitable times of year where necessary. For example, songbirds are highly likely to nest in the hedges, trees, shrub and scrub habitat, as well as in the new specified bird boxes on site. Consequently, landscaping works to wooded habitat areas are recommended to be undertaken between October and February, outside the nesting period, to avoid disturbance to breeding birds during the summer months, as described in Section 8. If however this is not practical, potential nesting habitat will need to be checked immediately prior to removal by a suitably qualified ecologist.

Depending on the type of bird and bat boxes that are installed, these boxes may require maintenance. It will be important to ensure that the new homeowners/ tenants know whether they will be responsible for this, or whether the site maintenance team will do this. They will also need to be informed about what they can and cannot to these boxes, such as removing them and shining lights on the bat boxes. All UK bat species and their roosts are legally protected under the Conservation Regulations 2017.

There are numerous badger setts present across the site. Although widespread and common in Cornwall, badgers and their setts are legally protected under the Protection of Badgers Act 1992. The landscape maintenance staff will need to be informed about the presence of badger setts as part of their induction process, and these will need to be marked as ecological constraints on any maintenance plans.

As the arisings from cut vegetation will be left in brash piles (see Section 7.8) on site, it is highly likely that hedgehogs will readily use these as hibernacula throughout the winter months. Hedgehogs are Species of Principal Importance (SPI) and Cornwall BAP Priority species. It will be important to ensure that these species and their hibernation habitat is not disturbed. It is also likely that once these become established that reptiles will also use these as hibernation features. Reptiles are partially protected under the Wildlife and Countryside Act 1981. Again, it will be important to ensure that these species are not intentionally disturbed. As above, the landscape maintenance staff will need to be informed to leave such features in a quiet location on site and not to disturb it.



6. Management objectives and principles for Landscape and Ecological Features

6.1. Management objectives

There are three key management aims and associated objectives for the soft landscape within the Site, as set out within the Landscape Strategy (Appendix 2 to the Design and Access Statement), these are:

Aim I: A healthy, safe and secure Site:

- Maintain and facilitate public access and recreation, promoting healthier lifestyles;
- Provide a positive connection between existing neighbourhoods and surrounding public open spaces;
- Provide a high quality, well maintained and attractive landscape setting for the Proposed Development and local community;
- Establish and maintain health and safety procedures for the management and maintenance of the Site complying with all statutory legislation and best practice; and
- Provide an attractive environment for people to walk through and enjoy.

Aim 2: Conserve and enhance the ecological and landscape value of the Site:

- Safeguard and enhance the biological and physical integrity of the Site;
- Maintain and enhance the connectivity of habitats both within the Site and to adjacent areas;
- Minimise adverse effects on ecologically sensitive areas by controlling public access; and
- Enhance public awareness and appreciation of the Site's wildlife value.

Aim 3: A sustainable Site:

- Maximise the sustainability of Site operations by minimising waste;
- Promote an ecological based approach to landscape management where possible, by minimising the use of herbicide and where possible using mechanical methods of management in favour of machinery; and
- Maintain a flexible landscape management approach which responds to landscape changes and user requirements; and adopt a transparent management approach.

6.2. Headline maintenance and management operations for strategic GI

The strategic green infrastructure is largely made up of the following habitat/ planting types, and these will all require management and monitoring. As the detailed design progresses, detailed management prescriptions and a work programme can be drawn up. Management prescriptions will need to align with the generic management operations set out within the relevant tables in section 7.



- Native tree and woodland planting (Sections 7.1, 7.7, 7.8 & 7.9)
- Hedges (both retained and new hedges) (Sections 7.1, 7.4, 7.5 & 7.6)
- Species-rich grasslands (Sections 7.1 & 7.3)
- Planting and grassland areas established around SUDS (Sections 7.1, 7.3 & 7.11)
- Road verges and amenity grass/ shrub areas (Sections 7.1, 7.2 7.3 & 7.10)

In accordance with Cornwall Planning Policy, and Biodiversity Net Gain requirements, all habitats on site will need to be maintained for a period of 30 years. An initial 10-year maintenance programme has been created, see Section 8.1. Habitat condition will be reviewed in the 10th year to determine a maintenance schedule for the remaining 20 years.

Maintenance operations should generally be carried out in accordance with:

- BS4428: Code of Practice for General Landscape Operations,
- BS7370-4: Grounds Maintenance of Soft Landscape
- BS3998:2010. Tree work: recommendations.
- BALI Consumer Advice Caring for your Newly Planted Trees.

6.3. Headline maintenance and management requirements for Reserved Matters applications

The individual development plots will have less semi-natural habitat and will also have a greater variety of amenity planting types to reflect the higher levels of human activity within these plots. The general management operations set out in section 7 will still be relevant to many of the vegetation types that are established within development plots, though may have more specific management requirements. Reserved Matters applications will need to be accompanied by a LEMP that responds to the detailed design and explains the rationale and sets out management prescriptions and a work programme for the maintenance of soft landscaping within the site.

In accordance with Cornwall Planning Policy, and Biodiversity Net Gain requirements, all habitats on site will need to be maintained for a period of 30 years, dating from completion of construction of each reserved matters application. An initial 10-year maintenance programme has been created. Habitat condition will be reviewed in the tenth year to determine a maintenance schedule for the remaining 20 years.

Maintenance operations should generally be carried out in accordance with:

- BS4428: Code of Practice for General Landscape Operations,
- BS7370-4: Grounds Maintenance of Soft Landscape
- BS3998:2010. Tree work: recommendations.
- BALI Consumer Advice Caring for your Newly Planted Trees.



7. General Maintenance Operations

7.1. Strategic Green Infrastructure Management

The following section provides generic habitat management operations for the types of habitats proposed to be created as part of the strategic Green Infrastructure within the site. As detailed design of these areas is progressed, these management operations can be refined or adapted as necessary to suit the specific planting specifications.

In addition to the generic maintenance requirements carry out the following:

Ref	Phase	Works	Timing	Programming
7.1.1		<i>Invasive species</i> Carry out annual inspection for non-native invasive plant species, recording locations of Schedule 9 species (e.g. Japanese knotweed, montbretia). If growth is spreading and/ or problematic, seek specialist advice regarding control and legal disposal.	During summer (July to September) when growth will be more readily seen.	
7.1.2	Long Term	Litter Remove litter by hand and dispose of off site, to keep public spaces free of general litter, garden waste, dog faeces and debris.	Year round	From completion of first RMA development, for completed areas
7.1.3		Tree Safety Undertake tree safety survey covering all site trees (initially covering existing retained trees but to include new tree planting after 5yrs) on an annual basis (or as required to fulfil legal/ insurance requirements) and carry out any maintenance operations as recommended. If any damage or failures are noted or reported seek arboricultural advice immediately.	Annual, in winter when trees not in leaf as better visibility.	

7.2. Amenity Grass (Seeded)

In addition to the generic maintenance requirements carry out the following:

Ref	Phase	Works	Timing	Programming
7.2.1	Establishment	Establishment Watering If dry conditions persist after sowing, undertake watering as required to ensure healthy establishment	Spring/ Summer	
7.2.2		Grass cutting – First cut When establishing sward reaches 80mm and ground conditions are reasonably dry undertake first cut of grass to 40mm	When sward reaches 80mm	Year Lonwards
7.2.3		Grass cutting Continue cutting to 25mm when sward reaches 80mm	March to November	
7.2.4	Long term	Repairl reinstatement Inspect condition of grass at least once annually and undertake any repair or reinstatement of damaged/ failed areas as required, through rectification of any rutting, disturbed or compacted ground and reseeding	Spring	

7.3. New Areas of Wildflower Meadow/ Wetland Wildflowers (Seeded)

In addition to the generic maintenance requirements carry out the following:

Ref	Phase	Works	Timing	Programming
7.3.1		Establishment Watering		
	Establishment	If dry conditions persist after sowing, undertake watering as required to ensure healthy establishment	Spring/ Summer	Year I

CEC3264-01h Langarth Garden Village

Report DRAFT- 12 October 2021

Ref	Phase	Works	Timing	Programming
7.3.2		Cutting (First Year) First cut – in spring/ summer, once sward height has reached 75mm cut to 50mm Subsequent cuts – cut to 50mm every 6 to 8 weeks until autumn In all cases, remove clippings	Spring to Autumn	
7.3.3		Spring cut Cut to 40-70mm to remove excess grass.	March	From Year 2
7.3.4	Long form	Autumn cut Cut to 40mm after flowering. Remove the clippings after 3 -4 days (allowing seeds to disperse)	September/October	
7.3.5		<i>Litter and weed control</i> Clear non-biodegradable rubbish. Control invasive species if required.	As required	From Year I
7.3.6		Repair/ reinstatement Inspect condition of grass at least once annually and undertake any repair or reinstatement of damaged/ failed areas as required, through rectification of any rutting, disturbed or compacted ground and reseeding with original wildflower mix	Autumn	From Year 2

7.4. Existing Hedges in Woody Condition

In addition to the generic maintenance requirements carry out the following:

Ref	Phase	Works	Timing	Programming
7.4.1		Trimming		
	Long term	Trim in small sections across the site, cutting approx. 20% (of total length) of hedge face each year in a 5-year rotational cycle. Where shrubs exist, trim back to A-shape of 2.5 - 3m high shrubs, overhanging sides of Cornish hedges and with shaded side of hedge at shallower angle. Do not trim or damage trees during hedge trimming.	December to February	Year I onwards

7.5. Existing Hedges to be Strengthened

In addition to the generic maintenance requirements carry out the following:

Ref	Phase	Works	Timing	Programming
7.5.1		Watering Undertake watering of whip planting for first year through dry periods, as required for healthy establishment	Late spring to summer as required	Years I
7.5.2	Establishmant	Stake, guard and mulch reinstatement Undertake regular checks on polythene guards and stakes and mulch matting and carry out reinstatement or re-fixing works as necessary. Hand weed tubes/guards where required.	Throughout season	Veens I to 2
7.5.3	– Establishment	Hedge face trimming Close trim hedge sides to reduce competition to planted woody species	Annually during winter (November to February)	Tears T to 5
7.5.4		Stake and guard removal Remove plant shelters and stakes. For shrubs, trim approx 1/3 rd off previous year's growth to promote bushier growth	Winter (November to February)	End of year 3

CEC3264-01h Langarth Garden Village Strategic LEMP Report DRAFT– 12 October 2021 29

Ref	Phase	Works	Timing	Programming
7.5.5		Cutting Do not Cut	n/a	Years 4 and 5
7.5.6		Replacement of failed planting At the end of each summer until five years from planting, check all planting for failures. Ensure any plants that have failed are replaced at the earliest opportunity to maintain an overall stocking rate of 100% through the maintenance period.	Check August Plant October- March	Years I to 5
7.5.7	Long Term	Trimming Trim in small sections across the site, cutting approx. 20% (of total length) of hedge face each year in a 5-year rotational cycle. Where shrubs exist, trim back to A-shape of 2.5 - 3m high shrubs, overhanging sides of Cornish hedges and with shaded side of hedge at shallower angle. Do not trim or damage trees during hedge trimming.	December to February	Year 6 onwards

7.6. New Cornish Hedge

In addition to the generic maintenance requirements carry out the following:

Ref	Phase	Works	Timing	Programming
7.6.1	Establishment	Watering Undertake watering of whip planting for first year through dry periods, as required for healthy establishment	Late spring to summer as required	Years I
7.6.2		Stake, guard and mulch reinstatement Undertake regular checks on polythene guards and stakes and mulch matting and carry out reinstatement or re-fixing works as necessary. Hand weed tubes/guards where required.	Throughout season	Years I to 3

CEC3264-01h Langarth Garden Village

Strategic LEMP

Report DRAFT- 12 October 2021

Ref	Phase	Works	Timing	Programming
7.6.3		Hedge face trimming Close trim hedge sides to reduce competition to planted woody species	Annually during winter (November to February)	
7.6.4		Stake and guard removal Remove plant shelters and stakes. For shrubs, trim about 1/3 rd off previous year's growth to promote bushier growth	Winter (November to February)	End of year 3
7.6.5		Cutting Do not cut	n/a	Years 4 and 5
7.6.6		Replacement of failed planting At the end of each summer until five years from planting, check all planting for failures. Ensure any plants that have failed are replaced at the earliest opportunity to maintain an overall stocking rate of 100% through the maintenance period.	Check August Plant October- March	Years I to 5
7.6.7	Long Term	Trimming Trim in small sections across the site, cutting approx. 20% (of total length) of hedge face each year in a 5-year rotational cycle. Where shrubs exist, trim back to A-shape of 2.5 - 3m high shrubs, overhanging sides of Cornish hedges and with shaded side of hedge at shallower angle. Do not trim or damage trees during hedge trimming.	December to February	Year 6 onwards

7.7. Native Shrub and Woodland Edge Planting

In addition to the generic maintenance requirements carry out the following:

Ref	Phase	Works	Timing	Programming
7.7.1	_	Watering Undertake watering of whip planting for first year through dry periods, as required for healthy establishment	Late spring to summer as required	
7.7.2		 Wildflower cutting between woody plants (first year) First cut – in spring/ summer, once sward height has reached 100mm strim to 50mm Subsequent cuts – strim to 50mm every 6 to 8 weeks until autumn In all cases, take care to avoid damage to woody stock and remove the clippings. 	Spring to Autumn	Years I
7.7.3	Establishment	Wildflower cutting between woody plants (subsequent years) Cut to 40mm after flowering. Remove the clippings after 3 -4 days (allowing seeds to disperse)	September/October	
7.7.4	_ Establishment	Stake, guard and mulch reinstatement Undertake regular checks on polythene guards and stakes and mulch matting and carry out reinstatement or re-fixing works as necessary. Hand weed tubes/guards where required.	Throughout season	Years I to 3
7.7.5		Stake and guard removal Remove plant shelters and stakes. For shrubs, trim approx 1/3 rd off previous year's growth to promote bushier growth	Winter (November to February)	End of year 3
7.7.6		Replacement of failed planting At the end of each summer until five years from planting, check all planting for failures. Ensure any plants that have failed are replaced at the earliest opportunity to maintain an overall stocking rate of 100% through the maintenance period.	Check August Plant October-March	Years I to 5

Ref	Phase	Works	Timing	Programming
7.7.7		Trimming/ clearance		
	Long Term	Each winter cut approx. 20% area in a piecemeal fashion (in a 5yr rotation) to a height of approx. 1.5m to maintain structural diversity.	December to February	Year 5 onwards
		Cut other areas only as per operational and access requirements.		

7.8. Woodland Planting

In addition to the generic maintenance requirements carry out the following:

Ref	Phase	Works	Timing	Programming
7.8.1		Watering Undertake watering of whip planting for first year through dry periods, as required for healthy establishment	Late spring to summer as required	
7.8.2	Establishment	Wildflower cutting between woody plants (first year) First cut – in spring/ summer, once sward height has reached 100mm strim to 50mm Subsequent cuts – strim to 50mm every 6 to 8 weeks until autumn In all cases, take care to avoid damage to woody stock and remove the clippings.	Spring to Autumn	Years I
7.8.3		Wildflower cutting between woody plants (subsequent years) Cut to 40mm after flowering. Remove the clippings after 3 -4 days (allowing seeds to disperse)	September/October	
7.8.4		Stake, guard and mulch reinstatement Undertake regular checks on polythene guards and stakes and mulch matting and carry out reinstatement or re-fixing works as necessary. Hand weed tubes/guards where required.	Throughout season	Years I to 3

CEC3264-01h Langarth Garden Village Strategic LEMP Report DRAFT– 12 October 2021 33

Ref	Phase	Works	Timing	Programming
7.8.5		Stake and guard removal Remove plant shelters and stakes. For shrubs, trim about 1/3 rd off previous year's growth to promote bushier growth	Winter (November to February)	End of year 3
7.8.6		Replacement of failed planting At the end of each summer until five years from planting, check all planting for failures. Ensure any plants that have failed are replaced at the earliest opportunity to maintain an overall stocking rate of 100% through the maintenance period.	Check August Plant October-March	Years I to 5
7.8.7	Long Term	Thinning When specimens reach overall height of 9m (around year 10 for nurse species or year 15 for hardwoods) commence thinning programme. Thin to favour hardwoods over nurse species. To be carried out by qualified arboricultural contractor with aim of removing dead or diseased specimens, reducing competition for light and growing space and creating an evenly distributed canopy of open foliage. Arisings left on site in brash piles for habitat value.	December to February	Year 10 on
7.9. Tree Planting (Larger Stock)

In addition to the generic maintenance requirements carry out the following:

Ref	Phase	Works	Timing	Programming	
7.9.1		Watering – year I Undertake watering as necessary to promote heathy growth of trees, ensuring that full depth of topsoil is wetted during each watering procedure. This is likely to require watering twice each week through any summer drought periods	May September	Year I	
7.9.2	Watering – years 2 & 3 Undertake watering as necessary to promote heathy growth of trees, ensuring that full depth of topsoil is wetted during each watering procedure. This is likely to require watering each week through any summer drought periods		May September	Year 2 and 3	
7.9.3	Establishment	 Stake, guard and mulch reinstatement and tie adjustment Undertake regular checks on strimmer guards and associated stakes and mulch matting and carry out reinstatement or re-fixing works as necessary. Remove by hand any weeds/grass which are growing within the guards. Adjust, refix or replace loose or defective ties, allowing for growth and to prevent chafing. 		Years I to 3	
7.9.4	-	Replacement of failed planting At the end of each summer until five years from planting, check all planting for failures. Ensure any trees that have failed are replaced at the next suitable planting season (winter).	Check August Plant October- March	Years I to 5	
7.9.5	9.5 Stake and guard removal After checking trees have firmly established (and that staking is no longer re stake, associated ties and strimmer guard.		End of season	Year 5	
7.9.6	Long term	Long term Until trees reach 5m overall height, prune crown by removing dead branches and reducing selected side branches by one third to preserve a well-balanced head and ensure the development of a single streng leader with minimum clean stem of 2m. Persona durling deal branches development of a single streng leader with minimum clean stem of 2m. Persona durling durling deal branches deal branches durling deal bra		Years 5 – 10	

CEC3264-01h Langarth Garden Village

Strategic LEMP Report DRAFT– 12 October 2021

Ref	Phase	Works	Timing	Programming
		branches and potentially weak or tight forks. In each case cut back to live wood in accordance with current best practice for pruning of trees. To be carried out by a qualified arboricultural contractor.		

7.10. Amenity and Ornamental Planting

In addition to the generic maintenance requirements carry out the following:

Ref	Phase	Works	Timing	Programming
7.10.1		Watering Undertake watering whenever dry periods persist for longer than two weeks, or if new planting is showing any signs of drought related stress		Year I
7.10.2	Weed control and mulchingMaintain mulching layer at a thickness of at least 50mm. Hand pull any persistent weeds ensuring removal of entire plant and increase thickness of mulch if weeds are a continuing problem.		Quarterly or as required, if weed coverage exceeds 10% or mulching layer is thinner than 50mm	Years I-5
7.10.3	Replacement of failed planting At the end of each summer until five years from planting, check all planting for failures. Ensure any plants that have failed are replaced the following spring to maintain an overall stocking rate of 100% through the maintenance period.		Check end of summer, replacement planting in following spring	Years I - 5
7.10.4	Long Term	Herbaceous plants and grasses For deciduous species, cut back to 10cm above ground. Remove cuttings. Monitor progress of more vigorous plants and control through hand weeding if required to maintain planting composition.		Year 2+

CEC3264-01h Langarth Garden Village Strategic LEMP

Report DRAFT– 12 October 2021

Ref	Phase	Works		Programming	
7.10.5		Weed control post establishment Depending on degree of ground coverage achieved at the end of year 4, weeding requirements after year 4 should be minimal. However, there should still be an annual inspection yearly to ensure no aggressive or invasive weeds have established, with problematic weed species treated/ removed as per section 4.2.1	July inspection, with any treatment carried out August – September	Year 5+	
7.10.6		Thinning Thin grasses and herbaceous plants only if groups / clumps become unmanageable and untidy. If this occurs, remove some clumps completely and rejuvenate remaining clumps by division and replanting.	Early spring		

7.11. Permanent Ponds (in SuDS Basins)

In addition to the generic maintenance requirements carry out the following:

Ref	Phase	Works	Timing	Programming
7.11.1		Algal bloom treatment		
		weather), treat with applications of barley straw, which when it breaks down releases compounds which inhibit algal growth. Applications should be at the rate of about 10g / m2 surface area.	February – April &	
	Establishment	Stuff the straw loosely in mesh bags (to be made of a biodegradable material) tie a float and anchor to the bottom of the pond with a stone so that the bag floats just below the surface. If algal growth is particularly heavy it can be raked out and left to dry and rot down adjacent to the pond.	September – November (if required)	Years I and 2 (if required)
		Treat as follows:		
		Feb-April Ist application		
		Sept-Nov 2nd application		

CEC3264-01h Langarth Garden Village Strategic LEMP Report DRAFT– 12 October 2021

Ref	Phase	Works	Timing	Programming
		Feb-April 3rd application and remove 1st		
		Sept-Nov 4th application and remove 2nd, etc.		
		(from BTCV Handbook – Ponds & Wetlands)		
7.11.2		Plant growth of aquatic plants		
		Plant growth should not be a problem in the first two years. Leave all die-back in-situ over winter to provide habitat for invertebrates.	April / May	Years I and 2
		Replace any lost or failed plants and protect from waterfowl as necessary.		
7.11.3		Leaks		
		Should a lined pond area fail to hold water, or there is a rapid fall in water level, there may be a leak in the liner. In this case the pond should be drained and the cause of the leak determined and fixed.	October	All years as appropriate
7.11.4		Plant growth of aquatic plants		
	Long term	Thin / remove about 20% of excess growth per year to prevent excessive silting and to retain open water. Cut back plants in October, leave submerged / floating plant material on the edge of the pond to allow invertebrates and amphibians to re-enter the water prior to disposal.	October	
		Pond areas must be monitored for invasive plants on a monthly basis and if occurring should be removed immediately.	Monthly	Year 3 onwards
7.11.5		Brambles and woody growth around pond edges		
		Remove any branches and suckers encroaching into pond maintenance access and meadow areas.	January / February	
		Trim brambles once a year to maximum 1.5m height.		

Ref	Phase	Phase Works		Programming
7.11.6		Control of any woody vegetation growing in lined ponds and basins		
		Remove any woody saplings that grow within lined areas or the wider basin by hand weeding to prevent damage to liner.		
7.11.7		Rubbish & debris		
		Remove any rubbish from the pond as soon as it appears. Remove excess twigs / leaves from surface in the autumn to prevent nutrient build-up. Allow for weekly inspections of SUDS ponds.	All year	
7.11.8		Siltation		
		Excessive silt build up should be removed in October to minimise disturbance to amphibians (at the end of amphibian season, but before they start to hibernate). Ensure that a layer of silt at least 200mm deep is left in the ponds. Do not disturb more than 30% of the pond each year.	October	Year 6 onwards as necessary

7.12. Bat, beebird boxes/ bricks

Ref	Phase	Works	Timing	Programming
7.12.1	Bat boxes attached to trees should be inspected annually by a licensed bat ecologist. Any ba or evidence of bats found in the boxes will be recorded, and boxes will be maintained/ replace as required.		Once, between May and October	All years
7.12.2	Long term	Bird boxes attached to trees should be inspected annually, between October and February (inclusive), to remove any old nesting material and undertake box maintenance/ replace boxes as required.	Once, between October and February	All years
7.12.3		Bee bricks built into buildings should be kept clear of overgrowing vegetation		All years

CEC3264-01h Langarth Garden Village

8. Work Programme, Management Responsibilities and Review

8.1. Indicative Ten-Year Work Programme

							Мо	nth	1					1				Y	ear	ų.			
Ref	Management Action 1 F M A M 1 1 A S O N D 1 2 3 4 5 6 7 8 9							9	10														
7 1	General Maintenance Operations - Whole Site			1.1			-			-	<u> </u>			- 1	-		-		<u> </u>		-		10
7.1.1	teneral Maintenance Operations - Whole Site	1		-	-	-	-		- T	-			_	-	_		-	-		-			
/.1.1	Invasive species												_										
7.1.2	Litter																						
7.1.3	Tree safety					1 08																	
7.2	Amenity grass areas (seeded)																						
721	Establishment watering													1							Т		
722	Grass outting - First out		2 7	1		7. 1		8				-									-	\rightarrow	-
7.2.2	Grass cutting - This cut		2									-	-		-		-		_				
7.2.5	Grass cutting	_	-										_										_
1.2.4	Repair/ reinstatement				_																		
7.3	New areas of wildflower meadow/ wetland wildflowers (s	seec	led))				×.0		6 13 ³													/
7.3.1	Establishment watering																						
732	Cutting (First Year)										8 °												
733	Spring out				-			8		-		-					1				_		
7.3.3	Spring cut	-	N				-	-					-	-	,				_		-	-	
7.3.4	Autumn cut			_						_	_		_										
7.3.5	Litter and weed control																						
7.3.6	Repair/ reinstatement	1																					
7.4	Existing Hedges in woody condition						× ×																
741	Trimming	-	-															10					
75	Existing Hedges to be strengthened	.0	1000 C				-			_					8 50	2	2	() <u> </u>					
7.5	Existing neages to be strengthened		-		-	r	1		- T				_										_
/.5.1	Watering																				$ \rightarrow $	$ \blacksquare $	
7.5.2	Stake, guard and mulch reinstatement																						
7.5.3	Hedge face trimming																						
7.5.4	Stake and guard removal										1												
755	Do not Cut						-			-											\rightarrow	-+	-
7.5.5	Do not out		0			1	2	-		-	-					-			_		\rightarrow	-+	
7.5.6					-			-			\rightarrow										\rightarrow	$ \rightarrow $	
7.5.7	Irimming																						
7.6	New Cornish hedge																						
7.6.1	Watering																						
762	Stake, guard and mulch reinstatement	1																			$\neg \uparrow$	\neg	
763	Hedge face trimming					i i	-	8		-			-		-		-		-		-		-
7.0.5			0	-	-		-	_		-			-		9			_	_	_		-	
7.6.4	Stake and guard removal									_								_				_	
7.6.5	Do not Cut												_		_			-					
7.6.6	Replacement of failed planting																						
7.6.7	Trimming																						8 0
7.7	Native shrub and woodland edge planting	5																				_	
771	Matrice sinds and woodland cage planting		0				1				- 1	-	-							_	<u> </u>		-
1.1.1	watering	_		_	_					_		_	_		_		_	_				\rightarrow	
7.7.2	Wildflower cutting between woody plants (first year)												_										
7.7.3	Wildflower cutting between woody plants (subsequent years)																						
7.7.4	Stake, guard and mulch reinstatement																						
775	Stake and guard removal		-				_					1				2					-		
776	Benjacement of foiled planting		9	-	-	-	5	-		-		-	-						_	-		-	-
7.7.0		_				_				-	-		_						_	_	-	\rightarrow	_
1.1.1	Trimming/ clearance																						
7.8	Woodland planting					1322 Mar				4 - 14													
7.8.1	Watering																						
782	Wildflower cutting between woody plants (first year)		1 1			1/					-												
702	Wildflewer cutting between woody plants (mist year)					S		6		-	-	-	-								-		
7.0.5	windhower cutting between woody plants (subsequent years)	-	1			_		_		-	0 - 10 	_	-			8	1					_	
7.8.4	Stake, guard and mulch reinstatement																				_	\rightarrow	_
7.8.5	Stake and guard removal																						
7.8.6	Replacement of failed planting													-									
7.8.7	Trimming/ clearance											1											
7.9	Tree planting (larger stock)			-	-	-											_	-				-	
701	Watering - year 1										- 1	Т	-						-		— т	— T	
7.9.1	Watering - year 1	-	à					-		\rightarrow	\rightarrow	_	_		_				_		\rightarrow	\rightarrow	
7.9.2	watering - year 2 & 3																			$ \rightarrow $	\rightarrow	$ \rightarrow $	
7.9.3	Stake, guard and mulch reinstatement and tie adjustment							1															
7.9.4	Replacement of failed planting																						
7.9.5	Stake and guard removal		8															1					
7.9.6	Formative pruning																				$\neg \uparrow$		
7.10	Amenity and ornamental planting					-	-		<u> </u>											-	_		
7 10 1		1	-		-	1		-		- 1		T	_		-		-	-	Т	-		—	
/.10.1	watering										$ \rightarrow $										\rightarrow	$ \rightarrow $	
7.10.2	Weed control and mulching													-									
7.10.3	Replacement of failed planting																						
7.10.4	Herbaceous plants and grasses																	1					
7 10 5	Weed control post establishment		0			1																	
7 10 6	Thinning					<u> </u>					-+	-+	-		-				_		\rightarrow	\rightarrow	
1.10.6		1				I					_		_										
7.11	Permanent ponds (in SUDS basins)	-	_	_	_	_						_	_	_			_						
7.11.1	Algal bloom treatment													1									
7.11.2	Weed growth of aquatic plants		6			1															T		
7 11 3	leaks	1																					
7 11 4	Plant growth of aquatic plants		-	3	-										-				-		-+		
7.11.4	Prevelation and use due provide and the second se													-	-		-						
7.11.5	Bramples and woody growth around pond edges					 								_									
7.11.6	Control of any woody vegetation growing in lined ponds											1.0											
7.11.7	Rubbish & debris																						
7.11.8	Siltation		-																				
7.12	Bat and Bird Boxes	100	_	-	-	-		()															
7 10 1		1	_		-	-		-		Т		Т	_	-			-		-	-			
7.12.1	bat poxes inspection (licensed ecologist)																						
1.12.2	Bird boxes maintenance					í			1														

40

CEC3264-01h Langarth Garden Village

Strategic LEMP

CE

Report DRAFT– 12 October 2021

8.2. Reporting and Review

The landscape contractor engaged for the delivery of each phase of work will need to keep a record of all site attendance and inspections during the contracted establishment period, and the contracts will need to ensure that they keep records and report as follows (these requirements also need to be set out within the LEMP for each individual project):

- Information required to be recorded on every site inspection:
 - Date and time spent on site
 - Weather conditions
 - Name of inspecting person / maintenance person
 - Tasks undertaken
 - Defects rectified
 - Issues observed and to be addressed, with action plan for addressing them
- The records are to be submitted on a quarterly basis for inspection by the developer, Landscape Architect and Planning Authority. Records are to be kept for a minimum of 5 years after the end of the 24 months establishment period.

Ongoing inspections will be required after the establishment period, these will need to be set out within the individual LEMPs, but as a minimum will need to include:

- yearly tree safety inspections from year 5 onwards (to be undertaken by qualified arboricultural consultant with experience in undertaking tree safety assessments), and
- yearly ROSPA inspections of the play equipment from year 2 onwards (undertaken by ROSPA inspector).

These specific inspections will be documented by the professionals and reports must be filed by the management company and added to the Health & Safety File for the site.

Additional ecological inspections to monitor the habitats and their conditions will be completed by a suitably experienced ecologist in Years 2, 5, 7, 10, 15, 20, 25 and 30 following occupation of 250 houses onwards, The monitoring process will be managed by the Langarth Stewardship Organisation (LSO); and they will also complete a new Biodiversity Net Gain Metric (using DEFRA version 2.0 or the metric applicable at the time) and accompanying report each time. These reports will be made available for review by Cornwall Council, as the Local Planning Authority.



A. Biodiversity Net Gain Strategy



Framework Biodiversity Net Gain Strategy for Langarth Garden Village Cornwall

12 October 2021 DRAFT

Prepared for: Cornwall Council

Prepared by: Cornwall Environmental Consultants (CEC Ltd)



Ref: CEC3264-01h

CEC®

cornwall environmental consultants ltd

five acres, allet, truro, cornwall, tr4 9dj t: 01872 245510 f: 01872 262071 e: enquiries@cecenvironment.co.uk www.cecenvironment.co.uk

Document Control

Version number	Prepared by	Approved by	
First Draft Issue	Jenny Stuart MSc, CEnv, MCIEEM	Natasha Collings-Costello MEnv Sci, CEnv, CSci, MCIEEM	Natasha Collings-Costello MEnv Sci, CEnv, CSci, MCIEEM

Disclosure

The information, opinion and advice which we have prepared and provided is true and has been prepared and provided in accordance with the CIEEM's Code of Professional Conduct and the British Standard for Biodiversity – Code of practice for planning and development (2013). We confirm that the opinions expressed are our true and professional bona fide opinions.



cornwall environmental consultants ltd

five acres, allet, truro, cornwall, tr4 9dj t: 01872 245510 f: 01872 262071 e: enquiries@cecenvironment.co.uk www.cecenvironment.co.uk

Contents

1. Intr	roduction2	
1.1.	Background & Aims 2	•
2. Min	imising impacts on biodiversity4	•
2.1.	Embedded Design Measures4	ļ
2.2.	Pre-construction and Construction Phase Mitigation Measures 4	ļ
2.3.	Operational Phase Mitigation Measures5)
3. Bio	diversity Net Gain Calculations and Assumptions	,
4. Del	ivering Biodiversity Net Gain 10)
Appendi	x A Biodiversity Net Gain Metric Spreadsheet11	



CEC[®] | cornwall environmental consultants ltd

five acres, allet, truro, cornwall, tr4 9dj t: 01872 245510 f: 01872 262071 e: enquiries@cecenvironment.co.uk www.cecenvironment.co.uk

I. Introduction

I.I. Background & Aims

Cornwall Environmental Consultants Ltd (CEC) were commissioned by Arcadis on behalf of Cornwall Council in July 2021 to prepare a Framework Biodiversity Net Gain Strategy for Langarth Garden Village. This is a standalone document, but also forms an appendix to the Strategic LEMP for Langarth Garden Village.

The Langarth Garden Village planning application is a hybrid application, incorporating a full application for construction of the Northern Access Road and an outline application for the rest of the Langarth Garden Village. As such, the planning application is accompanied by documents setting out the parameters for the proposed development (largely captured within the parameter plans and Design Code), but individual projects within the Site will be taken forward to detailed design and construction as separate Reserved Matters applications, by a range of different developers/ applicants. The Biodiversity Net Gain measurements and calculations for the hybrid application have therefore been based on outline design information, based on a number of assumptions and design parameters set out within the planning application documentation.

The National Planning Policy Framework and Cornwall Council Local Planning Policy, both require schemes to achieve a net gain for biodiversity, though they did not specify a means for achieving or measuring net gain. The UK Government announced in the 2019 Spring Statement that Biodiversity Net Gain will become mandatory in England. Whilst the timetable for implementation of this requirement nationally is not yet known, Cornwall Council have applied a requirement for major development schemes to achieve a minimum 10% Biodiversity Net Gain, from March 2020.

Biodiversity Net Gain has been measured in accordance with current guidance, and using the DEFRA Biodiversity Metric 2.0, as required by Cornwall Council. The metric calculated a gain in 28.77% habitat area units (+202.74 habitat units) and a gain of 12.5% in hedgerow units (+56.83 hedgerow units) (dated November 2020). This metric calculation is currently being updated to account for revisions to the application design and will be amended in due course. Once updated this Biodiversity Net Gain Strategy will be re-issued. The habitats will need to be monitored during the operational phase, to ensure that these habitat units are provided and the habitats are managed appropriately.

This Strategy aims to set out how the Biodiversity Net Gain proposed as part of this project will be delivered through the individual Reserved Matters applications, with reference to supporting documentation from the hybrid planning application.

This strategy summarises the steps that have been, or will be taken to minimise the adverse effect of the development on biodiversity (section 2) and summarises the pre-development and post-development biodiversity value of the onsite habitat (section 3). Section 4 summarises the steps that will need to be taken



CEC3264-01h Langarth Garden Village Framework Biodiversity Net Gain Strategy August 2021 – DRAFT issue as part of each Reserved Matters application to demonstrate that the predicted Biodiversity Net Gain will be delivered.



2. Minimising impacts on biodiversity

2.1. Embedded Design Measures

Impacts on biodiversity have been considered throughout the design process, and scheme design has embedded many measures that will avoid and mitigate potential adverse impacts. This is discussed in detail within section 11.5 of the Environmental Statement, but key points are summarised below:

- Retention and enhancement of existing open spaces (e.g. woodland and grassland) to ensure that there are suitable spaces for people to use for recreation, This will provide a Suitable Alternative Natural Green (SANG) area, to reduce the potential increase in recreational pressure on Penhale Dunes Special Area of Conservation (SAC), some 9km from the Site, and Carrine Common SAC, c. 1.5km from the Site.
- Provision of suitable drainage systems, throughout construction and operation, that will control surface water run-off and prevent pollution to the Fal and Helford SAC (c. 3.5km downstream of the Site at its closest point).
- Retention and enhancement of key green corridors within the Site (designed around existing river valleys, hedges with mature tree cover and quiet lanes with hedges on either side).
- Retention of existing wetland area, and provision of new Sustainable Drainage Systems (SUDS), which will provide additional wetland habitats within the Site.
- Creation of new hedges in mitigation for the sections of hedge that will be lost to the development and provide an overall enhancement in hedge habitat (see also Biodiversity Net Gain Calculations below).
- Provision of badger crossings under the new Northern Access Road at appropriate locations, to help badgers move around the landscape while providing a safe route for them to negotiate the new road.
- Lighting for the Northern Access Road to be designed with sensitivity for wildlife, particularly bat species.

2.2. Pre-construction and Construction Phase Mitigation Measures

Section 3.1 & 3.2 of the Strategic LEMP sets out the measures that will need to be taken prior to and during construction to minimise impacts on biodiversity. In summary, these comprise:

- Updated ecological surveys to be undertaken, as appropriate, for Reserved Matters applications.
- General best practice construction measures, such as:
 - Environmental awareness amongst site personnel, including Tool Box Talks provision



- Appropriate measures and equipment in place to minimise risk of pollution incidents
- Trenches being covered overnight, or having ramped sides, and general site tidiness to be maintained.
- Removal/ control of non-native invasive species
- Any badger setts directly impacted to be closed under a Natural England badger licence
- Any bat roosts impacted to be subject to obtaining a European Protected Species licence from Natural England
- Implementation of on-site woodlark mitigation strategy
- Clearance of vegetation and demolition of buildings should (in the case of the latter, where this is
 not impacted by advice for minimising impacts to bats) be undertaken between the months of
 October and February to avoid the risk of disturbing, damaging or destroying active bird's nests. If
 not possible, ecologists will advise on suitable alternative approaches.
- Build bee, bat and bird bricks/ boxes into new buildings
- Appropriate protection of retained trees

2.3. Operational Phase Mitigation Measures

Section 3.3 of the Strategic LEMP identifies the mitigation measures that will need to be implemented during the operational phase of the project. These are listed below:

- Preparation of a LEMP for each Reserved Matters application, following the framework set out within the strategic LEMP
- Appropriate management of the SANG (this will only apply to those Reserved Matters applications that incorporate the SANG)
- Where it occurs within a Reserved Matters application area, new and retained marshy grassland will need to be managed appropriately, and some areas to have restricted access to provide undisturbed areas for use by wintering birds, such as snipe
- Street lighting to be designed following best practice for minimising impacts to wildlife, and specifically avoid lighting bat roosts, woodland, hedges and natural habitats
- Ensure that garden fencing is permeable to hedgehogs, by providing gaps at least 13cm x 13cm at the base of close-boarded fences
- Include new bird nesting opportunities within the development at a ratio of one bird box per dwelling



• Creation and management of habitat for wintering woodlark, following the On-site Woodlark Mitigation and Compensation Strategy, and ensure that some areas of open grassland or arable habitats are set aside for breeding skylark (with restrictions on free-running dogs), at least during the breeding season.



3. Biodiversity Net Gain Calculations and Assumptions

The Biodiversity Net Gain calculations are summarised in table I below, and shown in full in the accompanying metric spreadsheet in Appendix A (dated November 2020). This metric calculation is currently being updated to account for revisions to the application design and will be amended in due course. Once updated this Biodiversity Net Gain Strategy will be re-issued. Measurements are taken from the following documents:

- Landscape parameters plan, ref: LAN_02.1-AHR-MP-ZZ-DR-A-92-009 Proposed Landscape
- Plot density parameter plan, ref: LAN_02.1-AHR-MP-ZZ-DR-A-93-010 Density Parameter Plan

On site baseline	Habitat units	704.73			
	Hedgerow units	453.13			
On site post-intervention	Habitat units	916.64			
	Hedgerow units	509.97			
Total net unit change	Habitat units	+ 202.74			
	Hedgerow units	+ 56.83			
Total net % change	Habitat units	+ 28.77%			
	Hedgerow units	+ 12.54%			

• NAR landscape design plans, ref: X_0312 NAR Landscape GA

Due to the size and complexity of this proposal, and the level of detail on scheme design, there are a number of assumptions that have been made regarding the post-construction biodiversity value.

Habitats and hedges are assigned a "strategic significance" within the metric. These have been taken from the Lagas website which shows strategic zones (https://lagas.co.uk/app/product/netgain_vectorzones). However the boundaries of the three zones are mapped in insufficient detail as to follow field boundaries. Therefore there has had to be some interpretation of the zone boundaries.

Housing plots have been divided into three different landcover types for the metric (proportions of each type varies between plot densities – these proportions are illustrated in Figures 3 & 4 of the Strategic LEMP);

- I. Communal Gardens/Semi-natural, Allotments: Defined as Allotment in metric
- 2. Private gardens: 50% input as urban/unvegetated garden and 50% as urban/vegetated garden in metric (this is considered to provide a realistic estimate of the area of gardens that will be vegetated, and which will be unvegetated (e.g. patios or decking).



3. Built up areas: Urban/developed land

Non-residential plots have been assessed as 90% "Urban/developed land" and 10% "Urban/introduced shrub"

Natural open spaces have been totalled and then allocated at a 30:50:20 ratio as follows:

- 30% put "Within area formally identified in local strategy"
- 50% in "Location ecologically desirable but not in local strategy"
- 20% in "Area/compensation not in local strategy/ no local strategy".

Areas of grassland that are being enhanced have been subtracted from these areas before they were put into the metric. They have been input as "other neutral grassland" as this is of medium distinctiveness, and condition as "moderate".

Parkland has been assumed to be low distinctiveness habitat "Modified grassland". In reality, these are likely to be more ecological diverse as they will have patches of tree and shrub planting and areas of more formally amenity grassland and other areas allowed to grow longer.

These areas do not exist in the valley bottoms (which are all "Within area formally identified in local strategy"), so away from the northernmost wooded valley, 10% have been put to "Location ecologically desirable but not in local strategy" and 90% to "Area/compensation not in local strategy/ no local strategy".

Street trees all appeared to be along the Northern Access Road (NAR) and A390 so 100% have been put in "Area/compensation not in local strategy/ no local strategy".

Equipped play areas are assumed to be 50% grass (amenity grassland) and 50% safety surface (sealed surface).

Hedges Phase I habitat survey and Landscape design do not map hedges in the same way. In Phase I survey they are mapped as linear features with only a length and no area. This is also how the DEFRA metric handles hedges. However, the landscape design maps hedges as areas, the length of these is also measured to be used in the metric for the hedge calculations.

The landscape design included 16.4ha for existing hedgerows and trees. Most hedges on site had arable or grassland either side but there are short sections, often close to the valley floors where scrub had spread out from the hedge. To prevent possible double counting of scrub habitat, only 90% of this existing hedge measurement has been added into the Metric as "mixed scrub" at fairly poor, with 100% in "Area/compensation not in local strategy/ no local strategy".

Existing hedgerow enhancement covered c. 8ha in the landscape design. This has been added into the Metric as "mixed scrub" at moderate condition, in addition to the linear hedge biodiversity units. 10% in "Within area formally identified in local strategy", 20% in "Location ecologically desirable but not in local



strategy" and 70% in "Area/compensation not in local strategy/ no local strategy".

Proposed new hedgerows covered 0.142ha these have been added into the Metric as "mixed scrub" at moderate condition in "Area/compensation not in local strategy/ no local strategy" (as above, these areas of scrub recognise that the hedges on site have considerable width, which is not reflected in the metric for hedges, which are treated as linear features with no width.

The landscape design did not show hedges within woodlands or those that formed the boundaries of woodlands. Therefore, if the woodland is retained it has been assumed that the hedge is also retained.

There will be new boundaries within the residential plots. All the plots with a housing density of up to 35 per ha will have boundaries fully planted or Cornish hedges, while some of the plots with densities up to 40 per ha will have the same. These have been included in the metric as "Native hedgerow – Associated with bank or ditch" and "Native hedgerow", with the new length of hedge split 50:50 between the two. As the management of the hedges will largely be in private hands, the condition has been assumed to be poor.

Following DEFRA guidance on watercourses, as the watercourses on site will be retained within their well vegetated corridors, which will be enlarged and enhanced, they are assessed as being in moderate condition and not included within the calculations.



4. Delivering Biodiversity Net Gain

In order to ensure that the required Biodiversity Net Gain is delivered as promised, each Reserved Matters Application will need to be designed in accordance with the requirements set out within the Landscape Parameters Plan and the Langarth Garden Village Design Code.

It will not be a requirement for each Reserved Matters application to re-calculate the Biodiversity Net Gain for each application. Much of the Biodiversity Net Gain for the Langarth Garden Village will be delivered through the Strategic Green Infrastructure that will be implemented in Phases I & 2.1t is anticipated that it will not be possible to deliver a minimum of 10% Biodiversity Net Gain as part of each individual Reserved Matters application, so the Biodiversity Net Gain will be assessed over the Langarth Garden Village as a whole.

Each Reserved Matters application will need to include a Biodiversity Conformity Statement and Implementation Timetable to demonstrate compliance with this Framework Biodiversity Strategy.



Appendix A Biodiversity Net Gain Metric Spreadsheet (dated November 2020)



The Biodiversity Metric 2.0 - Calculation Tool Start page

	Instructions	
Planning authority:	Cornwall Council	
Project name:	Langarth Garden Village	
Applicant:	Cornwall Council	
Application type:	Hybrid	
Planning application reference:		ivialit filetiu
Assessor:	Steve Adams	
Reviewer:	Jenny Stuart	
Revision:		
Assessment date:	20/10/2020	Results
Planning authority reviewer:		
	Cell style conventions	View all
	Enter data	
	Automatic lookup	Reset view
	Result	Reset view

Langarth Garden Village

Return to results menu

Headline Results

	Habitat units	704.73
On-site baseline	Hedgerow units	453.13
	River units	0.00
On site past intervention	Habitat units	907.48
On-site post-intervention	Hedgerow units	509.97
(Including habitat retention, creation, enhancement & succession)	River units	0.00
	Habitat units	0.00
Off-site baseline	Hedgerow units	0.00
	River units	0.00
Off_site post_intervention	Habitat units	0.00
On-site post-intervention	Hedgerow units	0.00
(Including habitat retention, creation, enhancement & succession)	River units	0.00
Total net unit change	Habitat units	202.74
rotarnet unit change	Hedgerow units	56.83
(including all on-site & off-site habitat retention/creation)	River units	0.00
Total net % change	Habitat units	28.77%
Total net % change	Habitat units Hedgerow units	28.77% 12.54%

Langarth Garden Village	Paturo to results
Detailed Results	menu

Summary Figures



On site area change by habitat group

Woodland and Intertidal Coastal forest sediment saltmarsh

Wetland Woodland and Intertidal Coastal saltmarsh Rocky shore Coastal lagoons forest sediment

On-site habitat retention category

biodiversity units

 Units Retained Baseline units enhanced

= Units succession Units lost

Rocky shore Coastal lagoon:

Urban

Existing area Proposed area Area change

Existing value Proposed value Unit change

Area / length retained

Area / length enhanced

Area / length successio Area / length lost

On-site habitat retention by category

area (hectares)

Unit change by habitat group

Wetland

	Basi	eline	Post development on site Onsite Cha									
Habitat group	Existing area	Existing value	Proposed area	Proposed value	Area change	Onsite Unit change						
Cropland	113.3	226.5	0.0	0.0	-113.3	-226.5						
Grassland	76.4	137.5	-34.8	114.5	-111.2	-23.0						
Heathland and shrub	7.3	20.9	18.5	150.0	11.2	129.1						
Rivers and lakes	0.3	0.0	0.0	5.0	-0.3	5.0						
Sparsely vegetated land	0.6	1.0	-0.5	-0.8	-1.1	-1.8						
Urban	32.6	7.8	126.4	241.8	93.8	234.0						
Wetland	0.0	0.0	0.0	0.0	0.0	0.0						
Woodland and forest	14.7	2.3	4.6	227.5	-10.2	225.2						
Intertidal sediment	0.0	0.0	0.0	0.0	0.0	0.0						
Coastal saltmarsh	0.0	0.0	0.0	0.0	0.0	0.0						
Rocky shore	0.0	0.0	0.0	0.0	0.0	0.0						
Coastal Jagoons	0.0	0.0	0.0	0.0	0.0	0.0						

Overall C	hange	
Area change	Unit change	
-113.3	-226.5	
-111.2	-23.0	
11.2	129.1	
-0.3	5.0	
-1.1	-1.8	
93.8	234.0	
0.0	0.0	
-10.2	225.2	
0.0	0.0	
0.0	0.0	
0.0	0.0	
0.0	0.0	

		Base	eline	Post develo	pment Off-site	Off-site Change		
	Habitat group	Existing area	Off-site Existing value	Proposed area	Off site Proposed value	Area change	Offsite Unit change	
ĺ	Cropland	0.0	0.0	0.0	0.0	0.0	0.0	
I	Grassland	0.0	0.0	0.0	0.0	0.0	0.0	
I	Heathland and shrub	0.0	0.0	0.0	0.0	0.0	0.0	
I	Rivers and lakes	0.0	0.0	0.0	0.0	0.0	0.0	
I	Sparsely vegetated land	0.0	0.0	0.0	0.0	0.0	0.0	

Urban	0.0	0.0	0.0	0.0	0.0	0.0
Wetland	0.0	0.0	0.0	0.0	0.0	0.0
Woodland and forest	0.0	0.0	0.0	0.0	0.0	0.0
Intertidal sediment	0.0	0.0	0.0	0.0	0.0	0.0
Coastal saltmarsh	0.0	0.0	0.0	0.0	0.0	0.0
Rocky shore	0.0	0.0	0.0	0.0	0.0	0.0
Coastal lagoons	0.0	0.0	0.0	0.0	0.0	0.0
Combined	Base	line	Combined Po	st development	Combin	ed change
Habitat group	Existing area	Existing value	Proposed area		Proposed area	Proposed value
		1				226.5
Cropland	113.3	226.5	0.0	0.0	-113.3	-220.5
Cropland Grassland	113.3 76.4	226.5 137.5	-34.8	0.0 114.5	-113.3	-226.5
Cropland Grassland Heathland and shrub	113.3 76.4 7.3	226.5 137.5 20.9	0.0 -34.8 18.5	0.0 114.5 150.0	-113.3 -111.2 11.2	-226.5 -23.0 129.1
Cropland Grassland Heathland and shrub Rivers and lakes	113.3 76.4 7.3 0.3	226.5 137.5 20.9 0.0	0.0 -34.8 18.5 0.0	0.0 114.5 150.0 5.0	-113.3 -111.2 11.2 -0.3	-226.5 -23.0 129.1 5.0
Cropland Grassland Heathland and shrub Rivers and lakes Sparsely vegetated land	113.3 76.4 7.3 0.3 0.6	226.5 137.5 20.9 0.0 1.0	0.0 -34.8 18.5 0.0 -0.5	0.0 114.5 150.0 5.0 -0.8	-113.3 -111.2 11.2 -0.3 -1.1	-226.5 -23.0 129.1 5.0 -1.8
Cropland Grassland Heathland and shrub Rivers and lakes Sparsely vegetated land Urban	113.3 76.4 7.3 0.3 0.6 32.6	226.5 137.5 20.9 0.0 1.0 7.8	0.0 -34.8 18.5 0.0 -0.5 126.4	0.0 114.5 150.0 5.0 -0.8 241.8	-113.3 -111.2 11.2 -0.3 -1.1 93.8	-226.5 -23.0 129.1 5.0 -1.8 234.0
Cropland Grassland Heathland and shrub Rivers and lakes Sparsely vegetated land Urban Wetland	113.3 76.4 7.3 0.3 0.6 32.6 0.0	226.5 137.5 20.9 0.0 1.0 7.8 0.0	0.0 -34.8 18.5 0.0 -0.5 126.4 0.0	0.0 114.5 150.0 5.0 -0.8 241.8 0.0	-113.3 -111.2 -0.3 -1.1 93.8 0.0	-220.3 -23.0 129.1 5.0 -1.8 234.0 0.0
Cropland Grassland Heathland and shrub Rivers and lakes Sparsely vegetated land Urban Wetland Woodland and forest	113.3 76.4 7.3 0.3 0.6 32.6 0.0 14.7	226.5 137.5 20.9 0.0 1.0 7.8 0.0 2.3	0.0 -34.8 18.5 0.0 -0.5 126.4 0.0 4.6	0.0 114.5 150.0 5.0 -0.8 241.8 0.0 227.5	-113.3 -111.2 11.2 -0.3 -1.1 93.8 0.0 -10.2	-228.5 -23.0 129.1 5.0 -1.8 234.0 0.0 225.2
Cropland Grassland Heathland and shrub Rivers and lakes Sparsely vegetated land Urban Wetland Woodland and forest Intertidial sediment	113.3 76.4 7.3 0.3 0.6 32.6 0.0 14.7 0.0	226.5 137.5 20.9 0.0 1.0 7.8 0.0 2.3 0.0	0.0 -34.8 18.5 0.0 -0.5 126.4 0.0 4.6 0.0	0.0 114.5 150.0 -0.8 241.8 0.0 227.5 0.0	-113.3 -111.2 11.2 -0.3 -1.1 93.8 0.0 -10.2 0.0	-228.5 -23.0 129.1 5.0 -1.8 234.0 0.0 225.2 0.0
Cropland Grassland Heathland and shrub Rivers and lakes Sparsely vegetated land Urban Wetland Woodland and forest Intertidal sediment Coastal saltmarsh	113.3 76.4 7.3 0.3 0.6 32.6 0.0 14.7 0.0 0.0	226.5 137.5 20.9 0.0 1.0 7.8 0.0 2.3 0.0 0.0	0.0 -34.8 18.5 0.0 -0.5 126.4 0.0 4.6 0.0 0.0	0.0 114.5 150.0 -0.8 241.8 0.0 227.5 0.0 0.0	-113.3 -111.2 11.2 -0.3 -1.1 93.8 0.0 -10.2 0.0 0.0	-228.3 -23.0 129.1 5.0 -1.8 234.0 0.0 225.2 0.0 0.0
Cropland Grassland Heathland and shrub Rivers and lakes Sparsely vegetated land Urban Wetland Woodland and forest Intertidial sediment Coastal saltmarsh Rocky shore	113.3 76.4 7.3 0.3 0.6 32.6 0.0 14.7 0.0 0.0 0.0	226.5 137.5 20.9 0.0 1.0 7.8 0.0 2.3 0.0 0.0 0.0 0.0	0.0 -34.8 18.5 0.0 -0.5 126.4 0.0 4.6 0.0 0.0 0.0 0.0	0.0 114.5 150.0 5.0 -0.8 241.8 0.0 227.5 0.0 0.0 0.0 0.0	-113.3 -111.2 11.2 -0.3 -1.1 93.8 0.0 -10.2 0.0 0.0 0.0 0.0	-228.3 -23.0 129.1 5.0 -1.8 234.0 0.0 225.2 0.0 0.0 0.0

Lang	arure	Jaruen vin	age
A-1	Site	Habitat	Baselin

Main Menu

arth Garden Village
Site Habitat Baseline
Condense / Show Rows
Condense / Show Rows

		Habitats and areas		Habitat distinctiveness	Habitat condition	Ecological connectivity	Strategic significance		Ecological		Retention category biodiversity value							Bespoke	ce Comments	
Re	f Broad Habitat	Habitat type	Area (hectares)	Distinctiveness	Condition	Ecological connectivity	Strategic significance	Suggested action to address habitat losses	Total habitat units	Area	a Ari ed enhai	rea Area Inced successi	Baseline units	e Baseline units	Baseline units	Area lost	Units lost	agreed for unacceptable losses	Assessor comments	Reviewer comments
1	Cropland	Cropland - Cereal crops	12.3886	Low	N/A - Agricultural	Low	Area/compensation not in local	Same distinctiveness or better	24.78				0.00	0.00	0.00	12.39	24.78		Governs arable	
2	Cropland	Cropland - Cereal crops	74.0244	Low	N/A -	Low	Area/compensation not in local	Same distinctiveness or better	148.05				0.00	0.00	0.00	74.02	148.05		Langarth arable	
3	Cropland	Cropland - Cereal crops	13.3513	Low	N/A -	Low	Area/compensation not in local	Same distinctiveness or better	26.70				0.00	0.00	0.00	13.35	26.70		Maiden Green arable	
4	Cropland	Cropland - Cereal crops	13.503	Low	N/A -	Low	Area/compensation not in local	Same distinctiveness or better	27.01				0.00	0.00	0.00	13.50	27.01		Willow Green arable	
5	Heathland and shrub	Heathland and shrub - Bramble scrub	0.0316	Medium	Poor	Low	Area/compensation not in local	Same broad habitat or a higher	0.13				0.00	0.00	0.00	0.03	0.13			
6	Heathland and shrub	Heathland and shrub - Bramble scrub	0.0387	Medium	Poor	low	Area/compensation not in local	Same broad habitat or a higher	0.15				0.00	0.00	0.00	0.04	0.15			
-	Heathland and shrub	Heathland and shrub - Bramble scrub	0.109	Medium	Moderate	low	strategy/ no local strategy Location ecologically desirable but	Same broad habitat or a higher	0.96		0.1	109	0.00	0.96	0.00	0.00	0.00			
	Heathland and shrub	Heathland and shrub - Bramble scrub	0.021	Medium	Poor	Low	not in local strategy Area/compensation not in local	distinctiveness habitat required Same broad habitat or a higher	0.08	_			0.00	0.00	0.00	0.02	0.08			
	Manthland and shock	Heathland and shrub - Bramble scrub	0.015	Madium	Deer	Law	strategy/ no local strategy Area/compensation not in local	distinctiveness habitat required Same broad habitat or a higher	0.05		_		0.00	0.00	0.00	0.02	0.05			
H	Heathland and shock	Heathland and shrub - Bramble scrub	0.015	Madium	Deer	Low	strategy/ no local strategy Area/compensation not in local	distinctiveness habitat required Same broad habitat or a higher	1.76				0.00	0.00	0.00	0.44	1.76			
	Headiland and shrub	Heathland and shrub - Bramble scrub	0.44	Wediam	PUUI	LOW	strategy/ no local strategy Location ecologically desirable but	distinctiveness habitat required Same broad habitat or a higher	1.76		_		0.00	0.00	0.00	0.44	1.70			
1	Heatniand and shrub	Heathland and shrub - Bramble scrub	0.03	Medium	Poor	LOW	not in local strategy Location ecologically desirable but	distinctiveness habitat required Same broad habitat or a higher	0.13		_		0.00	0.00	0.00	0.03	0.13			
1	Heathland and shrub	Heathland and shrub - Bramble scrub	0.21	Medium	Poor	Low	not in local strategy Within area formally identified in	distinctiveness habitat required Same broad habitat or a higher	0.92	_			0.00	0.00	0.00	0.21	0.92			
1	Heathland and shrub	Heathland and shruh - Bramble scrub	0.13	Medium	Poor	Low	local strategy Within area formally identified in	distinctiveness habitat required Same broad babitat or a higher	0.60		_		0.00	0.00	0.00	0.13	0.60			
14	Heathland and shrub	Heathland and shrub - Bramble scrub	0.04	Medium	Poor	Low	local strategy Within area formally identified in	distinctiveness habitat required	0.18				0.00	0.00	0.00	0.04	0.18			
1	Heathland and shrub	Transmission and an operations action	0.12	Medium	Poor	Low	local strategy	distinctiveness habitat required	0.55	_	_		0.00	0.00	0.00	0.12	0.55			
10	Heathland and shrub	Heathland and shrub - Bramble scrub	0.06	Medium	Poor	Low	not in local strategy	distinctiveness habitat required	0.26				0.00	0.00	0.00	0.06	0.26			
13	Heathland and shrub	Heathland and shrub - Bramble scrub	0.06	Medium	Poor	Low	Within area formally identified in local strategy	Same broad habitat or a higher distinctiveness habitat required	0.28	_			0.00	0.00	0.00	0.06	0.28			
11	Heathland and shrub	Heathland and shrub - Bramble scrub	0.04	Medium	Poor	Low	Within area formally identified in local strategy	Same broad habitat or a higher distinctiveness habitat required	0.18				0.00	0.00	0.00	0.04	0.18			
19	Heathland and shrub	Heathland and shrub - Bramble scrub	0.58	Medium	Poor	Low	Area/compensation not in local strategy/ no local strategy	Same broad habitat or a higher distinctiveness habitat required	2.32				0.00	0.00	0.00	0.58	2.32			
20	Heathland and shrub	Heathland and shrub - Bramble scrub	0.067	Medium	Poor	Low	Location ecologically desirable but not in local strategy	Same broad habitat or a higher distinctiveness habitat required	0.29				0.00	0.00	0.00	0.07	0.29			
2:	Heathland and shrub	Heathland and shrub - Bramble scrub	0.052	Medium	Poor	Low	Within area formally identified in local strategy	Same broad habitat or a higher distinctiveness habitat required	0.24		0.0	152	0.00	0.24	0.00	0.00	0.00			
2	Heathland and shrub	Heathland and shrub - Bramble scrub	0.496	Medium	Poor	Low	Within area formally identified in local strategy	Same broad habitat or a higher distinctiveness habitat required	2.28		0.4	196	0.00	2.28	0.00	0.00	0.00			
2	Heathland and shrub	Heathland and shrub - Bramble scrub	0.095	Medium	Poor	Low	Within area formally identified in local strategy	Same broad habitat or a higher distinctiveness habitat required	0.44		0.0	095	0.00	0.44	0.00	0.00	0.00			
24	Heathland and shrub	Heathland and shrub - Bramble scrub	0.605	Medium	Poor	Low	Area/compensation not in local strategy/ no local strategy	Same broad habitat or a higher distinctiveness habitat required	2.42				0.00	0.00	0.00	0.61	2.42			
2!	Heathland and shrub	Heathland and shrub - Gorse scrub	0.068	Medium	Moderate	Low	Location ecologically desirable but not in local strategy	Same broad habitat or a higher distinctiveness habitat required	0.60		0.0	068	0.00	0.60	0.00	0.00	0.00			
21	Heathland and shrub	Heathland and shrub - Gorse scrub	0.041	Medium	Poor	Low	Location ecologically desirable but	Same broad habitat or a higher distinctiveness babitat required	0.18		0.0	041	0.00	0.18	0.00	0.00	0.00			
2	Heathland and shrub	Heathland and shrub - Gorse scrub	0.03	Medium	Poor	Low	Area/compensation not in local	Same broad habitat or a higher distinctiveness habitat required	0.12				0.00	0.00	0.00	0.03	0.12			
21	Grassland	Grassland - Bracken	0.05	Medium	Poor	Low	Area/compensation not in local strategy/ no local strategy	Same broad habitat or a higher distinctiveness habitat required	0.20				0.00	0.00	0.00	0.05	0.20			
25	Grassland	Grassland - Bracken	0.021	Medium	Poor	Low	Area/compensation not in local	Same broad habitat or a higher	0.08				0.00	0.00	0.00	0.02	0.08			
30	Grassland	Grassland - Modified grassland	0.16	Low	Poor	Low	Area/compensation not in local strategy/ no local strategy	Same distinctiveness or better	0.32				0.00	0.00	0.00	0.16	0.32			
3:	Grassland	Grassland - Modified grassland	2.873	Low	Poor	Low	Area/compensation not in local	Same distinctiveness or better	5.75		2.8	373	0.00	5.75	0.00	0.00	0.00		Governs grassland	
3	Grassland	Grassland - Modified grassland	6.714	Low	Poor	Low	Area/compensation not in local	Same distinctiveness or better	13.43		5.7	/14	0.00	11.43	0.00	1.00	2.00		Governs grassland	
3	Grassland	Grassland - Modified grassland	3.807	Low	Poor	Low	Within area formally identified in	Same distinctiveness or better	8.76		-		0.00	0.00	0.00	3.81	8.76			
34	Grassland	Grassland - Modified grassland	10.256	Low	Poor	Low	Area/compensation not in local	Same distinctiveness or better	20.51				0.00	0.00	0.00	10.26	20.51			
3	Grassland	Grassland - Modified grassland	3.907	Low	Poor	Low	strategy/ no local strategy Within area formally identified in	habitat required Same distinctiveness or better	8.99		-		0.00	0.00	0.00	3.91	8.99			
24	Grassland	Grassland - Modified grassland	0.76	low	Poor	low	local strategy Area/compensation not in local	Same distinctiveness or better	1.52		-		0.00	0.00	0.00	0.76	152			
2	Grassland	Grassland - Modified grassland	0.99	Low	Poor	Low	strategy/ no local strategy Area/compensation not in local	habitat required Same distinctiveness or better	1.98				0.00	0.00	0.00	0.99	1.98			
3.	Grassland	Grassland - Modified grassland	0.77	Low	Pour	Low	strategy/ no local strategy Area/compensation not in local	habitat required Same distinctiveness or better	1.50	_	-		0.00	0.00	0.00	0.99	1.50			
31	Grassiand	Grassland - Modified grassland	0.01	LOW	POOR	LOW	strategy/ no local strategy Area/compensation not in local	habitat required Same distinctiveness or better	0.02	_	_		0.00	0.00	0.00	0.01	0.02			
3	Grassland	Grassland - Modified grassland	1.14	Low	Poor	Low	strategy/ no local strategy Area/compensation not in local	habitat required Same distinctiveness or better	2.28				0.00	0.00	0.00	1.14	2.28			
41	Grassland	0	0.67	Low	Poor	Low	strategy/ no local strategy	habitat required	1.34				0.00	0.00	0.00	0.67	1.34			

41 Grassland	Grassland - Modified grassland	0.14	Low	Poor	Low	Area/compensation not in local	Same distinctiveness or better	0.28		0.00 0.00 0.00	0.14	0.28			
42 Grateland	Grassland - Modified grassland	0.69	low	Roor	low	Area/compensation not in local	habitat required Same distinctiveness or better	1.29		0.00 0.00 0.00	0.69	1.29			
	Grassland - Modified grassland	0.05	1.514	100	1.011	strategy/ no local strategy Within area formally identified in	habitat required Same distinctiveness or better	1.50		0.00 0.00	0.05	1.30			
43 Grassland	for start the Product of	0.02	Low	Poor	Low	local strategy	habitat required	0.05		0.00 0.00 0.00	0.02	0.05			
44 Grassland	Grassiano - Modimed grassiano	0.05	Low	Poor	Low	local strategy	habitat required	0.12		0.00 0.00 0.00	0.05	0.12			
45 Grassland	Grassland - Modified grassland	0.03	Low	Poor	Low	Within area formally identified in local strategy	Same distinctiveness or better habitat required	0.07		0.00 0.00 0.00	0.03	0.07			
46 Grassland	Grassland - Modified grassland	0.11	Low	Poor	Low	Within area formally identified in local strategy	Same distinctiveness or better habitat required	0.25		0.00 0.00 0.00	0.11	0.25			
47 Grassland	Grassland - Modified grassland	1.49	Low	Poor	Low	Within area formally identified in	Same distinctiveness or better	3.43		0.00 0.00 0.00	1.49	3.43			
48 Grassland	Grassland - Modified grassland	1.12	Low	Poor	Low	Within area formally identified in	Same distinctiveness or better	2.58		0.00 0.00 0.00	1.12	2.58			
49 Grassland	Grassland - Modified grassland	0.09	law	Poor	law	Vithin area formally identified in	Same distinctiveness or better	0.21		0.00 0.00 0.00	0.09	0.21			
To Creation	Grassland - Modified grassland	0.02		Dear	1 mm	local strategy Within area formally identified in	habitat required Same distinctiveness or better	0.05		0.00 0.00 0.00	0.02	0.05			
SU Grassianu	Grassland - Modified grassland	0.02	LOW	POU	LOW	local strategy Within area formally identified in	habitat required Same distinctiveness or better	0.05		0.00 0.00	0.02	0.03			
51 Grassland	Grantland - Modified grantland	0.38	Low	Poor	Low	local strategy Within area formally identified in	habitat required	0.87		0.00 0.00 0.00	0.38	0.87			
52 Grassland	Chasting - Modified Restand	0.04	Low	Poor	Low	local strategy	habitat required	0.09		0.00 0.00 0.00	0.04	0.09			
53 Grassland	Grassland - Modified grassland	1.114	Law	Poor	Low	strategy/ no local strategy	habitat required	2.23		0.00 0.00 0.00	1.11	2.23			
54 Grassland	Grassland - Modified grassland	5.564	Low	Moderate	Low	Area/compensation not in local strategy/ no local strategy	Same distinctiveness or better habitat required	22.26		0.00 0.00 0.00	5.56	22.26			
55 Grassland	Grassland - Modified grassland	2.89	Low	Poor	Low	Location ecologically desirable but not in local strategy	Same distinctiveness or better babitat required	6.36		0.00 0.00 0.00	2.89	6.36			
56 Grassland	Grassland - Modified grassland	2.097	Low	Poor	Low	Area/compensation not in local	Same distinctiveness or better	4.19		0.00 0.00 0.00	2.10	4.19			
57 Grassland	Grassland - Modified grassland	0.641	Law	Poor	Low	Location ecologically desirable but	Same distinctiveness or better	1.41		0.00 0.00 0.00	0.64	1.41			
58 Grassland	Grassland - Modified grassland	1,485	Low	Poor	Low	Location ecologically desirable but	Same distinctiveness or better	3.27		0.00 0.00 0.00	1.49	3,27			
59 Grandand	Grassland - Modified grassland	14	Law	Roor		not in local strategy Area/compensation not in local	habitat required Same distinctiveness or better	2.90		0.00 0.00 0.00	1.40	2.90			
Grassianu	Grassland - Modified grassland	1.4	LOW	POUL	LOW	strategy/ no local strategy Location ecologically desirable but	habitat required Same distinctiveness or better	2.00		0.00 0.00	1.40	2.00			
60 Grassland	Grassland - Modified gravitand	2.543	Low	Poor	Low	not in local strategy Area/compensation not in local	habitat required	5.59		0.00 0.00 0.00	2.54	5.59			
61 Grassland	Grassianu - Woulineu grassianu	3.983	Low	Poor	Low	strategy/ no local strategy	habitat required	7.97		0.00 0.00 0.00	3.98	7.97			
62 Grassland	Grassiand - Modified grassland	1.3379	Low	Poor	Low	Area/compensation not in local strategy/ no local strategy	saine distinctiveness or better habitat required	2.68		0.00 0.00 0.00	1.34	2.68			
63 Grassland	Grassland - Modified grassland	1.6817	Low	Poor	Low	Area/compensation not in local strategy/ no local strategy	Same distinctiveness or better habitat required	3.36		0.00 0.00 0.00	1.68	3.36			
64 Grassland	Grassland - Modified grassland	2.8397	Low	Poor	Low	Area/compensation not in local strategy/ no local strategy	Same distinctiveness or better habitat required	5.68		0.00 0.00 0.00	2.84	5.68			
65 Grassland	Grassland - Modified grassland	2.4972	Low	Poor	Low	Area/compensation not in local	Same distinctiveness or better	4.99		0.00 0.00 0.00	2.50	4.99			
66 Grassland	Grassland - Modified grassland	1.4442	Law	Poor	Low	Area/compensation not in local	Same distinctiveness or better	2.89		0.00 0.00 0.00	1.44	2.89			
67 Grassland	Grassland - Modified grassland	0.9831	Law	Poor	Low	Area/compensation not in local	Same distinctiveness or better	1.97		0.00 0.00 0.00	0.98	1.97			
68 Grateland	Grassland - Modified grassland	0.4165	law	Roor	low	Area/compensation not in local	Same distinctiveness or better	0.92		0.00 0.00 0.00	0.42	0.92			
Constant	Grassland - Other neutral grassland	0.026	Madium	Deer	Laure .	strategy/ no local strategy Location ecologically desirable but	habitat required Same broad habitat or a higher	0.11	0.006	0.00 0.11 0.00	0.00	0.00	Go	overns grassland	
	Grassland - Other neutral grassland	0.020		1001		not in local strategy Within area formally identified in	distinctiveness habitat required Same broad habitat or a higher	0.11	0.010	0.00 0.11 0.00	0.00	0.00			
70 Grassland	Grassland - Other neutral grassland	0.165	Medium	Poor	Low	local strategy Within area formally identified in	distinctiveness habitat required Same broad babitat or a bisher	0.76		0.00 0.00 0.00	0.17	0.76			
71 Grassland		0.092	Medium	Poor	Low	local strategy	distinctiveness habitat required	0.42		0.00 0.00 0.00	0.09	0.42	C .	success according of a stand stands	
72 Grassland	Grassiand - Other neutral grassiand	0.337	Medium	Moderate	Low	not in local strategy	distinctiveness habitat required	2.97	0.337	0.00 2.97 0.00	0.00	0.00	du	overns grassianu - steep siopes	
73 Grassland	Grassland - Other neutral grassland	0.438	Medium	Moderate	Low	Location ecologically desirable but not in local strategy	Same broad habitat or a higher distinctiveness habitat required	3.85	0.438	0.00 3.85 0.00	0.00	0.00	Go	overns grassland - steep slopes	
74 Grassland	Grassland - Other neutral grassland	0.627	Medium	Moderate	Low	Location ecologically desirable but not in local strategy	Same broad habitat or a higher distinctiveness habitat required	5.52	0.627	0.00 5.52 0.00	0.00	0.00	Go	overns grassland - steep slopes	
75 Grassland	Grassland - Other neutral grassland	0.502	Medium	Good	Low	Location ecologically desirable but not in local strategy	Same broad habitat or a higher distinctiveness habitat required	6.63	0.502	6.63 0.00 0.00	0.00	0.00	Go	overns grassland - steep slopes	
76 Grassland	Grassland - Other neutral grassland	0.702	Medium	Good	Low	Location ecologically desirable but not in local strategy	Same broad habitat or a higher distinctiveness habitat required	9.27	0.702	9.27 0.00 0.00	0.00	0.00	Go	overns grassland - steep slopes	
77 Grassland	Grassland - Other neutral grassland	0.43	Medium	Poor	Low	Within area formally identified in	Same broad habitat or a higher	1.98	0.43	0.00 1.98 0.00	0.00	0.00	La	ngarth marshy grassland	
78 Grassland	Grassland - Other neutral grassland	0.35	Medium	Popr	Low	Within area formally identified in	Same broad habitat or a higher	1.61	0.35	0.00 1.61 0.00	0.00	0.00	La	ngarth marshy grassland	
70 Creations	Grassland - Other neutral grassland	0.94	Madium	Madazata	1 mar	local strategy Location ecologically desirable but	distinctiveness habitat required Same broad habitat or a higher	7.20	0.84	0.00 7.20 0.00	0.00	0.00	La	ngarth marshy grassland	
R0 Creation	Grassland - Other neutral grassland	0.15	Madium	Dear	Law	not in local strategy Location ecologically desirable but	distinctiveness habitat required Same broad habitat or a higher	0.66	0.04	0.00 0.66 0.00	0.00	0.00	La	ngarth marshy grassland	
Grassiand	- Grassland - Other neutral grassland	0.13	wedium	FUOF	LOW	not in local strategy Location ecologically desirable but	distinctiveness habitat required Same broad habitat or a higher	0.00	0.15	0.00 0.00 0.00	0.00	0.00	M	aiden Green grassland on steep slopes	
Grassland	Grassland - Other neutral anarcland	U.432	medium	moderate	LOW	not in local strategy Location ecologically desirable but	distinctiveness habitat required Same broad habitat or a higher	3.80	0.432	3.80 0.00	0.00	0.00		aiden Green grassland on steep slopes	
82 Grassland	Graning - Other motors graning	0.239	Medium	Moderate	Low	not in local strategy	distinctiveness habitat required	2.10	0.239	0.00 2.10 0.00	0.00	0.00			
83 Grassland	Grassiania - Orner medical grassiania	0.0247	Medium	Poor	Low	strategy/ no local strategy	distinctiveness habitat required	0.10		0.00 0.00 0.00	0.02	0.10			
84 Grassland	Grassiand - Uther neutral grassiand	0.0399	Medium	Moderate	Low	not in local strategy	distinctiveness habitat required	0.35		0.00 0.00 0.00	0.04	0.35			
85 Lakes	Lakes - Ponds (Non- Priority Habitat)	0.0428	High	Poor	Medium	Within area formally identified in local strategy	Same habitat required	0.32	0.0428	0.00 0.32 0.00	0.00	0.00	wi	illow Green pond	
86 Lakes	Lakes - Ponds (Non- Priority Habitat)	0.25	High	Poor	Medium	Within area formally identified in local strategy	Same habitat required	1.90	0.25	0.00 1.90 0.00	0.00	0.00	La	ngarth fishing pond	
87 Lakes	Lakes - Ponds (Non- Priority Habitat)	0.04	High	Poor	Medium	Within area formally identified in local strategy	Same habitat required	0.30	0.04	0.00 0.30 0.00	0.00	0.00	La	ngarth pond	
88 Lakes	Lakes - Ponds (Non- Priority Habitat)	0.01	High	Poor	Medium	Location ecologically desirable but	Same habitat required	0.07	0.01	0.00 0.07 0.00	0.00	0.00	La	ngarth pond	
89 Lakes	Lakes - Ponds (Non- Priority Habitat)	0.0003	High	Poor	Medium	Location ecologically desirable but	Same habitat required	0.00		0.00 0.00 0.00	0.00	0.00			
90 Lakes	Lakes - Ponds (Non- Priority Habitat)	0.0007	High	Poor	Medium	Location ecologically desirable but	Same habitat required	0.01		0.00 0.00 0.00	0.00	0.01			
91 Woodland and format	Woodland and forest - Lowland mixed deciduous woodland	1 972	High	Moderato	Madium	not in local strategy Within area formally identified in	Same habitat required	20.05	1972	0.00 29.95 0.00	0.00	0.00	Go	overns Wood	
woodland and forest	Woodland and forest - Lowland mixed deciduous woodland	1.9/3	nign	mouerate	weulum	local strategy Location ecologically desirable but	Same hautat required	25.55	1.9/3	3.00 25.95 0.00	0.00	0.00			
92 Woodland and forest	Woodland and forest - Lowland mixed deciduous vesseliked	0.429	High	Good	Medium	not in local strategy Area/compensation not in local	Same habitat required	9.34	0.429	9.34 0.00 0.00	0.00	0.00			
93 Woodland and forest	Mendland and forest. Lowland mixed decideous v 19914	0.295	High	Moderate	Medium	strategy/ no local strategy	Same habitat required	3.89	0.295	0.00 3.89 0.00	0.00	0.00			
94 Woodland and forest	wooulana and iorest - Lowiand mixed deciduous woodland	0.199	High	Moderate	Medium	local strategy	Same habitat required	3.02	0.199	0.00 3.02 0.00	0.00	0.00			
95 Woodland and forest	Woodland and forest - Lowland mixed deciduous woodland	0.382	High	Good	Medium	Location ecologically desirable but not in local strategy	Same habitat required	8.32	0.382	8.32 0.00 0.00	0.00	0.00			
96 Woodland and forest	Woodland and forest - Lowland mixed deciduous woodland	0.034	High	Moderate	Medium	Location ecologically desirable but not in local strategy	Same habitat required	0.49	0.034	0.00 0.49 0.00	0.00	0.00			
97 Woodland and forest	Woodland and forest - Lowland mixed deciduous woodland	0.218	High	Moderate	Medium	Within area formally identified in local strategy	Same habitat required	3.31	0.218	0.00 3.31 0.00	0.00	0.00			

98 Woodland and forest	Woodland and forest - Lowland mixed deciduous woodland	1.04	High	Moderate	Medium	Within area formally identified in	Same habitat required	15.79		1.04	0.00	15.79 0.00	0.00	0.00			
99 Woodland and forest	Woodland and forest - Lowland mixed deciduous woodland	0.66	High	Poor	Medium	Within area formally identified in	Same habitat required	5.01		0.66	0.00	5.01 0.00	0.00	0.00	w	/oodland round fishing ponds, large amounts of Rhod.	
100 Woodland and forest	Woodland and forest - Lowland mixed deciduous woodland	0.12	High	Moderate	Madium	Location ecologically desirable but	Same habitat required	1.74		0.06	0.00	0.87 0.00	0.06	0.97	So	ome loss to NAR	
100 Woodiand and iorest	Woodland and forest - Lowland mixed deciduous woodland	0.12	nigi	Moderate	Medium	not in local strategy Location ecologically desirable but	Same nabitat required	1.74		0.00	0.00	0.87 0.00	0.00	0.87			
101 Woodland and forest	Wandand and family 1 miland minut devidence considered	0.13	High	Moderate	Medium	not in local strategy	Same habitat required	1.89		0.13	0.00	1.89 0.00	0.00	0.00			
102 Woodland and forest	woodiand and rorest - cowand mixed decideds woodiand	0.14	High	Poor	Medium	strategy/ no local strategy	Same habitat required	0.92		0.14	0.00	0.92 0.00	0.00	0.00			
103 Woodland and forest	Woodland and forest - Lowland mixed deciduous woodland	0.18	High	Good	Medium	Area/compensation not in local strategy/ no local strategy	Same habitat required	3.56	0.18		3.56	0.00 0.00	0.00	0.00			
104 Woodland and forest	Woodland and forest - Lowland mixed deciduous woodland	0.06	High	Moderate	Medium	Within area formally identified in local strategy	Same habitat required	0.91		0.06	0.00	0.91 0.00	0.00	0.00			
105 Woodland and forest	Woodland and forest - Lowland mixed deciduous woodland	0.288	High	Moderate	Medium	Location ecologically desirable but not in local strategy	Same habitat required	4.18		0.288	0.00	4.18 0.00	0.00	0.00			
106 Woodland and forest	Woodland and forest - Lowland mixed deciduous woodland	0.1141	High	Poor	Medium	Area/compensation not in local	Same habitat required	0.75			0.00	0.00 0.00	0.11	0.75			
107 Woodland and forest	Woodland and forest - Lowland mixed deciduous woodland	0.0879	High	Poor	Medium	Area/compensation not in local	Same habitat required	0.58		0.0879	0.00	0.58 0.00	0.00	0.00			
108 Woodland and forest	Woodland and forest - Lowland mixed deciduous woodland	0.0473	High	Moderate	Medium	Area/compensation not in local	Same babitat required	0.62		0.0473	0.00	0.62 0.00	0.00	0.00			
	Woodland and forest - Lowland mixed deciduous woodland					strategy/ no local strategy Area/compensation not in local	and the later of t	0.00			0.00						
109 Woodiand and iorest	Heathland and shrub - Mixed scrub	0.0442	nigi	Moderate	iweuluiii	strategy/ no local strategy Location ecologically desirable but	Same broad habitat or a higher	0.38		0.0442	0.00	0.58 0.00	0.00	0.00			
110 Heathland and shrub	Heathland and shrub - Mixed scrub	0.219	Medium	Moderate	Low	not in local strategy Area/compensation not in local	distinctiveness habitat required	1.93	_	0.219	0.00	1.93 0.00	0.00	0.00			
111 Heathland and shrub		0.022	Medium	Moderate	Low	strategy/ no local strategy	distinctiveness habitat required	0.18		0.022	0.00	0.18 0.00	0.00	0.00			
112 Heathland and shrub	Heatniand and shrub - Mixed scrub	0.015	Medium	Moderate	Low	local strategy	distinctiveness habitat required	0.14			0.00	0.00 0.00	0.02	0.14			
113 Heathland and shrub	Heathland and shrub - Mixed scrub	0.083	Medium	Good	Low	Within area formally identified in local strategy	Same broad habitat or a higher distinctiveness habitat required	1.15	0.083		1.15	0.00 0.00	0.00	0.00			
114 Heathland and shrub	Heathland and shrub - Mixed scrub	0.16	Medium	Poor	Low	Area/compensation not in local strategy/ no local strategy	Same broad habitat or a higher distinctiveness habitat required	0.64		0.192	0.00	0.77 0.00	Error in Areas	Error in Areas			
115 Heathland and shrub	Heathland and shrub - Mixed scrub	0.051	Medium	Poor	Low	Location ecologically desirable but not in local strategy	Same broad habitat or a higher distinctiveness habitat or a	0.22		0.051	0.00	0.22 0.00	0.00	0.00			
116 Heathland and shrub	Heathland and shrub - Mixed scrub	0.072	Medium	Poor	Low	Location ecologically desirable but	Same broad habitat or a higher	0.32		0.072	0.00	0.32 0.00	0.00	0.00			
117 Heatbland and shrub	Heathland and shrub - Mixed scrub	0.1901	Medium	Poor	Low	Area/compensation not in local	Same broad habitat or a higher	0.76		0.1901	0.00	0.76 0.00	0.00	0.00			
118 Heathland and all the	Heathland and shrub - Mixed scrub	0.0429	Madium	Pere		strategy/ no local strategy Area/compensation not in local	distinctiveness habitat required Same broad habitat or a higher	0.19		0.0420	0.00	0.18 0.00	0.00	0.00			
Heatmand and shrub	Heathland and shrub - Mixed scrub	0.0458	weulum	POOI	LOW	strategy/ no local strategy Area/compensation not in local	distinctiveness habitat required Same broad habitat or a higher	0.10		0.0458	0.00	0.18 0.00	0.00	0.00			
119 Heathland and shrub	Constated and Durlard Makamaral	0.0089	Medium	Moderate	Low	strategy/ no local strategy	distinctiveness habitat required	0.07		0.0089	0.00	0.07 0.00	0.00	0.00			
120 Sparsely vegetated land	Sparsery vegecated rand - Roberty Epitemeteral	0.0064	Low	Poor	Low	strategy/ no local strategy	habitat required	0.01			0.00	0.00 0.00	0.01	0.01			
121 Sparsely vegetated land	Sparsely vegetated land - Ruderal/Ephemeral	0.01	Low	Poor	Low	Area/compensation not in local strategy/ no local strategy	Same distinctiveness or better habitat required	0.02			0.00	0.00 0.00	0.01	0.02			
122 Sparsely vegetated land	Sparsely vegetated land - Ruderal/Ephemeral	0.48	Low	Poor	Low	Area/compensation not in local strategy/ no local strategy	Same distinctiveness or better habitat required	0.96			0.00	0.00 0.00	0.48	0.96			
123 Urban	Urban - Amenity grassland	1.2569	Low	Poor	Low	Area/compensation not in local strategy/ no local strategy	Same distinctiveness or better habitat required	2.51			0.00	0.00 0.00	1.26	2.51	w	/illow Green amenity grassland	
124 Urban	Urban - Amenity grassland	0.091	Low	Poor	Low	Area/compensation not in local	Same distinctiveness or better	0.18			0.00	0.00 0.00	0.09	0.18	Go	overns amenity grassland	
125 Urban	Urban - Amenity grassland	1.1	Law	Poor	Low	Location ecologically desirable but	Same distinctiveness or better	2.42			0.00	0.00 0.00	1.10	2.42	La	angarth amenity grassland	
126 Urban	Urban - Amenity grassland	0.279	Law	Poor	low	Area/compensation not in local	Same distinctiveness or better	0.56			0.00	0.00 0.00	0.28	0.56	м	taiden Green amenity grasslan	
100 01001	Urban - Artificial unvegetated, unsealed surface	0175		1001		strategy/ no local strategy Location ecologically desirable but	habitat required	0.30			0.00	0.00 0.00	0.10	0.50			
127 Urban	Linhan - Developed land: sealed surface	1.32	V.Low	N/A - Other	Low	not in local strategy Area/compensation not in local	Compensation Not Required	0.00	_		0.00	0.00 0.00	1.32	0.00			
128 Urban	orban - beveloped when sense and acce	0.606	V.Low	N/A - Other	Low	strategy/ no local strategy	Compensation Not Required	0.00			0.00	0.00 0.00	0.61	0.00			
129 Urban	Urban - Introduced shrub	0.05	Low	Poor	Low	local strategy	habitat required	0.12			0.00	0.00 0.00	0.05	0.12			
130 Urban	Urban - Introduced shrub	0.29	Low	Poor	Low	Area/compensation not in local strategy/ no local strategy	Same distinctiveness or better habitat required	0.58			0.00	0.00 0.00	0.29	0.58			
131 Urban	Urban - Vacant/derelict land/ bareground	0.0181	Low	Poor	Low	Area/compensation not in local strategy/ no local strategy	Same distinctiveness or better habitat required	0.04			0.00	0.00 0.00	0.02	0.04			
132 Urban	Urban - Vacant/derelict land/ bareground	0.03	Low	Poor	Low	Within area formally identified in	Same distinctiveness or better	0.07			0.00	0.00 0.00	0.03	0.07			
133 Woodland and forest	Woodland and forest - Wet woodland	0.666	High	Moderate	Medium	Location ecologically desirable but	Same habitat required	9.67		0.666	0.00	9.67 0.00	0.00	0.00			
134 Woodland and forest	Woodland and forest - Wet woodland	1.75	High	Moderate	Medium	Within area formally identified in	Same babitat required	26.57		1.75	0.00	26.57 0.00	0.00	0.00			
13F Weedland and ferent	Woodland and forest - Wet woodland	1.04	link	Madamia	Madium	local strategy Location ecologically desirable but	Come babitat energiand	15.10		1.04	0.00	15.10 0.00	0.00	0.00			
155 Woodiand and iorest	Woodland and forest - Wet woodland	1.04	nigii	Moderate	Medium	not in local strategy Within area formally identified in	same natitat required	13.10		1.04	0.00	13.10 0.00	0.00	0.00			
136 Woodland and forest	Wandland and facest. Makusedland	0.44	High	Moderate	Medium	local strategy Within area formally identified in	Same habitat required	6.68		0.44	 0.00	6.68 0.00	0.00	0.00			
137 Woodland and forest	woodaard and forest - wet woodaard	0.04	High	Moderate	Medium	local strategy	Same habitat required	0.61		0.04	0.00	0.61 0.00	0.00	0.00			
138 Woodland and forest	Woodland and forest - Wet woodland	0.075	High	Moderate	Medium	not in local strategy	Same habitat required	1.09		0.075	0.00	1.09 0.00	0.00	0.00			
139 Woodland and forest	Woodland and forest - Wet woodland	0.376	High	Moderate	Medium	Location ecologically desirable but not in local strategy	Same habitat required	5.46		0.376	0.00	5.46 0.00	0.00	0.00			
140 Woodland and forest	Woodland and forest - Wet woodland	0.486	High	Moderate	Medium	Location ecologically desirable but not in local strategy	Same habitat required	7.06		0.486	0.00	7.06 0.00	0.00	0.00	ΙT		
141 Woodland and forest	Woodland and forest - Wet woodland	0.187	High	Moderate	Medium	Location ecologically desirable but not in local strategy	Same habitat required	2.72		0.187	0.00	2.72 0.00	0.00	0.00			
142 Woodland and forest	Woodland and forest - Wet woodland	0.476	High	Moderate	Medium	Within area formally identified in	Same habitat required	7.23		0.476	0.00	7.23 0.00	0.00	0.00			
143 Woodland and forest	Woodland and forest - Wet woodland	0.1	High	Poor	Medium	Area/compensation not in local	Same babitat required	0.66		0.1	0.00	0.66 0.00	0.00	0.00			
144 Woodland and 6	Woodiand and forest - Wet woodiand	0.401	Link.	Moderate	Madium	strategy/ no local strategy Location ecologically desirable but	Same habitat environt	7.12		0.401	0.00	7.12 0.00	0.00	0.00			
woodiand and forest	Heathland and shrub - Bramble scrub	0.451	nign	Moderace	wiedium	not in local strategy Area/compensation not in local	Same broad habitat or a higher	7.15		0.451	0.00	.13 0.00	0.00	0.00	C.	ormac compound	
145 Heathland and shrub	Linkson Developed lands control surface	0.6444	Medium	Moderate	Low	strategy/ no local strategy	distinctiveness habitat required	5.16			0.00	0.00 0.00	0.64	5.16	-	erence research building	
146 Urban	Chart-Developed who, search and service	0.0195	V.Low	N/A - Other	Low	strategy/ no local strategy	Compensation Not Required	0.00			0.00	0.00 0.00	0.02	0.00			
147 Urban	Urban - Developed land; sealed surface	0.1845	V.Low	N/A - Other	Low	Area/compensation not in local strategy/ no local strategy	Compensation Not Required	0.00			0.00	0.00 0.00	0.18	0.00	Cc	ormaccompound car park	
148 Urban	Urban - Developed land; sealed surface	3.9135	V.Low	N/A - Other	Low	Area/compensation not in local strategy/ no local strategy	Compensation Not Required	0.00	3.9135		0.00	0.00 0.00	0.00	0.00	P8	&R roads	
149 Woodland and forest	Woodland and forest - Lowland mixed deciduous woodland	1.248	High	Moderate	Medium	Area/compensation not in local strategy/ no local strategy	Same habitat required	16.47		1.248	0.00	16.47 0.00	0.00	0.00	P8	&R woodland planting	
150 Urban	Urban - Developed land; sealed surface	0.0286	V.Low	N/A - Other	Low	Area/compensation not in local strategy/ no local strategy	Compensation Not Required	0.00	0.0286		0.00	0.00 0.00	0.00	0.00	P8	&R building	
151 Heathland and shrub	Heathland and shrub - Mixed scrub	0.5803	Medium	Moderate	Low	Area/compensation not in local	Same broad habitat or a higher	4.64		0.5803	0.00	4.64 0.00	0.00	0.00	P8	&R shrub planting	
152 Sparsely vegetated land	Sparsely vegetated land - Ruderal/Ephemeral	01	Law	Poor	Low	Area/compensation not in local	Same distinctiveness or better	0.20	0.7		0.20	0.00 0.00	0.00	0.00	P8	&R floors of catchment ponds	
152	Urban - Introduced shrub	1 1164	1000	Pere		strategy/ no local strategy Area/compensation not in local	habitat required Same distinctiveness or better	2.22	0.1	1,1164	0.00	2 22 0.00	0.00	0.00	P8	&R intro shrubs	
155 Urban	Woodland and forest - Lowland mixed deviduous woodland	1.1164	LOW	Poor	LOW	strategy/ no local strategy Area/compensation not in local	habitat required	2.23		1.1104	0.00	2.23 0.00	0.00	0.00			
154 Woodland and forest	woosana and lorest - cowiand mixed deciduous woodland	0.5704	High	Moderate	Low	strategy/ no local strategy	Same habitat required	6.84		0.5704	0.00	6.84 0.00	0.00	0.00			

155	Grassland	Grassland - Other neutral grassland	2.5194	Medium	Fairly Poor	Low	Area/compensation not in local	Same broad habitat or a higher	15.12		2.5194	0.00 1	5.12 0	.00 0.0	0.00			
156	Heathland and shrub	Heathland and shrub - Bramble scrub	0.5283	Medium	Moderate	Low	Area/compensation not in local	Same broad habitat or a higher	4.23		0.5283	0.00	4.23 0	.00 0.0	0.00			
157	Urban	Urban - Amenity grassland	0.4581	Low	Poor	Low	Area/compensation not in local	Same distinctiveness or better	0.92	-		0.00	0.00 0	.00 0.4	0.92			
158	Urban	Urban - Developed land; sealed surface	1.1696	V.Low	N/A - Other	Low	Area/compensation not in local	Compensation Not Required	0.00			0.00	0.00 C	.00 1.1	0.00			
159	Urban	Urban - Developed land; sealed surface	0.2866	V.Low	N/A - Other	Low	Area/compensation not in local	Compensation Not Required	0.00	-		0.00	0.00 0	.00 0.2	0.00			
160	Woodland and forest	Woodland and forest - Other coniferous woodland	0.3367	Low	Poor	Low	Area/compensation not in local	Same distinctiveness or better	0.67	-		0.00	0.00 0	.00 0.3	0.67			
161	Heathland and shrub	Heathland and shrub - Gorse scrub	1.2796	Medium	Poor	Low	Area/compensation not in local	Same broad habitat or a higher	5.12	-		0.00	0.00 0	.00 1.2	5.12			
162	Urban	Urban - Amenity grassland	0.2159	Low	Poor	Low	Area/compensation not in local	Same distinctiveness or better	0.43	-		0.00	0.00 0	.00 0.2	0.43			
163	Urban	Urban - Developed land; sealed surface	0.0158	V.Low	N/A - Other	Low	Area/compensation not in local	Compensation Not Required	0.00			0.00	0.00 C	.00 0.0	0.00			
164	Urban	Urban - Developed land; sealed surface	20.13	V.Low	N/A - Other	Low	Area/compensation not in local	Compensation Not Required	0.00			0.00	0.00 C	.00 20.	3 0.00		Roads, pavements, car parks etc	
165							strategy/ no local strategy											
166										-								
167										-								
169																		
170										-		 	_		_			
171										-			_				-	
173																		
174												 						
175													_		_		-	
177										-								
178																		
179										-		 	_		_			
180										-					_			
182																		
183												 	_		_	_		
184										-			_				-	
186																		
187												 			_			
188										-					_			
190																		
191												 			_			
192										-		 			_	-		
194																		
195																		
196										-								
198																		
199												 						
200										-					_			
202																		
203																		
204										-			_		_		-	
205										-								
207																		
208										-			_					
209										H							1	
211																		
212																		

Langarth Garden Village										
A-2 Site Habitat Creation										
Condense / Show Columns Co	indense / Show	Rows	1							
Main Menu	Instruction	s.								
	matruction.	Post developm	ent/ post inter	vention habitats				1		
				Ecological	Strategic significance	Temporal multiplier	Difficulty		Co	mments
Proposed habitat	(hectares)	Distinctiveness	Condition	Ecological	Strategic significance	Time to target	Difficulty of creation	delivered	Assessor comments	Reviewer comments
Urban - Allotments	1.035	Madium	Eaistu Deer	low	Location ecologically desirable but not in	t t	category	6.62	Communal gardens/semi natural/allotments in	
Urban - Vegetated garden	1.025	Medium	Fairly Poor	LOW	local strategy Location ecologically desirable but not in	1	LOW	0.53	residential areas	
Urban - Un versetated same	0.969	Low	Poor	Low	local strategy	1	Low	2.06	50% Private gardens in residential plots	
Union - Onvegetated galactic	0.969	V.Low	N/A - Other	Low	local strategy	0	Low	0.00	50% Private gardens in residential plots	
Urban - Developed land; sealed surface	1.883	V.Low	N/A - Other	Low	Location ecologically desirable but not in local strategy	0	Low	0.00	Built areas in residential plots	
Urban - Allotments	22.173	Medium	Fairly Poor	Medium	Area/compensation not in local strategy/ no local strategy	1	Low	141.22	communal gardens/semi natural/allotments in residential areas	
Urban - Vegetated garden	6.298	Low	Poor	Medium	Area/compensation not in local strategy/ no local strategy	1	Low	13.37	50% Private gardens in residential plots	
Urban - Un-vegetated garden	6.298	V.Low	N/A - Other	Low	Area/compensation not in local strategy/ no local strategy	0	Low	0.00	50% Private gardens in residential plots	
Urban - Developed land; sealed surface	38.105	V.Low	N/A - Other	Low	Area/compensation not in local strategy/	0	Low	0.00	Built areas in residential plots	
Urban - Introduced shrub	0.069	Low	Poor	Low	Within area formally identified in local	1	Low	0.15	Non-residential plots assumed 90% built,10%	
Urban - Developed land; sealed surface	0.622	V.Low	N/A - Other	Low	Strategy Within area formally identified in local	0	Low	0.00	Non-residential plots assumed 90% built,10%	
Urban - Introduced shrub	0.28	Low	Roor	Low	strategy Location ecologically desirable but not in	1	Low	0.59	soft landscaping Non-residential plots assumed 90% built,10%	
Urban - Developed land; sealed surface	3.63	Milan	N/A Other	Low	local strategy Location ecologically desirable but not in	-	Low	0.00	soft landscaping Non-residential plots assumed 90% built,10%	
Urban - Introduced shrub	2.52	v.cow	N/A - Otilei	LOW	local strategy Area/compensation not in local strategy/		LOW	0.00	soft landscaping Non-residential plots assumed 90% built,10%	
Lickan Developed lands coaled surface	1.452	Low	Poor	Low	no local strategy	1	Low	2.80	soft landscaping	
Conclude Other souther and state	13.085	V.Low	N/A - Other	Low	no local strategy	0	Low	0.00	soft landscaping	
Grassiano - Other neutral grassiano	2.495	Medium	Moderate	Low	no local strategy	10	Low	13.98	Governs Farm	
Heathland and shrub - Mixed scrub	0.801	Medium	Moderate	Low	Within area formally identified in local strategy	3	Low	6.62	Hedge enhancement	
Heathland and shrub - Mixed scrub	1.602	Medium	Moderate	Low	Location ecologically desirable but not in local strategy	3	Low	12.67	Hedge enhancement	
Heathland and shrub - Mixed scrub	5.606	Medium	Moderate	Low	Area/compensation not in local strategy/ no local strategy	3	Low	40.30	Hedge enhancement	
Grassland - Other neutral grassland	1.678	Medium	Fairly Poor	Low	Area/compensation not in local strategy/	5	Low	8.43	Read verses along NAR 8, 420	
Urban - Developed land; sealed surface	0.359	V.Low	N/A - Other	Low	Area/compensation not in local strategy/	0	Low	0.00	Noad verges along livin & ASS	
Urban - Amenity grassland	0.359	Low	Fairly Poor	Low	no local strategy Area/compensation not in local strategy/	2	Low	1.00	riay area - sarety surface	
Heathland and shrub - Mixed scrub	0.142	Madium	Moderate	1000	no local strategy Area/compensation not in local strategy/	,	low	1.02	Play area - grass	
Grassland - Other neutral grassland	0.142	weatum	mouerate	LOW .	no local strategy Within area formally identified in local	3	LOW	1.02	New hedges	
- Graceland - Other neutral graceland	6.673	Medium	Moderate	Low	strategy	10	Low	42.99	Natural open space	
Grassianu - Other neutral grassianu	1.122	Medium	Moderate	Low	local strategy	10	Low	6.91	Natural open space	
Grassland - Other neutral grassland	3.433	Medium	Moderate	Low	Area/compensation not in local strategy/ no local strategy	10	Low	19.23	Natural open space	
Grassland - Modified grassland	0.996	Low	Moderate	Low	Location ecologically desirable but not in local strategy	10	Low	3.07	Parks	
Grassland - Modified grassland	8.961	Low	Moderate	Low	Area/compensation not in local strategy/ no local strategy	10	Low	25.10	Parks	
Urban - Street Tree	0.858	Low	Moderate	Low	Area/compensation not in local strategy/ no local strategy	27	Low	1.31	Street trees along NAR & A39	
Woodland and forest - Lowland mixed deciduous woodland	3.362	High	Moderate	Medium	Within area formally identified in local	32+	High	5.39	Encert on northern edge of Country	
Woodland and forest - Lowland mixed deciduous woodland	0.821	High	Moderate	Medium	Location ecologically desirable but not in	32+	High	1.26	Forest at Governs & MG in desirable strategy	
Woodland and forest - Lowland mixed deciduous woodland	0.893	High	Moderate	Madium	local strategy Area/compensation not in local strategy/	274	High	1.74	area.	
Urban - Sustainable urban drainage feature	4.2		Mederate		no local strategy Within area formally identified in local		t to be	2.00	Forest at Governs but outide strategy areas	
Urban - Sustainable urban drainage feature	1.3	LOW	Moderate	LOW	strategy Location ecologically desirable but not in	3	Medium	3.00	SuDS basins	
IIrhan - Sustainable urhan drainane feature	1.967	Low	Moderate	Low	local strategy	3	Medium	5.21	SuDS basins	
IIrhan - Sustainable urhan drainane feature	2.734	Low	Moderate	Low	no local strategy	3	Medium	6.58	SuDS basins	
	0.126	Low	Moderate	Low	no local strategy	3	Medium	0.30	NAR SuDS basin	
Urban - Artificial lake or pond	0.363	Medium	Moderate	Low	Within area formally identified in local strategy	3	Low	3.00	NAR SuDS wetpond	
Urban - Artificial lake or pond	0.065	Medium	Moderate	Low	Location ecologically desirable but not in local strategy	3	Low	0.51	NAR SuDS wetpond	
Urban - Artificial lake or pond	0.034	Medium	Moderate	Low	Area/compensation not in local strategy/ no local strategy	3	Low	0.24	NAR SuDS wetpond	
Urban - Bioswale	1.06	Low	Moderate	Low	Within area formally identified in local strategy	1	Medium	3.15	SuDS swales	
Urban - Bioswale	1.05	Low	Moderate	Low	Location ecologically desirable but not in	1	Medium	2.99	SuDS swales	
Urban - Bioswale	9.314	Low	Moderate	Low	Area/compensation not in local strategy/	1	Medium	24.09	SuDS surgles	
Urban - Allotments	0.838	Medium	Fairly Poor	Low	Location ecologically desirable but not in	1	Low	5.34	Split residential plots - communal	
Urban - Vegetated garden	0.300	Low	Peer	Low	local strategy Location ecologically desirable but not in	-	Low	0.63	gardens/allotments etc	
Urban - Un-vegetated garden	0.255	LOW	POOI	Low	local strategy Location ecologically desirable but not in		Low	0.03	Split residential plots - 50% private gardens	
Urban - Developed land: sealed surface	0.299	V.LOW	N/A - Other	LOW	local strategy	U	LOW	0.00	Split residential plots - 50% private gardens	
Lidean Alletmonts	0.957	V.Low	N/A - Other	Low	local strategy	0	Low	0.00	Split residential plots - built areas	
orban - Anothenits	2.672	Medium	Fairly Poor	Low	no local strategy	1	Low	15.47	gardens/allotments etc	
Urban - Vegetated garden	0.85	Low	Poor	Low	Area/compensation not in local strategy/ no local strategy	1	Low	1.64	Split residential plots - 50% private gardens	
Urban - Un-vegetated garden	0.85	V.Low	N/A - Other	Low	Area/compensation not in local strategy/ no local strategy	0	Low	0.00	Split residential plots - 50% private gardens	
Urban - Developed land; sealed surface	3.363	V.Low	N/A - Other	Low	Area/compensation not in local strategy/ no local strategy	0	Low	0.00	Split residential plots - built areas	
Urban - Allotments	0.603	Medium	Fairly Poor	Low	Within area formally identified in local strategy	1	Low	4.02		
Urban - Allotments	0.082	Medium	Fairly Poor	Low	Location ecologically desirable but not in	1	Low	0.52		
Heathland and shrub - Mixed scrub	14.765	Medium	Fairly Poor	Low	Area/compensation not in local strategy/	2	Low	82.50	Evicting hadron 8. *	
Urban - Developed land; sealed surface	27.78	V.Low	N/A - Other	N/A	no local strategy Area/compensation not in local strategy/	0	Low	0.00	causing neages & trees	
					no local strategy	-				
	-									
-										

A-3 Site Habitat Enhancement Condense / Show Columns Condense / Show Rows														
Main Menu Instructions				Poit development/ poit war	weekson hated	1911								
	Baseline habitats	Charge in dist	activeness and condition		//13	Distinctiveness	Condition	Ecological connectivity Ecological	Strategic significance	Temporal multiplier	Diffculty multipliers Diffculty of	Habitat units	Comments	
net 7	Baseline habitat	Proposed national (Pre-populated but can be overridden)	Distinctiveness charge	Condition change	(and the l	-	100 m	connectivity score	Strategic significance	condition/years	enhancement category		Assessor comments	Reviewer comments
21	Heathland and shrub - Bramble scrub Heathland and shrub - Bramble scrub	Heathland and shrub - Bramble scrub Heathland and shrub - Bramble scrub	Medium - Medium Medium - Medium	Moderate - Fairly Good Poor - Fairly Good	0.109	Medium	Fairly Good	LDW	in local strategy Location ecologically desirable but not in local strategy	1	LOW	0.54		
22 23	Heathland and shrub - Bramble scrub	Heathland and shrub - Bramble scrub	Medium - Medium	Poor - Fairly Good	0.496	Medium	Fairly Good	Low	Location ecologically desirable but not in local strategy Location ecologically desirable but not	3	Low	5.12		
25	Heathland and shrub - Gorse sorub	Heathland and shrub - Gorse scrub	Medium - Medium	Moderate - Fairty Good	0.068	Medium	Fairly Good	LOW	in local strateev Location ecologically desirable but not in local strateev	2	Low	674		
26 31	Heathland and shrub - Gonse sorub Grassland - Modified grassland	Heathland and shrub - Gonse scrub Grassland - Other neutral grassland	Medium - Medium	Poor - Moderate Lower Distinctiveness Habitat -	0.041 2.872	Medium Medium	Moderate Moderate	LOW	in local strategy Location ecologically desirable but not Location ecologically desirable but not	5	Low	0.33	former manifest	
22	Grassland - Modified grassland	Grassland - Other neutral grassland	Low - Medium	Adodenska Lower Distinctiveness Habitat - Adodenska	\$714	Medium	Moderate	LOW	In local streams Location ecologically desirable but not in local streams	10	Low	28.98	Grame environ	
69 72	Grassland - Other neutral grassland Grassland - Other neutral grassland	Grassland - Other neutral grassland Grassland - Other neutral grassland	Medium - Medium Medium - Medium	Poor - Fairly Good Moderate - Fairly Good	0.026	Medium	Fairly Good	LOW	in local strategy Location ecologically desirable but not in local strategy	12	Low	023	Governs eramland	
23	Grassland - Other neutral grassland	Grassland - Other neutral grassland	Medium - Medium	Moderate - Fairly Good	0.438	Medium	Fairly Good	LOW	Within area formally identified in local strategy Within was formally identified in local	10	Low	474		
74	Grassland - Other neutral grassland Grassland - Other neutral grassland	Grassland - Other neutral grassland Grassland - Other neutral grassland	Medium - Medium Medium - Medium	Moderate - Fairly Good Poor - Fairly Good	0.627	Medium	Fairly Good	LOW	Itrateev Location ecologically desirable but not in local strateev	10	Low	678 274	Langarth marshy erassland	
78	Grassland - Other neutral grassland	Grassland - Other neutral grassland	Medium - Medium	Poor - Fairly Good	0.35	Medium	Fairly Good	LOW	Location ecologically desirable but not in local strategy location ecologically desirable but not	12	Low	2.05	Lanearth marshy exaculand	
80	Grassland - Other neutral grassland	Grassland - Other neutral grassland	Medium - Medium	Poor - Moderate	0.15	Medium	Moderate	LOW	in local strategy Location ecologically desirable but not in local strategy	10	Low	112	Lanearth marshy executed	
81 82	Grassland - Other neutral grassland	Grassland - Other neutral grassland	Medium - Medium	Moderate - Fairly Good	0.432	Medium	Fairly Good	Medum	Area/compensation not in local strategy/ no local strategy Within area formally identified in local	10	Low	4.47	Maiden Green steep slope erassland	
65	Lakes - Fonds (Non-Priority Habitat)	Lakes - Ponds (Non-Priority Habitat)	High - High	Poor - Moderate	0.0428	High	Moderate	Medium	strateev Within area formally identified in local strateau	2	Low	663	Maiden Green steep slope arassland	
86 87	Lakes - Ponds (Non-Priority Habitat) Lakes - Ponds (Non-Priority Habitat)	Lakes - Ponds (Non-Priority Habitat) Lakes - Ponds (Non-Priority Habitat)	High - High	Poor - Moderate Poor - Moderate	0.25	нар. Нар.	Moderate Moderate	Medium	Within area formally identified in local strategy Within area formally identified in local	2	LOW	266	Lanearth fishing sond	
	Lakes - Ponds (Non-Priority Habitat)	Lakes - Ponds (Non-Priority Habitat)	High - High	Poor - Moderate	0.01	High	Moderate	Medium	trateev Area/compensation not in local strateev/ no local strateev	2	Low	0.13	Lanzarth sond	
91 93	Woodland and forest - Lowland mixed deciduous woodland Woodland and forest - Lowland mixed deciduous woodland	Woodland and forest - Lowland mixed deciduous woodland Woodland and forest - Lowland mixed deciduous woodland	High - High High - High	Moderate - Fairly Good Moderate - Fairly Good	1.972	нір. Нір.	Fairly Good	Medium	In a sea tormany identified in local trategy Location ecologically desirable but not in local strategy	20	High High	31.16		
94 95	Woodland and forest - Lowland mixed deciduous woodland	Woodland and forest - Lowland mixed deciduous woodland	High - High	Moderate - Fairty Good	0.199	High	Fairly Good	Medium	Location ecologically desirable but not in local strateev Within area formally identified in local	20	High	3.01		
97	Woodland and forest - Lowland mixed deciduous woodland Woodland and forest - Lowland mixed deciduous woodland	Woodland and forest - Lowland mixed deciduous woodland Woodland and forest - Lowland mixed deciduous woodland	High - High	Moderate - Fairly Good	0.218	High High	Fairly Good	Medum	strategy Within area formally identified in local strategy	20	Hgh	2.44		
98 99	Woodland and forest - Lowland mixed deciduous woodland Woodland and forest - Lowland mixed deciduous woodland	Woodland and forest - Lowland mixed deciduous woodland Woodland and forest - Lowland mixed deciduous woodland	High - High	Moderate - Fairly Good	1.04	High	Fairly Good	Medium	Location ecologically desirable but not in local strateev Location ecologically desirable but not	20	Hab	1571		
100	Woodland and forest - Lowland mixed deciduous woodland	Woodland and forest - Lowland mixed deciduous woodland	High - High	Moderate - Fairty Good	0.06	High	Fairly Good	Medium	Area/compensation not in local annutate if no local strutume	20	High	0.82		
921	Woodland and forest - Lowland mixed deciduous woodland Woodland and forest - Lowland mixed deciduous woodland	Woodland and forest - Lowland mixed deciduous woodland Woodland and forest - Lowland mixed deciduous woodland	High - High High - High	Moderate - Fairly Good Poor - Moderate	0.12	High High	Fairly Good Moderate	Medium	Location ecologically desirable but not	20	High High	2.05		
134	Woodland and forest - Lowland mixed deciduous woodland	Woodland and forest - Lowland mixed deciduous woodland	High - High	Moderate - Fairly Good	0.06	High	Fairly Good	Medium	Area/compensation not in local strategy/ no local strategy Area/compensation net in local	20	High	0.82		
107	Woodland and forest - Lowland mixed deciduous woodland Woodland and forest - Lowland mixed deciduous woodland	w-wittand and torest - Lowland mixed deciduous woodland Woodland and forest - Lowland mixed deciduous woodland	High - High	Moderate - Fairly Good Poor - Moderate	0.288	High High	Harry Good	Medium	states/ no local statesy Area/compensation not in local statesy/ no local statesy	10	High	0.50		
108	Woodland and forest - Lowland mixed deciduous woodland Woodland and forest - Inveloved mixed deciduous and	Woodland and forest - Lowland mixed deciduous woodland	High - High	Moderate - Fairly Good	0.0473	High	Fairly Good	Low	Location ecologically desirable but not in local strategy Area/compensation not in local	20	High	045		
110	Heathland and shrub - Mixed scrub	Heathland and shrub - Mixed scrub	Medium - Medium	Moderate - Fairly Good	0.219	Medium	Fairly Good	LOW	strateev/ no local strateev Area/compensation not in local strateev/ no local strateev	2	Low	2.16		
111	Heathland and shrub - Mixed scrub Heathland and shrub - Mixed scrub	Heathland and shrub - Mixed scrub Heathland and shrub - Mixed scrub	Medium - Medium Medium - Medium	Moderate - Fairly Good Poor - Moderate	0.022	Medium	Fairly Good Moderate	Low	In local strateev Location ecologically desirable but not	2	Low	0.24		
115	Heathland and shrub - Mixed scrub	Heathland and shrub - Mixed scrub	Medium - Medium	Poor - Moderate	0.051	Medium	Moderate	LOW	++ Iocal Itrasev Area/compensation not in local strateev/ no local strateev Area/compensation not in local	\$	Low	6.37		
116	Heathland and shrub - Mixed scrub Heathland and shrub - Mixed scrub	Heathland and shrub - Miked scrub	Medium - Medium	Poor - Moderate Poor - Moderate	0.072	Medium	Moderate Moderate	LOW	Area/compensation not in local strategy (on local strategy Area/compensation not in local strategy (on local strategy	5	Low	0.53 1.40		
118	Heathland and shrub - Mixed scrub	Heathland and shrub - Mixed scrub	Medium - Medium	Poor - Moderate	0.0438	Medium	Moderate	Medium	Location ecologically desirable but not in local strategy Within area formally identified in local	\$	Low	0.39		
122	weathand and forest - Wet woodland	Woodland and forest - Wet woodland	Medium - Medium High - High	Moderate - Fairly Good Moderate - Fairly Good	0.666	High	Fairly Good	Medium	strategy Location ecologically desirable but not in local strategy	10	Medium	10.80		
134 135	Woodland and forest - Wet woodland	Woodland and forest - Wet woodland	High - High	Moderate - Fairly Good	175	High	Fairly Good	Medium	Within area formally identified in local strategy Within area formally identified in local	10	Medium	29.68		
136	Woodland and forest - Wet woodland	Woodand and forest - Wet woodland	High - High	Moderate - Fairly Good	0.44	High	Fairly Good	Medium	strategy Location ecologically desirable but not in local strategy	10	Medium	7.14		
127	Woodland and forest - Wet woodland	Woodland and forest - Wet woodland Woodland and forest - Wet woodland	High - High High - High	Moderate - Fairly Good	0.04	High High	Fairly Good	Medum	Location ecologically desirable but not in local strateev Location ecologically desirable but not	10	Medium	045		
139	Woodland and forest - Wet woodland	Woodland and forest - Wet woodland	High - High	Moderate - Fairly Good	0.376	High	Fairly Good	Medium	in local strateev Location ecologically desirable but not in local strateev	10	Medium	6.10		
14D 141	Woodland and forest - Wet woodland Woodland and forest - Wet woodland	Woodland and forest - Wet woodland Woodland and forest - Wet woodland	High - High High - High	Moderate - Fairly Good Moderate - Fairly Good	0.486	High	Fairly Good	Medum Medum	Within area tormany identified in local thrateev Area/compensation not in local	10	Medium Medium	824		
142	Woodland and forest - Wet woodland	Woodland and forest - Wet woodland	High - High	Moderate - Fairty Good	0.476	High	Fairly Good	Medium	Location ecologically desirable but not in local strategy Associations and in local	10	Medium	7.72		
143	Woodland and forest - Wet woodland Woodland and forest - Wet woodland	Woodland and forest - Wet woodland Woodland and forest - Wet woodland	High - High High - High	Poor - Moderate Moderate - Fairly Good	0.1	High High	Moderate Fairly Good	Medium	strateev/ no local strateev Location ecologically desirable but not in local strateev	10	Medium	0.97		
569	Woodland and forest - Lowland mixed deciduous woodland	Woodland and forest - Lowland mixed deciduous woodland	High - High	Moderate - Fairly Good	1.248	High	Fairly Good	Medium	Area/compensation not in local strategy/ no local strategy Area/compensation not in local	20	High	17.14		
153	Urban - Introduced shrub	Urban - Introduced shrub	Low-Low	Poor - Fairly Poor	1.1164	Low	Fairly Poor	Low	stateev/ no local stateev Area/compensation not in local strateev/ no local strateev	1	Low	331		
154	Woodland and forest - Lowland mixed deciduous woodland Grassland - Other neutral anasland	Woodland and forest - Lowland mixed deciduous woodland Grassland - Other neutral anasiland	High - High Medium - Medium	Moderate - Fairly Good Fairly Poor - Moderate	0.5704	High Medium	Fairly Good	Medium	Area/compensation not in local strategy/ no local strategy Area/compensation not in local	20	High	7.83		
156	Heathland and shrub - Bramble scrub	Heathland and shrub - Bramble scrub	Medum - Medum	Moderate - Fairly Good	0.5283	Medium	Fairly Good	LOW	Area/compensation not in local strateev/ no local strateev	1	LOW	\$25		
								-						
								-						
								-						
								-						
								_						

0-1 Sit	e Heage	Baseline															
60	detse / Show Main Mer	Columns Condense / Show Rows															
ſ		UK Habitata - existing habitata		Habitat	Habitat	Ecological	Strateek similicance	1	Ecological		Retection	category bi	odiversity va	ike		Com	nests
Baseline	Hedge	Hedgerowtype	length XM	Distinctiveness	Condition	Connectivity Ecological	Strategic significance	Suggested action to	Total hedgerow	Length	Length	Units	Units	Length	Units	Assessor comments	Reviewer comments
1	1	Hedge Ornamental Non Native	0.034	VLow	Moderate	Low	Within area formally identified in local strategy	Same distinctiveness band or better	0			0	0	0.034	0		
2	2	Hedge Ornamental Non Native	0.023	VLow	Moderate	Low	Within area formally identified in local strategy	Same distinctiveness band or better Same distinctiveness	0			0	0	0.023	0		
4	4	Native Hedgerow	0.057	Low	Moderate	LOW	Area/compensation not in local strategy/ no local strategy/ no	band or better Same distinctiveness band or better	0.228			0	0	0.025	0.228		
5	5	Native Hedgerow	0.107	Low	Moderate	Low	Area/compensation not in local strategy/ no local strategy	Same distinctiveness band or better	0.428			0	٥	0.107	0.428		
6	6	Native Hedgerow	0.063	Low	Poor	Low	Area/compensation not in local strategy/ no local strategy Area/compensation not in local strategy/ no	Same distinctiveness band or better Same distinctiveness	0.126		0.063	0	0.126	0	0		
7 8	7	Native Hedgerow	0.058	Low	Good	Low	Area/compensation not in local strategy/ no	hand or hatter Same distinctiveness	0.348		0.058	0	0.348	0	0.724		
9	9	Native Hedgerow - Associated with bank or ditch	0.235	Medium	Poor	Low	Area/compensation not in local strategy/ no local strategy	Like for like or better	0.94		0.235	0	0.94	0	0		
10	20	Native Hedgerow - Associated with bank or ditch	0.236	Medium	Poor	Low	Area/compensation not in local strategy/ no local strategy	Like for like or better	0.944			0	٥	0.235	0.944		
11	11	Native Hedgerow - Associated with bank or ditch	0.109	Medium	Poor	Low	Area/compensation not in local strategy/ no Area/compensation not in local strategy/ no	Like for like or better	0.436		0.109	0	0.436	0	0 128		
13	13	Native Hedgerow - Associated with bank or ditch	0.192	Medium	Poor	Low	local strategy Area/compensation not in local strategy/ no local strategy	Like for like or better	0.768	0.192		0.768	0	0	0		
14	ы	Native Hedgerow - Associated with bank or ditch	0.158	Medium	Poor	Low	Location ecologically desirable but not in local strategy	Like for like or better	0.6952		0.158	0	0.6952	0	0		
15	15	Native Hedgerow - Associated with bank or ditch	0.239	Medium	Poor	Low	Area/compensation not in local strategy/ no local strategy	Like for like or better	0.956		0.2	0	0.8	0.039	0.156		
16	12	Native Hedgerow - Associated with bank or ditch	0.099	Medium	Moderate	Low	Area/compensation not in local strategy/ no Area/compensation not in local strategy/ no	Like for like or better	0.792		0.099	0	0.792	0	0		
18	18	Native Hedgerow - Associated with bank or disch	0.13	Medium	Peor	Low	local strategy Area/compensation not in local strategy/ no	Like for like or better	0.52		0.09	0	0.36	0.04	0.16		
19	29	Native Hedgerow - Associated with bank or ditch	0.08	Medium	Peor	Low	Area/compensation not in local strategy/ no local strategy	Like for like or better	0.32		0.08	0	0.32	0	0		
20	20	Native Hedgerow - Associated with bank or ditch	0.127	Medium	Moderate	Low	Area/compensation not in local strategy/ no local strategy Area/compensation not in local strategy/ no	Like for like or better	1.016			0	0	0.127	1.016		
21	21	Native Hedgerow - Associated with bank or ditch	0.088	Medium	Good	Low	local strategy Area/compensation not in local strategy/ no	Like for like or better	1.056		0.068	0	1.056	0.137	0		
23	23	Native Hedgerow - Associated with bank or ditch	0.132	Medium	Moderate	Low	Area/compensation not in local strategy/ no local strategy	Like for like or better	1.056			0	0	0.132	1.056		
24	24	Native Hedgerow - Associated with bank or ditch	0.262	Medium	Moderate	Low	Area/compensation not in local strategy/ no local strategy	Like for like or better	2.096	0.121		0.968	٥	0.141	1.128		
25	25	Native Hedgerow - Associated with bank or ditch	0.084	Medium	Poor	Low	Aver/compensation not in local strategy/ no local strategy Area/compensation not in local strategy/ no	Like for like or better	0.236			0	0	0.084	0.336		
25	27	Native Hedgerow - Associated with bank or ditch	0.066	Medium	Moderate	Low	local stratery Area/compensation not in local strategy/ no	Like for like or better	0.528		0.066	0	0.528	0	0		
28	28	Native Hedgerow - Associated with bank or ditch	0.088	Medium	Moderate	Low	Area/compensation not in local strategy/ no local strategy	Like for like or better	0.704			0	0	0.088	0.704		
29	29	Native Hedgerow - Associated with bank or ditch	0.187	Medium	Good	Low	Area/compensation not in local strategy/ no local strategy	Like for like or better	2.244			0	٥	0.187	2.244		
20	30	Native Hedgerow - Associated with bank or ditch	0.084	Medium	Moderate	Low	Area/compensation not in local strategy/ no local strategy Area/compensation not in local strategy/ no	Like for like or better	0.672	-	0.054	0	0.672	0	0		
22	22	Native Hedgerow - Associated with bank or ditch	0.05	Medium	Poor	Low	local strategy Area/compensation not in local strategy/ no local strategy/ no	Like for like or better	0.2	0.05	avel	0.2	0	0	0		
22	22	Native Hedgerow - Associated with bank or ditch	0.141	Medium	Moderate	Low	Area/compensation not in local strategy/ no local strategy	Like for like or better	1.128	0.141		1.128	٥	0	0		
24	34	Native Hedgerow - Associated with bank or ditch	0.092	Medium	Poor	Low	Area/compensation not in local strategy/ no local strategy	Like for like or better	0.368	0.092		0.368	0	0	0		
ы Ж	25 26	Native Hedgerow - Associated with bank or ditch Native Hedgerow - Associated with head or Airch	0.019	Medium	Poor Poor	Low	www.nn area normally identified in local strategy Within area formally identified in local strategy	Like for like or better	0.022	0.019		0.0874	0	0	0		
27	22	Native Hedgerow - Associated with bank or ditch	0.006	Medium	Peor	Low	Ana/compensation not in local strategy/ no local strategy	Like for like or better	0.024	0.006		0.024	0	0	0		
28	28	Native Hedgerow - Associated with bank or ditch	0.115	Medium	Peor	Low	Area/compensation not in local strategy/ no local strategy	Like for like or better	0.46		0.115	0	0.46	0	0		
29	29	Native Hedgerow - Associated with bank or ditch	0.03	Medium	Moderate	Low	weaycompensation not in local strategy/ no local strategy Area/compensation not in local strategy/ ~~	Like for like or better	0.24		0.03	0	0.24	0	0		
41	40	Native Hedgerow - Associated with bank or ditch Native Hedgerow - Associated with bank or ditch	0.027	Medium	Moderate Moderate	Low	Area/compensation not in local strategy/ no	Like for like or better Like for like or better	0.256		0.027	0	0.216	0.032	0		
42	42	Native Hedgerow - Associated with bank or ditch	0.245	Medium	Poor	Low	Local strategy Location ecologically desirable but not in local strategy	Like for like or better	1.078		0.16	0	0.704	0.085	0.374		
43	43	Native Hedgerow - Associated with bank or ditch	0.13	Medium	Moderate	Low	Area/compensation not in local strategy/ no local strategy	Like for like or better	1.04	0.8		6.4	٥	Error in Areas	Error in Aceas		
44	44	Native Hedgerow - Associated with bank or ditch	0.096	Medium	Good	Low	Area/compensation not in local strategy/ no local strateev Area/compensation not in local strategy/ no	Like for like or better	1.152	0.096	0.304	1.152	0	0	0		
46	46	Native Hedgerow - Associated with bank or ditch	0.136	Medium	Poor	Low	Area/compensation not in local strategy/ no	Like for like or better	0.544	0.015	0.121	0.06	0.484	1.46-17	5.66-17		
47	47	Native Hedgerow - Associated with bank or ditch	0.127	Medium	Poor	Low	Location ecologically desirable but not in local strategy	Like for like or better	0.5588	0.08	0.047	0.352	0.2068	٥	2.86-17		
48	41	Native Hedgerow - Associated with bank or ditch	0.164	Medium	Moderate	Low	Location ecologically desirable but not in local strategy	Like for like or better	1.4422		0.164	0	1.4432	0	0		
49	49 50	Native Hedgerow - Associated with bank or ditch	0.207	Medium	Good	Low	local strategy Area/compensation not in local strategy/ no	Like for like or better	1.656		0.142	0	1.204	0.207	1.656		
51	51	Native Species Rich Hedgerow - Associated with bank or ditch	0.149	High	Moderate	Medium	Area/compensation not in local strategy/ no local strategy/	Like for like	1.9668		0.149	0	1.9668	0	0		
52	52	Native Species Rich Hedgerow - Associated with bank or ditch	0.089	High	Moderate	Medium	Area/compensation not in local strategy/ no local strategy	Like for like	1.1748		0.089	0	1.1748	0	0		
53	9 9	Native Species Rich Hedgerow - Associated with bank or dtch Native Species Rich Medeemar, Associated with bank or dtch	0.064	High	Poor	Medium	local strategy Area/compensation not in local strategy/ no	Like for like	0.4224		0.039	0	0	0.054	0.4224		
55	55	Native Species Rich Hedgerow - Associated with bank or dtch	0.121	High	Moderate	Medium	local strategy Area/compensation not in local strategy/ no local strategy	Like for like	1.5972		0.121	0	1.5972	0	0		
56	54	Native Species Rich Hedgerow - Associated with bank or dtch	0.135	High	Moderate	Medium	Area/compensation not in local strategy/ no local strategy	Like for like	1.782	0.092	0.09	0.4224	1.188	0.013	0.1716		
57	57	Native Species Rich Hedgerow - Associated with bank or ditch	0.168	High	Moderate	Medium	Area/compensation not in local strategy/ no local strategy Area/compensation not in local strategy/ no	Like for like	2.2176		0.064	0	1.1088	0.084	1.1088		
59	58	Native Species Icch Hedgerow - Associated with bank or ditch	0.102	High	Moderate	Medum	koral strategy Area/compensation not in local strategy/ no	Like for like	1.3464		0.08	0	1.584	0.057	0.2904		
60	60	Native Hedgerow - Associated with bank or ditch	0.05	Medium	Good	Low	Area/compensation not in local strategy/ no local strategy	Like for like or better	0.6			0	0	0.05	0.6		
61	61	Native Hedgerow with trees - Associated with bank or dtch	0.206	Medium	Moderate	Low	Location ecologically desirable but not in local strategy	Like for like or better	1.8128			0	٥	0.206	1.8128		
62	0	Native Hedgerow with trees - Associated with bank or ditch Nation Snaries Birk Medenma with trees - Associated with bank or ditch	0.188	Medium	Good	Low	Location ecologically desirable but not in local	Like for like or better	2.256	0.043	0.188	0 93654	2.256	0	0		
64	64	Native Hedgerow with trees - Associated with bank or dich	0.066	Medium	Good	Low	strategy Area/compensation not in local strategy/ no local strategy/ no	Like for like or better	0.792		0.066	0	0.792	0	0		
65	65	Native Hedgerow with trees - Associated with bank or ditch	0.119	Medium	Good	Low	Area/compensation not in local strategy/ no local strategy	Like for like or better	1.428		0.119	0	1.428	0	0		
66	66	Native Hedgerow with trees - Associated with bank or ditch	0.138	Medium	Good	Low	Area/compensation not in local strategy/ no local strategy Area/compensation not in local strategy/ no	Like for like or better	1.656		0.138	0	1.656	0	0		
67 68	6	Native Hedgerow with trees - Associated with bank or ditch	0.214	Medium	Good	Low	local strategy Area/compensation not in local strategy/ no	Like for like or better	2.568	0.214		2.568	0	0	0		
69	69	Native Hedgerow with trees - Associated with bank or ditch	0.203	Medium	Good	Low	Area/compensation not in local strategy/ no local strategy/	Like for like or better	2.436		0.15	0	1.8	0.053	0.636		
70	20	Native Hedgerow with trees - Associated with bank or dtch	0.092	Medium	Moderate	Low	Within area formally identified in local strategy	Like for like or better	0.2944	0.032		0.2944	٥	0	0		
71	71	Native Hedgerow with trees - Associated with bank or dich	0.079	Medium	Moderate	Low	Area/compensation not in local strategy/ no local strategy Area/compensation not in local strategy/ no	Like for like or better	0.632		0.079	0	0.632	0	0		
73	20	Native Species Rich Hedgerow - Associated with bank or disch	0.265	High	Good	Medium	local strategy Area/compensation not in local strategy/ no	Like for like	5.247	0.265		5.247	0	0	0		
74	24	Native Species Rich Hedgerow - Associated with bank or ditch	0.044	High	Good	Medium	Area/compensation not in local strategy/ no local strategy	Like for like	0.8712			0	0	0.044	0.8712		
75	75	Native Species Rich Hedgerow - Associated with bank or ditch	0.118	High	Good	Medium	Area/compensation not in local strategy/ no local strategy	Like for like	2.3364	0.118		2.3364	٥	0	0		
76	25	Native Species Rich Hedgerow - Associated with bank or dtch	0.204	High	Good	Medium	Area/compensation not in local strategy/ no Area/compensation not in local strategy/ no	Like for like	4.0392	0.204		4.0392	0	0	0		
78	28	Native Species Rich Hedgerow - Associated with bank or dich	0.09	High	Good	Medum	local strateev Area/compensation not in local strategy/ no local strategy	Like for like	1.782	0.09		1.782	0	0	0		
79	29	Native Species Rich Hedgerow - Associated with bank or ditch	0.139	High	Good	Medum	Area/compensation not in local strategy/ no local strategy	Like for like	2.7522	0.139		2.7522	٥	٥	0		
80	80	Native Species Rich Hedgerow - Associated with bank or ditch	0.156	High	Good	Medum	Area/compensation not in local strategy/ no local strateau Area/compensation not in local strateau/	Like for like	3.0658	0.156		3.0688	0	0	0		
#1 #2	81		0.031	High	Good	Medum	Area/compensation not in local strategy/ no	Like for like	0.6128	0.001		0.6138	0	0	0		
83	83	Native Species Rich Hedgerow - Associated with bank or disch	0.121	High	Good	Medium	Area/compensation not in local strategy/ no local strategy	Like for like	2.3958	0.121		2.3958	٥	0	0		
84	54	Native Species Rich Hedgerow - Associated with bank or dtch	0.17	High	Good	Medum	Area/compensation not in local strategy/ no local strategy	Like for like	2.366	0.17		2.366	٥	0	0		
85	85	native Species Rich Hedgerow - Associated with bank or ditch	0.127	High	Good	Medum	Area/compensation not in local strategy/ no	Like for like	2.5146	0.127		2.5146	0	0	0		
87	## \$2	Native Species Rich Hedgerow - Associated with bank or dich	0.038	High	Good	Medum	local strategy Area/compensation not in local strategy/ no	Like for like	0.7524			0	0	0.028	0.7524		
88	88	Native Species Rich Hedgerow - Associated with bank or ditch	0.057	High	Good	Medum	Area/compensation not in local strategy/ no local strategy	Like for like	1.1296			0	٥	0.057	1.1286		
89	89	Native Species Rich Hedgerow - Associated with bank or dich	0.043	High	Good	Medum	Area/compensation not in local strategy/ no local strategy Area/compensation not in local strategy	Like for like	0.8534	0.043		0.8514	0	0	0		
90	90 91	native Species Rich Hedgerow - Associated with bank or dich Native Species Rich Hedgerow - Associated with bank or disch	0.016	High	Good Goo4	Medum Medum	local strategy Area/compensation not in local strategy/ no	Like for like	0.9108	0.046		0.9308	0	0.019	0		
92	92	Native Species Rich Hedgerow - Associated with bank or disch	0.089	High	Good	Medum	Area/compensation not in local strategy/ no local strategy	Like for like	1.7622			0	0	0.089	1.7622		
93	93	Native Species Rich Hedgerow - Associated with bank or dtch	0.198	High	Good	Medum	Area/compensation not in local strategy/ no local strategy	Like for like	3.9204	0.298		3.9204	٥	0	0		
94	94	Native Species Rich Hedgerow with trees - Associated with bank or ditch	0.02	High	Good	Medium	local strategy Within any forgethic density	Like for like	0.296			0	0	0.02	0.296		
96	96	Native Species Rich Hedgerow with trees - Associated with bank or ditch	0.045	High	Good	Medum	Location ecologically desirable but not in local granew	Like for like	0.9800	0.045		0.9801	0	0	0		
97	97	Native Species Rich Hedgerow with trees - Associated with bank or ditch	0.384	High	Good	Medium	Within area formally identified in local strategy	Like for like	8.74368	0.284		8.74368	٥	0	0		
98	98	Native Species Rich Hedgerow with trees - Associated with bank or ditch	0.039	High	Good	Medum	Location ecologically desirable but not in local strategy Area/compensation not in local stratem/ m	Like for like	0.64942	0.039		0.54942	0	0	0		
99 100	99 100	Native Species Rich Hedgerow with trees - Associated with bank or ditch Native Species Rich Hedgerow with trees - Associated with bank ~ 49-h	0.256	High	Good	Medum Medum	Local strateev Location ecologically desirable but not in local	Like for like	0.60904	0.256	-	5.0688	0	0	0		
101	101	Native Species Rich Hedgerow with trees - Associated with bank or ditch	0.1	High	Good	Medum	strategy Area/compensation not in local strategy/ no local strategy	Like for like	1.98	0.1		1.98	٥	٥	0		
102	102	Native Species Rich Hedgerow with trees - Associated with bank or ditch	0.114	High	Good	Medium	Within area formally identified in local strategy	Like for like	2.59578	0.114	-	2.59578	٥	0	0		
103	103	Native Species Rich Hedgerow with trees - Associated with bank or ditch	0.042	High	Good	Medum	Avea/compensation not in local strategy/ no local strategy Area/compensation not in local strategy/ ~~	Like for like	0.8335		-	0	0	0.042	0.8316		
105	105	Native Species Rich Hedgerow with trees - Associated with bank or ditch	0.05	High	Good	Medum	local strateev Within area formally identified in local strateev	Like for like	4.07582	0.179		4.07583	0	0	0.99		
106	106	Native Species Rich Hedgerow with trees - Associated with bark or ditch	0.259	High	Good	Medium	Area/compensation not in local strategy/ no local strategy	Like for like	5.1282	0.259		5.1282	٥	0	0		
107	107	Native Species Rich Hedgerow with trees - Associated with bank or ditch	0.017	High	Good	Medum	Area/compensation not in local strategy/ no local strategy	Like for like	0.3366			0	٥	0.017	0.3366		
108	108	Native Species Rich Hedgerow with trees - Associated with bank or ditch	0.016	High	Good	Medum	Avea/compensation not in local strategy/ no local strategy Area/compensation not in local strategy/ no	Like for like	0.3168			0	0	0.016	0.3168		
110	110	Native Species Rich Hedgerow with trees - Associated with bark or ditch	0.028	High	Good	Medum	Area/compensation not in local strategy/ no krea/compensation not in local strategy/ no	Like for like	0.5544	0.026		0.5544	0	0	0		
111	111	Native Species Rich Hedgerow with trees - Associated with bank or ditch	0.026	High	Good	Medium	Area/compensation not in local strategy/ no local strategy	Like for like	0.5148	0.026		0.5148	٥	0	0		
112	112	Native Species Rich Hedgerow with trees - Associated with bank or ditch	0.022	High	Good	Medium	Area/compensation not in local strategy/ no local strategy	Like for like	0.4356			0	٥	0.022	0.4356		
113	113	Native Species Rich Hedgerow with trees - Associated with bank or ditch	0.05	High	Good	Medum	Within area formally identified in local strategy	Like for like	1.1365	0.05		1.1385	0	0	0		
115	114	menve synches with Hedgerow with these - Associated with bank or ditch Native Species Rich Hedgerow with these - Associated with bank or Alexis	0.129	High	Good	Medum	www.area tormany identified in local strategy Within area formally identified in local strategy	Like for like	9.88210	0.434		9.88219	0	0	0		
116	116	Native Species Rich Hedgerow with trees - Associated with bank or ditch	0.036	High	Good	Medum	Location ecologically desirable but not in local strategy	Like for like	0.78408	0.036		0.78428	0	0	0		
117	117	Native Species Rich Hedgerow with trees - Associated with bank or ditch	0.324	High	Good	Medium	Within area formally identified in local strategy	Like for like	7.27748	0.324		7.37748	٥	0	0		
118	118	Native Species Rich Hedgerow with trees - Associated with bank or ditch	0.058	High	Good	Medium	Location ecologically desirable but not in local strategy	Like for like	1.26324	0.058		1.26324	0	0	0		
119	119	wave species each reagence with trees - Associated with bank or ditch Native Species Rich Hederrow	0.045	High Medium	Good	Nedum	www.manea.normally.identified in local strategy Area/compensation not in local strategy/ no	Like for like	0.54	0.205		0.54	0	0	0		
121	121	Native Species Rich Hedgerow	0.091	Medium	Moderate	Low	Area/compensation not in local strategy/ no local strategy	Like for like or better	0.728	0.091		0.728	٥	0	٥		
122	122	Native Species Rich Hedgerow	0.038	Medium	Good	Low	Area/compensation not in local strategy/ no local strategy Location ecology also Assisted in the	Like for like or better	0.456	0.038		0.456	٥	0	0		
123	123	Native Species Rich Hedgerow - Associated with bank or ditch	0.112	High	Moderate	Medum	Location ecologically desirable but not in local	Like for like	1.62.04	0.000	0.112	0.7****	1.63624	0	0		
	- 44					essill	vivatees			0.001				Junit	- 10/01		

| 126 126

 | |

 | 1 |
 |
 |
 | | _
 | | | _
 | | _ | -
 | | |

---|---
--
--
---|--
--

--
---|---|--
--
---|---|--
---|---|--|--|--|
| 437 437

 | Native Species Rich Hedgerow - Associated with bank or disch | 0.172

 | High | Good
 | Medium
 | local strateev
Area/compensation not in local strategy/ no
 | Like for like | 3.4254
 | 0.14 | | 2.772
 | 0 | 0.033 | 0.6534
 | | |
| 129 170

 | Native Species Rich Hedeerow - Associated with bank or dech | 0.17

 | Hat | Moderata
 | Medare
 | local strateev
Area/compensation not in local strategy/ no
 | Like for the | 2.244
 | | 0.17 | 0
 | 2.264 | 0 |
 | | |
| 129 129

 | Native Species Rich Hedgerow - Associated with bank or disch | 0.164

 | High | Good
 | Medium
 | Location ecologically desirable but not in local
 | Like for like | 3.57192
 | 0.364 | | 3.57192
 | ٥ | 0 | 0
 | | |
| 130 130

 | Native Species Rich Hedgerow - Associated with bank or disch | 0.113

 | Hgt | Moderate
 | Medium
 | Area/compensation not in local strategy/ no
local strategy
 | Like for like | 1.4935
 | | 0.113 | 0
 | 1.4916 | 0 | 0
 | | |
| 131 131

 | Native Species Rich Hedgerow - Associated with bank or disch | 0.319

 | High | Moderate
 | Medium
 | Area/compensation not in local strategy/ no
local strategy
 | Like for like | 4.2108
 | | 0.289 | 0
 | 3.8148 | 0.03 | 0.396
 | | |
| 132 132

 | Native Species Rich Hedgerow - Associated with bank or disch | 0.162

 | High | Good
 | Medium
 | Location ecologically desirable but not in local
strategy
 | Like for like | 3.52836
 | 0.162 | | 3.52836
 | 0 | 0 | 0
 | | |
| 122 123

 | Native Species Rich Hedgerow - Associated with bank or disch | 0.163

 | High | Good
 | Medium
 | Area/compensation not in local strategy/ no
local strategy
Area/compensation not in local strategy/
 | Like for like | 3.2234
 | 0.13 | | 2.574
 | 0 | 0.033 | 0.6534
 | | |
| 136 136

 | -anne aperies examiningerow - Associated with bank or disch
Native Species Rich Hedgerow - Associated with have or disch | 0.1
0.2

 | Hat | Goo4
 | Medure
 | local strateev
Area/compensation not in local strategy/ no
 | Like for the | 2.96
 | | 0.1 | 2.94
 | 1.42 | 0 | 0
 | | |
| 136 136

 | Native Species Rich Hedgerow - Associated with bank or dich | 0.161

 | Hgt | Good
 | Medium
 | local strateev
Location ecologically desirable but not in local
 | Like for like | 3.50658
 | 0.361 | | 3.50658
 | 0 | 0 | 0
 | | |
| 137 137

 | Native Species Rich Hedgerow - Associated with bank or disch | 0.09

 | High | Good
 | Medium
 | strategy
Area/compensation not in local strategy/ no
local strategy
 | Like for like | 1.782
 | 0.077 | | 1.5246
 | ٥ | 0.013 | 0.2574
 | | |
| 138 138

 | Native Species Rich Hedgerow - Associated with bank or disch | 0.124

 | High | Good
 | Medium
 | Area/compensation not in local strategy/ no
local strategy
 | Like for like | 2.4552
 | 0.052 | | 1.0296
 | ٥ | 0.072 | 1.4256
 | | |
| 139 139

 | Native Species Rich Hedgerow - Associated with bank or ditch | 0.066

 | High | Moderate
 | Medium
 | Area/compensation not in local strategy/ no
local strategy
 | Like for like | 0.8712
 | | 0.066 | 0
 | 0.8712 | 0 | 0
 | | |
| 540 140

 | Native Species Rich Hedgerow - Associated with bank or disch | 0.028

 | High | Good
 | Medium
 | Avea/compensation not in local strategy/ no
local strategy
 | Like for like | 0.5544
 | 0.028 | | 0.5544
 | 0 | 0 | 0
 | | |
| 141 141

 | Native Species Rich Hedgerow - Associated with bank or disch | 0.178

 | High | Good
 | Medium
 | Area/compensation not in local strategy/ no
local strategy
Area/compensation not in local strategy/
 | Like for like | 3.5244
 | 0.178 | | 3.5244
 | 0 | 0 | 0
 | | |
| 142 142

 | native Species Rich Hedgerow - Associated with bank or disch | 0.121

 | High | Good
 | Medium
 | local strategy no
local strategy
Area/compensation not in local strategy/ no
 | Like for like | 2.3658
 | 0.121 | | 2.3958
 | 0 | 0 | 0
 | | |
| and 143

 | merve species rock Hedgerow - Associated with bank or disch
Native Species Rich Hedgerow - Associated | 0.048

 | High | Good
 | Medium
 | local strategy
Area/compensation not in local strategy/ no
 | Like for the | 4.602
 | 0.048 | | 4.60%
 | 0 | 0 | 0
 | | |
| 144
145 145

 | Native Species Rich Hedgerow - Associated with bank or disch | 0.207

 | High | Good
 | Medum
 | local strateev
Area/compensation not in local strategy/ no
 | Like for like | 4.0995
 | 0.237 | | 2.366
 | 0 | 0.037 | 0.7326
 | | |
| 146 146

 | Native Species Rich Hedgerow - Associated with bank or disch | 0.128

 | не | Good
 | Medium
 | local strateev
Area/compensation not in local strategy/ no
 | Like for like | 2.5344
 | 0.128 | | 2.5344
 | 0 | 0 | 0
 | | |
| 147 147

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.027

 | High | Good
 | Medium
 | ocal strategy
Within area formally identified in local strategy
 | Like for like | 0.65479
 | 0.627 | | 0.65479
 | ٥ | ٥ | 0
 | | |
| 148 148

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.095

 | High | Good
 | Medium
 | Within area formally identified in local strategy
 | Like for like | 2.15315
 | 0.095 | | 2.16315
 | ٥ | 0 | 0
 | | |
| 149 149

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.058

 | High | Good
 | Medium
 | Location ecologically desirable but not in local
strategy
 | Like for like | 1.25324
 | 0.058 | | 1.26324
 | ٥ | 0 | 0
 | | |
| 150 150

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.13

 | High | Moderate
 | Medium
 | wwa/compensation not in local strategy/ no
local strategy
 | Like for like | 1.716
 | | 0.1 | 0
 | 1.32 | 0.03 | 0.396
 | | |
| 151 151

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.128

 | High | Good
 | Medium
 | Area/compensation not in local strategy/ no
local strategy
Area/compensation not in local strategy/
 | Like for like | 2.5344
 | 0.128 | | 2.5344
 | 0 | 0 | 0
 | | |
| 152 152

 | manyer species kich Hedgerow with trees - Associated with bank or ditch
Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.049

 | High | Good
 | Medum
Martir
 | local strategy
Area/compensation not in local strategy/ no
 | Like for the | 0.9702
 | 0.001 | | 0
 | 0 | 0.049 | 0.9702
 | | |
| 154 154

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.113

 | High | Good
 | Medium
 | local strategy
Location ecologically desirable but not in local
 | Like for like | 2.46114
 | 0.113 | | 2.46114
 | 0 | 0 | 0
 | | |
| 155 155

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.121

 | High | Moderate
 | Medium
 | Location ecologically desirable but not in local
utrateev
 | Like for like | 1.75692
 | | 0.121 | 0
 | 1.75692 | 0 | 0
 | | |
| 156 156

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.132

 | High | Good
 | Medium
 | Location ecologically desirable but not in local
ubateev
 | Like for like | 2.87496
 | 0.132 | | 2.87496
 | ٥ | 0 | 0
 | | |
| 157 157

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.092

 | High | Good
 | Medium
 | Location ecologically desirable but not in local
strategy
 | Like for like | 2.00376
 | 0.092 | | 2.00376
 | ٥ | 0 | 0
 | | |
| 158 158

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.266

 | High | Good
 | Medium
 | Avea/compensation not in local strategy/ no
local strategy
 | Like for like | 5.2668
 | 0.366 | | 5.2668
 | 0 | 0 | 0
 | | |
| 159 159

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.067

 | High | Good
 | Medum
 | wew/compensation not in local strategy/ no
local strategy
 | Like for like | 1.3266
 | 0.067 | | 1.3266
 | 0 | 0 | 0
 | | |
| 160

 | menve species such reagenow with trees - Associated with bank or ditch | 0.087

 | High | Good
 | Medium
 | www.en.area.tormally.identified in local strategy
Area/compensation not in local strategy/ no
 | Like for like | 2.1070
 | 0.087 | | 2.98099
 | 0 | 0 | 0
 | | |
| 161
162 160

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.141

 | regri
High | Good
 | Medure
 | local strategy
Location ecologically desirable but not in local
 | Like for like | 3,09276
 | 0.10 | | 2.0978
 | 0 | 0 | 0
 | | |
| 163 163

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.117

 | High | Good
 | Medum
 | Unation ecologically desirable but not in local
 | Like for like | 2.54826
 | 0.117 | | 2.54826
 | 0 | 0 | 0
 | | |
| 164 164

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.262

 | High | Good
 | Medium
 | Location ecologically desirable but not in local
strategy
 | Like for like | 5.75636
 | 0.262 | | 5.70636
 | 0 | 0 | 0
 | | |
| 165 165

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.226

 | High | Good
 | Medium
 | Area/compensation not in local strategy/ no
local strategy
 | Like for like | 4.4741
 | 0.226 | - | 4.4748
 | ٥ | 0 | 0
 | | |
| 166 166

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.078

 | High | Good
 | Medium
 | Area/compensation not in local strategy/ no
local strategy
 | Like for like | 1.5444
 | 0.078 | | 1.5444
 | ٥ | 0 | 0
 | | |
| 167 167

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.085

 | High | Good
 | Medium
 | Avea/compensation not in local strategy/ no
local strategy
 | Like for like | 1.683
 | 0.085 | | 1.683
 | 0 | 0 | 0
 | | |
| 168 168

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.143

 | High | Good
 | Medium
 | Area/compensation not in local strategy/ no
local strategy
Area/compensation not in local strategy/ no
 | Like for like | 2.8354
 | 0.113 | | 2.2374
 | 0 | 0.03 | 0.594
 | | |
| 109 109

 | Native species Rich Hedgerow with trees - Associated with bank or ditch | 0.034

 | High | Good
 | Medium
 | Area/compensation not in local strategy/ no
 | Like for like | 22(2.0
 | 0.000 | | 0
 | 0 | 0.034 | 0.6732
 | | |
| 170

 | Native Species Rich Hedgerow with trees - Associated with bark or ditch | 0.106

 | High | Good
 | Medum
 | local strateev
Area/compensation not in local strategy/ no
 | Like for like | 2.1264
 | 0.026 | | 2.1284
 | 0 | 0 | 0
 | | |
| 172 172

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.064

 | High | Good
 | Medium
 | iocal strategy
Within area formally identified in local strategy
 | Like for like | 1.45728
 | 0.064 | | 1.45728
 | ٥ | 0 | 0
 | | |
| 173 173

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.079

 | High | Good
 | Medium
 | Area/compensation not in local strategy/ no
local strategy
 | Like for like | 1.5642
 | 0.06 | | 1.188
 | ٥ | 0.019 | 0.3762
 | | |
| 174 174

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.138

 | High | Good
 | Medium
 | Area/compensation not in local strategy/ no
local strategy
 | Like for like | 2.7224
 | 0.069 | | 1.3662
 | ٥ | 0.069 | 1.3662
 | | |
| 175 175

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.086

 | High | Good
 | Medium
 | Avea/compensation not in local strategy/ no
local strategy
Area/compensation not in local strategy
 | Like for like | 1.7028
 | 0.086 | | 1.7028
 | 0 | 0 | 0
 | | |
| 176 176

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.085

 | High | Good
 | Medium
 | Area/compensation not in local strategy no
 | Like for like | 1.683
 | 0.085 | | 1.683
 | 0 | 0 | 0
 | | |
| 178 178

 | narve systems such reagenow with trees - Associated with bank or ditch
Native Species Rich Hedgenow with trees - Associated with bank or dirch | 0.071

 | Hah | Good
 | Medum
Medum
 | local strateev
Area/compensation not in local strategy/ no
 | Like for like | 1.4058
 | 0.044 | | 1.4058
 | 0 | 0 | 0
 | | |
| 179 179

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.039

 | High | Good
 | Medium
 | local strateev
Area/compensation not in local strategy/ no
 | Like for like | 0.7722
 | 0.039 | | 0.7722
 | 0 | 0 | 0
 | | |
| 180 180

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.064

 | High | Good
 | Medium
 | Area/compensation not in local strategy/ no
Incal strategy
 | Like for like | 1.6632
 | 0.084 | | 1.6632
 | ٥ | ٥ | 0
 | | |
| 181 181

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.117

 | High | Good
 | Medium
 | Area/compensation not in local strategy/ no
local strategy
 | Like for like | 2.3166
 | 0.087 | | 1.7226
 | ٥ | 0.03 | 0.594
 | | |
| 192 192

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.129

 | High | Good
 | Medium
 | Area/compensation not in local strategy/ no
local strategy
 | Like for like | 2.5542
 | 0.129 | | 2.5542
 | ٥ | 0 | 0
 | | |
| 183 183

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.114

 | High | Good
 | Medium
 | Avea/compensation not in local strategy/ no
local strategy
 | Like for like | 2.2572
 | 0.114 | | 2.2572
 | 0 | 0 | 0
 | | |
| 184 184

 | Native Species Rich Hedgerow with trees - Associated with bank or ditch | 0.108

 | High | Good
 | Medium
 | wwaycompensation not in local strategy/ no
local strategy
 | Like for like | 2.1284
 | 0.228 | | 2.1384
 | 0 | 0 | 0
 | | |
| 185 185

 | Native Species Rich Hedgerow - Associated with bank or disch | 0.069

 | High | Good
 | Medium
 | Within area formally identified in local strategy
Location ecologically desirable but not in factor
 | Like for like | 1.57113
 | 0.069 | | 1.57113
 | 0 | 0 | 0
 | | |
| 186 186

 | native Species Rich Hedgerow - Associated with bank or dtch | 0.055

 | High | Good
 | Medium
 | strategy
 | Like for like | 1.1979
 | 0.055 | | 1.1979
 | ŏ | 0 | 0
 | | |
| 197 107

 | Native Sparles Birk Madesonar According to the book of | 0.117

 | | Medante
 | Martin
 | Analyzan bioreron in a number of a standard in a
 | Like for the | 1.4000
 | 0.443 | | 1 1000
 | | A |
 | | |
| 187 187
188 ····

 | Native Species Rich Hedgerow - Associated with bank or disch
Native Species Rich Hedgerow - Associated - the bank or disch | 0.113

 | | Moderate
Moderate
 | Medium
Martin
 | Area/compensation not in local strategy/ no
 | Like for like | 1.4935
 | 0.113 | | 1.4916
 | 0 | 0 | 0.9574
 | | |
| 187 187
188 188
189 189

 | Native Species Rich Hedgerow - Associated with bank or ditch
Native Species Rich Hedgerow - Associated with bank or ditch
Native Species Rich Hedgerow - Associated with bank or ditch | 0.113
0.103
0.045

 | Hgh | Moderate
Moderate
Poor
 | Medum
Medum
Medum
 | Ana/compensation not in local strategy no
local strategy
Ana/compensation not in local strategy/ no
local strategy
Ana/compensation not in local strategy/ no
local strategy
 | Like for like
Like for like
Like for like | 1.4935
1.3596
0.297
 | 0.113 | 0.045 | 1.4916
0.4092
0
 | 0 | 0.072 | 0.9504
 | | |
| 187 1.87 188 1.88 189 1.89 190 1.90

 | Native Species Rich-Hedgenser - Auscitated with bank or disch
Native Species Rich-Hedgenser - Auscitated with bank or disch
Native Species Rich-Hedgenser - Auscitated with bank or disch
Native Species Rich-Hedgenser - Auscitated with bank or disch | 0.113
0.103
0.045
0.116

 | High
High
High | Moderate
Moderate
Poor
Poor
 | Medum
Medum
Medum
Medum
 | Area/compensation not in local strategy no
local strategy
Area/compensation not in local strategy no
local strategy
Area/compensation not in local strategy no
local strategy
Area/compensation is local strategy no
local strategy
 | Like for like
Like for like
Like for like
Like for like | 1.4935
1.3596
0.297
0.7656
 | 0.113 | 0.045 | 1.4916
0.4092
0
 | 0 0.297 0.7656 | 0
0.072
0 | 0.9504
 | | |
| 187 187 188 188 189 189 190 190 191 191

 | Native Species Rich Hedgerson - Associated with bank or drith.
Native Species Rich Hedgerson - Associated with bank or drith.
Native Species Rich Hedgerson - Associated with bank or drith.
Native Species Rich Hedgerson - Associated with bank or drith.
Native Species Rich Hedgerson - Associated with bank or drith. | 0.112
0.103
0.045
0.116
0.117

 | High
High
High
High | Moderate
Noderate
Poor
Poor
Poor
 | Medum
Medum
Medum
Medum
 | Very (componention of a final manage in
Area/componention and in local strategy no
local strategy no
local strategy no
Area/componention and in local strategy no
local strategy
Area/componention and in local strategy no
local strategy
Area/componention and in local strategy no
local strategy
 | Like for like
Like for like
Like for like
Like for like
Like for like | 1.4955
1.3596
0.297
0.7656
0.7722
 | 0.112 | 0.045
0.115
0.117 | 1.4916
0.4092
0
0
 | 0 0.297 0.7656 0.7722 | 0.072 0 0 0 0 | 0.9504
 | | |
| 187 187 188 188 189 189 190 190 191 191 192 193

 | Notice Species RED Highers - Associated with tank or disth
factore Species RED Highers - Associated with tank or disth
Nation Species RED Highers - Associated with tank or disth
Nation Species RED Highers - Associated with tank or disth
Nation Species RED Highers - Associated with tank or disth
Nation Species RED Highers - Associated with tank or disth
Nation Species RED Highers - Associated with tank or disth | 0.113
0.103
0.045
0.116
0.117
0.152

 | High
High
High
High | Moderate
Noderate
Poor
Poor
Poor
-
 | Medum
Medum
Medum
Medum
Medum
 | Presponsibility of the responsibility of
New Jorgenson and Intransion. In stratings (no
Deal intransion
New Jorgenson and New Jorgenson
New | Like for like
Like for like
Like for like
Like for like
Like for like
Like for like
 | 1.4935
1.3596
0.227
0.7656
0.7722
1.0032
 | 0.031 | 0.045 | 1.4916
0.4092
0
0
0 | 0
0.297
0.7656
0.7722
1.0032
 | 0 0.072 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 | 0.9504 | | |
| 187 187
188 188
189 189
190 190
191 191
192 192
193 193

 | March agreem, Ruh Andgerur - Anaccia de Mar bar or dah.
March Spercen Mar Mangerur - Anaccia de Mar de dah.
March Spercen Mar Mangerur - Anaccia de Mar bar or dah.
March Spercen Mar Mangerur - Anaccia de val bar or dah.
March Spercen Mar Mangerur - Anaccia de Mar Mar dah.
March Spercen Mar Mangerur - Anaccia de Mar de dah.
March Spercen Mar Mangerur - Anaccia de val bar or dah.
March Spercen Mar Mangerur - Anaccia de val bar or dah.
March Spercen Mar Mangerur - Anaccia de val bar or dah. | 0.113
0.103
0.045
0.116
0.117
0.152
0.152
0.225
0.194

 | Нф
Нф
Нф
Нф
Нф
Нф | Moderate
Noderate
Poor
Poor
Poor
Poor
Poor
Gaori
 | Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
 | Areaccompanies and another and any of the
Area/companies on the local another
local another
local another
and another
Area/companies on the local another
Area/companies on the local another
Area/companies and in local another
Area/companies and in local another
Area/companies and in local another
Area/companies on the local another
Area/companies on the local another
local another
local another
Area/companies on the local another
Area/companies on the local another
local another
Area/companies on the local another
local another
Area/companies on the local another
Area/companie | Like for like
Like for like
 | 1.4915
1.3595
0.297
0.7656
0.7722
1.0022
1.485
3.662
 | 0.001 | 0.045
0.116
0.117
0.152
0.15 | 1.4916
0.4092
0
0
0
0
0
0
0
0 | 0
0.297
0.7656
0.7722
1.0032
0.99
0
 | 0.072 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 | 0.9504 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 | | |
| 187 187 188 188 189 189 190 160 191 192 192 193 194 194 195 165

 | Notice genesis, Nichologieur-Anscolated with basics and sha
Martine Generics, Nichologieur-Anscolated with basics and shi
Martine Generics, Nichologieur-Anscolated with basics and shi
Martine Generics, Nichologieur-Anscolated with basics and shi
Nationa Generics, Nichologieur-Anscolated with basics and shi
Nationa Generics, Nichologieur-Anscolated with basics and shi
Martine Generics, Nichologieur-Anscolated with basics and shi | 0.113
0.103
0.045
0.116
0.117
0.152
0.225
0.185
0.239

 | Нф
Нф
Нф
Нф
Нф
Нф
Нф | Moderate
Noderate
Poor
Poor
Poor
Poor
Poor
Good
Moderate
 | Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
 | Interpretentionen Bediationen
Bediationen
Bediationen in bediationen
Bediationen in bediationen
Bediationen
Analytemperation and in bediationen
Hensperentation mit in bediationen
hensperentationen in bediationen in bediationen
hensperentationen in bediationen in b | Like for like
Like for like
 | 1.4935
1.3596
0.237
0.7656
0.7722
1.0022
1.485
3.663
3.1548 | 0.113
 | 0.045
0.115
0.152
0.15
0.179 | 1.4916
0.4092
0
0
0
0
0
0
2.663
0 | 0
0.297
0.7656
0.7722
1.0032
0.99
0
2.3628
 | 0
0.072
0
0
0
0
0.075
0
0
0.05 | 0.9504
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | | |
| 187 187 188 188 189 189 190 100 191 101 192 102 193 103 194 194 195 105

 | Nach globen Richtigerer - Anzeiter der Bister er Ahn
Keiner Specie Richtigerer - Anzeiter der Bister er Ahn
Nacher Specie Richtigerer - Anzeiter der Bister er Ahn
Bister Specie Richtigerer - Anzeiter der Bister er Ahn | 0.113
0.103
0.015
0.116
0.117
0.152
0.152
0.185
0.229
0.089

 | нар
Нар
Нар
Нар
Нар
Нар
Нар
Нар | Moderate
Noderate
Poor
Poor
Poor
Poor
Poor
Good
Moderate
Good
 | Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
 | Independent Beid Interfere
Beid Interfere
Beid Interfere
Independent
Beid Interfere
Anskompendente Ander
Interfere
Anskompendente Ander
Beid Interfere
Anskompendente Ander
Beid Interfere
Anskompendente Ander
Beid Interfere
Anskompendente Ander
Beid Interfere
Beid Interfere
Anskompendente Ander
Beid Interfere
Beid I | Like for like
 | 1.4995
1.3596
0.297
0.7656
0.7722
1.0032
1.485
3.663
3.1548
1.93842 | 0.113 0.091 0.185 0.089
 | 0.045
0.115
0.152
0.15
0.17 | 1.4916
0.4092
0
0
0
0
0
0
0
2.662
0
1.92842 | 0
0.297
0.7656
0.7722
1.0032
0.99
0
2.3628
0
 | 0
0.072
0
0
0
0
0.075
0
0.075
0
0
0.06 | 0
0.9504
0
0
0
0.495
0
0.792
0 | | |
| 1107 100 100 100 100 101 102 103 104 105 105 106 107 108 109 100 101 102 103 104 105 106 106 107

 | Note desire Reisingers - Ancorate and have or also
Note legisles Reisingers - Ancorate and have or also | 0.113
0.103
0.045
0.116
0.117
0.152
0.152
0.255
0.239
0.239
0.239

 | нар
Нар
Нар
Нар
Нар
Нар
Нар
Нар
Нар | Moderate
Moderate
Poor
Poor
Poor
Poor
Good
Moderate
Good
 | Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
 | Noti construite to the distribution of the dis | Like for like
Like for like
 | 1.4995
0.297
0.7656
0.7722
1.0022
1.485
2.663
1.1588
1.159802
2.7852 | 0.113 0.091 0.185 0.185
 | 0.045
0.116
0.117
0.152
0.15
0.179
0.188 | 1.4916
0.4092
0
0
0
0
0
0
2.663
0
1.92842
0 | 0
0.297
0.7656
0.7722
1.0032
0.99
0
2.3638
0
2.3638
0
2.3638
 | 0
0.072
0
0
0
0
0.075
0
0
0.075
0
0
0
0
0
0
0 | 0
0.9504
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | | |
| 1127 1188 1189 1180 1180 1180 1181

 | Nach Gener Rechtlichunger- Anzeitarder Bister dem
Keiner Beschlicht Rechtigere- Anzeitarder Bister auf mit-
Nicht geschlicht Rechtlicher - Anzeitarder Bister er mit-
Bister Beschlicht Rechtlicher- Anzeitarder Bister er mit-
Bister Beschlicht Rechtlicherer- Anzeitarder Bister auf mit-
Bister Beschlicht Richtigerer- Anzeitarder Bister auf mit- | 0.113
0.103
0.045
0.116
0.117
0.152
0.225
0.185
0.239
0.089
0.211
0.115

 | нф
Нф
Нф
Нф
Нф
Нф
Нф | Moderate
Moderate
Poor
Poor
Poor
Good
Moderate
Poor
Poor
 | Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
 | More Constrained and a second se | Like for like
Like for like
 | 1.4995
1.3595
0.297
0.7525
1.0022
1.485
2.463
1.59842
2.7852
2.7852
0.7792 | 0.113 0.031 0.185 0.089 0.089
 | 0.045
0.115
0.152
0.15
0.15
0.179
0.181 | 1.4916
0.4092
0
0
0
0
0
2.663
0
1.92842
0
0 | 0
0.297
0.7656
0.7722
1.0032
0.99
0
2.3628
0
2.3628
0
2.3892
 | 0
0.072
0
0
0
0
0.075
0
0
0.075
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0.9504
0
0
0
0
0
0.495
0
0.792
0
0.295
0.295 | | |
| 1127 118 118 118 119 110

 | Next genesis Residences water and have a deal
March learners Residences and have a deal
Next learners Residences and have a deal
Next learners Residences and have a deal
Next learners Residences Annotated and have a deal
Reside learners Residences Annotated and have a deal
Residence Residences Annotated and have a deal
Residence Residences Annotated and have a deal
Residence Residence Residence Residence Residence Residence Residence
Residence Residences Annotated and have a deal
Residence Residence Residenc | 0.113
0.103
0.105
0.116
0.117
0.152
0.225
0.229
0.229
0.211
0.115
0.211

 | на
на
на
на
на
на
на
на
на
на
на
на | Moderate
Moderate
Poor
Poor
Poor
Poor
Good
Moderate
Poor
Noderate

 | Medum
Medum
Nedum
Nedum
Nedum
Nedum
Nedum
Medum
Medum
Nedum
 | Notice Constrained and a second secon | Like for like
Like for like
 | 1.4995
1.3595
0.297
0.3655
0.7722
1.6032
1.485
2.663
2.1548
1.92842
2.7852
0.759
0.8759 | 0.112 0.021 0.185 0.089
 | 0.045
0.116
0.117
0.152
0.15
0.15
0.179
0.181 | 1.4916
0.4092
0
0
0
0
2.663
0
1.92842
0
0
0
0
0 | 0
0.297
0.7656
0.7722
1.0032
0.99
0
2.3628
0
2.3628
0
2.3692
0
0
 | 0
0.072
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0.9504
0
0
0
0
0.495
0
0.792
0
0.792
0
0.2395
0.3759
0.3712 | | |
| 117 118 118 118 118 110 1111 1111 1111 1111

 | Note General Residence - Accessed and have or dis-
Note legench Residence - Accessed and have or dis-
Notes legench Residence - Maximi and have or dis-
Notes legench Residence - Accessed and have or dis-
Notes legench Residence - Maximi and have or dis-
Notes legench Residence - Maximi and have or dis-
Notes legench Residence - Maximi and ha | 0.113
0.045
0.103
0.104
0.107
0.152
0.152
0.152
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255

 | нар
Нар
Нар
Нар
Нар
Нар
Нар
Нар
Нар
Нар | Moderate
Moderate
Picor
Picor
Picor
Picor
Good
Moderate
Ficor
Picor
Moderate
Picor
Good
Moderate

 | Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
 | Kold Careford, K | Like for like
Like for like
 | 1.4995
1.3566
0.297
0.3665
0.7722
1.0032
1.485
1.6934
1.59342
2.7852
0.759
0.8759
0.8759
1.445
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1.59342
1. | 0.112 0.021 0.185 0.089 0.089
 | 0.045
0.115
0.117
0.152
0.15
0.179
0.181 | 1.4916
0.4092
0
0
0
0
0
1.663
0
1.52842
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0.297
0.7656
0.7722
1.0032
0.99
0
2.3628
0
2.3628
0
2.3628
0
0
2.3692
0
0
0
0
0
0 | 0
0.072
0
0
0
0.075
0
0.065
0.055
0.055
 | 0
0.9504
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | | |
| 137 148 149 149 149 140 140 141 142 143 143 144 145 145 145 145 145 145 145 145

 | Nucle devolution Relatingenza - Anciotate del Instituto d'An-
Marcine (secolo Relatingenza - Anciotate del Instituto d'An-
Norto Ingelo Relatingenza - Anciotate del Instituto d'An-
Marcine (secolo Relatingenza - Anciotate del Instituto d'An- | 0.113
0.045
0.103
0.116
0.117
0.152
0.225
0.152
0.225
0.225
0.225
0.225
0.229
0.089
0.211
0.115
0.259
0.066
0.259
0.056

 | 140
140
140
140
140
140
140
140
140
140 | Moderate
Moderate
Poor
Poor
Poor
Good
Moderate
Poor
Moderate
Poor
Moderate
Good
Moderate
Good

 | Nedum
Nedum
Medun
Medun
Nedun
Nedun
Nedun
Nedun
Nedun
Nedun
Nedun
Nedun
Nedun
Nedun
Nedun
Nedun
 | Marchardson Edit Carloss, and a second secon | Like for the
Like for the
 | 1.495 1.356 0.297 0.365 0.3722 1.0032 1.485 2.663 1.588 2.385 0.575 1.148 1.52842 2.385 2.385 2.354 | 0.113 0.031 0.089 0.089 0.089
 | 0.045
0.115
0.15
0.15
0.19
0.18 | 1.4916
0.4092
0
0
0
0
0
0
1.93842
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0.297
0.7656
0.7722
1.0032
0.99
0
2.3638
0
2.3638
0
2.3892
0
0
2.3892
0
0
2.3892
0
0
2.3892
0
0
0
2.3892
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0.072
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
 | 0
0.9504
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | | |
| asc 3.127 asc 3.138 asc 3.138 asc 3.120 sta 3.121

 | Note Seven Res Notes we Accessed with Note of Min.
Note Seven Res Notes Weight Network Accessed with Network of Min.
Notes Seven Res Notes Notes Network Network Network Network
Network Seven Res Network Network Network Network Network Network
Network Seven Res Network Network Network Network Network
Network Seven Res Network Network Network Network Network Network
Network Seven Res Network Netw | 0.112
0.103
0.045
0.116
0.117
0.152
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255

 | 140
140
140
140
140
140
140
140
140
140 | Moderste
Moderste
Poor
Poor
Poor
Poor
Poor
Good
Moderste
Good
Moderste
Good
Moderste

 | Nedum
Nedum
Medum
Medum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
 | Bell and any and any | Like to the
Like to the
 | 1.4995 1.3056 0.327 0.3654 0.3772 1.465 1.5032 1.465 1.53842 2.3553 0.379 0.8732 1.1484 2.3653 2.3534 0.538 | 0.113 0.031 0.089 0.089 0.128
 | 0.045
0.115
0.117
0.152
0.15
0.179
0.181 | 1.4916
0.4092
0
0
0
0
0
1.52342
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0.297
0.7656
0.7722
1.0032
0.99
0
2.3638
0
2.3892
0
0
2.3892
0
0
2.3892
0
0
2.3892
0
0
2.3892
0
0
0
2.34716
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
 | 0
0.072
0
0
0
0.075
0
0
0.056
0.056
0.056
0.056
0.056
0.056
0.056 | 0
0.9504
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | | |
| BB 2.127 IM 2.167

 | Nach Gener Rechtlicherer - Anzeitsterfer Hann eine Anster
Nach Bereich Rechtlicherer - Anzeitsterfer Hann eine An-
Bereit geschen Richtigerer - Anzeitsterfer Hann eines An-
Bereit geschen Richtigerer - Anzeitsterfer H | 0.113
0.045
0.116
0.117
0.152
0.225
0.229
0.089
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.211
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110
0.2110000000000

 | 140
140
140
140
140
140
140
140
140
140 | Moderste
Moderste
Poor
Poor
Poor
Poor
Poor
Good
Moderste
Good
Moderste
Good
Moderste
Good

 | Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
 | Marine Ma | Like or tee
Like to the
Like to the | 1.495
1.356
0.327
0.7522
1.0022
1.485
1.5382
2.3552
0.759
0.8712
1.148
1.53842
2.3553
0.759
0.8712
1.1484
2.5384
0.528
0.53856
 | 0.113 0.031 0.039 0.089 0.128 0.128
 | 0.045
0.115
0.152
0.152
0.153
0.159
0.181
0.181 | 1.4916
0.4092
0
0
0
0
0
1.92842
0
1.92842
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0.297
0.7656
0.7722
1.0002
0.99
0
2.3628
0
2.3692
0
0
2.3692
0
0
2.3692
0
0
3.4726
0
0
0
3.4726
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0.0772
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
 | 0.9504
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | | |
| BB 2.02 BB 2.02 BD 2.02

 | Note Specific Res Notes with a second | 0.113
0.065
0.116
0.117
0.122
0.225
0.185
0.239
0.239
0.231
0.239
0.231
0.239
0.231
0.239
0.231
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239
0.239

 | 140
140
140
140
140
140
140
140
140
140 | Moderate
Picor
Picor
Picor
Picor
Picor
Good
Moderate
Good
Moderate
Good
Moderate
Good
Moderate
Good

 | Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
 | | Like or the
Like to the
 | 1.4995 1.3564 1.3564 0.3565 0.3565 0.3752 1.6032 1.445 1.65342 0.3544 1.4534 1.4634 1.4634 1.4634 1.4634 1.53442 1.53442 1.4634 1.464 | 0.113 0.031 0.031 0.185 0.089 0.128 0.128 0.128
 | 0.045
0.115
0.117
0.152
0.153
0.159
0.181
0.181 | 1.4936
0.4002
0
0
0
0
0
1.50342
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0.397
0.7654
0.7722
0.99
0
2.3628
0
0
2.3628
0
0
0
2.3655
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
 | 0
0.0772
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0.0504
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | | |
| BI 2.127 BI 3.181 SI 3.161 BI 3.161 <td>Note devolved her highly any weak and and her for the
More beyon her highly any weak and any her highly any
Note highly her highly any weak any her highly
Note highly her highly any her highly
Note highly her highly any her highly
Note highly her highly any weak and had her highly
Note highly her highly any weak and had her highly
Note highly her highly any weak and her highly
Note highly her highly and her highly
Note highly her highly and her highly
Note highly her highly any highly and her hi</td> <td>0.113
0.065
0.116
0.117
0.152
0.255
0.255
0.255
0.218
0.218
0.211
0.211
0.211
0.251
0.251
0.251
0.251
0.251</td> <td>Hap
Hap
Hap
Hap
Hap
Hap
Hap
Hap
Hap
Hap</td> <td>Moderate
Picor
Picor
Picor
Picor
Picor
Good
Moderate
Good
Moderate
Good
Moderate
Good
Moderate
Good
Moderate
Good
Moderate</td> <td>Nedum
Medum
Medum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum</td> <td>Korr Carriers Korr Ca</td> <td>Laker in
Laker in
Laker</td> <td>1.4995
1.3566
0.3565
0.3565
0.3565
1.6022
1.485
2.663
1.485
2.7852
0.759
0.4772
1.464
1.4678
1.4678
0.5385
0.5385
1.4578
2.23465</td> <td>0.113
0.651
0.185
0.089
0.089
0.089
0.028
0.128</td> <td>0.045 0.115 0.15 0.15 0.18 0.18 0.263</td> <td>1.4936
0.4092
0
0
0
0
0
1.93462
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td> <td>0
0.2397
0.7656
0.7722
0.059
0
0.23695
0
0
2.3695
0
0
0
0
0
2.3695
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td> <td>0
0.072
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td> <td>0.9504
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td> <td></td> <td></td>
 | Note devolved her highly any weak and and her for the
More beyon her highly any weak and any her highly any
Note highly her highly any weak any her highly
Note highly her highly any her highly
Note highly her highly any her highly
Note highly her highly any weak and had her highly
Note highly her highly any weak and had her highly
Note highly her highly any weak and her highly
Note highly her highly and her highly
Note highly her highly and her highly
Note highly her highly any highly and her hi | 0.113
0.065
0.116
0.117
0.152
0.255
0.255
0.255
0.218
0.218
0.211
0.211
0.211
0.251
0.251
0.251
0.251
0.251

 | Hap
Hap
Hap
Hap
Hap
Hap
Hap
Hap
Hap
Hap | Moderate
Picor
Picor
Picor
Picor
Picor
Good
Moderate
Good
Moderate
Good
Moderate
Good
Moderate
Good
Moderate
Good
Moderate
 | Nedum
Medum
Medum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum
Nedum

 | Korr Carriers Korr Ca | Laker in
Laker | 1.4995
1.3566
0.3565
0.3565
0.3565
1.6022
1.485
2.663
1.485
2.7852
0.759
0.4772
1.464
1.4678
1.4678
0.5385
0.5385
1.4578
2.23465 | 0.113
0.651
0.185
0.089
0.089
0.089
0.028
0.128
 | 0.045 0.115 0.15 0.15 0.18 0.18 0.263 | 1.4936
0.4092
0
0
0
0
0
1.93462
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0.2397
0.7656
0.7722
0.059
0
0.23695
0
0
2.3695
0
0
0
0
0
2.3695
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
 | 0
0.072
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0.9504
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | | |
| BID Separation BID Separation Separation Separation Separation </td <td>Next generation in the second second</td> <td>0.113
0.065
0.116
0.117
0.152
0.225
0.225
0.229
0.089
0.211
0.115
0.239
0.089
0.211
0.115
0.239
0.089
0.211
0.115
0.239
0.051
0.116
0.231
0.117
0.115
0.225
0.225
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.255
0.235
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.</td> <td></td> <td>Moderste
Moderste
Peor
Peor
Peor
Peor
Peor
Peor
Peor
Peo</td> <td>Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum</td> <td>Marine Marine Santan
Marine Sa</td> <td>Like to fue
Like to fue</td> <td>1.4995
1.1556
0.297
0.3056
1.0022
1.4022
1.403
1.52842
2.3552
0.3925
1.1464
2.855
0.3926
0.528
1.4578
1.4578
1.25868
0.52868</td> <td>0.113
0.651
0.185
0.089
0.089
0.089
0.028</td> <td>0.045
0.135
0.137
0.132
0.15
0.15
0.15
0.15
0.15
0.15
0.15
0.15</td> <td>1.855
0.662
0
0
0
0
2.663
0
1.9282
0
0
0
1.9282
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td> <td>0
0.2397
0.7656
0.7722
1.0032
0.969
0
2.3628
0
0
2.3628
0
0
0
2.3628
0
0
0
2.3628
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td> <td>0
0.072
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td> <td>0
0.0504
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td> <td></td> <td></td>
 | Next generation in the second |
0.113
0.065
0.116
0.117
0.152
0.225
0.225
0.229
0.089
0.211
0.115
0.239
0.089
0.211
0.115
0.239
0.089
0.211
0.115
0.239
0.051
0.116
0.231
0.117
0.115
0.225
0.225
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.235
0.255
0.235
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.

 | | Moderste
Moderste
Peor
Peor
Peor
Peor
Peor
Peor
Peor
Peo

 | Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
 | Marine Marine Santan
Marine Sa | Like to fue
Like to fue | 1.4995
1.1556
0.297
0.3056
1.0022
1.4022
1.403
1.52842
2.3552
0.3925
1.1464
2.855
0.3926
0.528
1.4578
1.4578
1.25868
0.52868
 | 0.113
0.651
0.185
0.089
0.089
0.089
0.028 | 0.045
0.135
0.137
0.132
0.15
0.15
0.15
0.15
0.15
0.15
0.15
0.15 | 1.855
0.662
0
0
0
0
2.663
0
1.9282
0
0
0
1.9282
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
 | 0
0.2397
0.7656
0.7722
1.0032
0.969
0
2.3628
0
0
2.3628
0
0
0
2.3628
0
0
0
2.3628
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0.072
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0.0504
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 |
 | |
| BB 1.12 BB 1.28 BD 1.28 BD 1.20 BD

 | Note desire Residence Accordered Institute of the
Market desire Residence Accordered Institute of the
Note Residence Residence Accordered Institute of the
Note Residence Residence Accordered Institute of the
Note Residence Residence Accordered Instite of the
Note Residence Residence Accordered I | 0.113
0.103
0.065
0.115
0.127
0.152
0.185
0.229
0.089
0.239
0.066
0.051
0.064
0.051
0.064
0.217
0.064

 | | Moderste
Moderste
Pror
Pror
Pror
Pror
Pror
Good
Moderste
Good
Moderste
Good
Moderste
Good
Moderste
Good
Moderste

 | Nedun
Nedun
Nedun
Nedun
Nedun
Nedun
Nedun
Nedun
Nedun
Nedun
Nedun
Nedun
Nedun
Nedun
Nedun
Nedun
Nedun
Nedun
Nedun
 | | Lake for the
Cake for the
 | 1.495 1.156 2.027 0.366 0.772 1.002 1.402 2.463 1.404 1.428 2.385 0.79 0.873 1.164 1.428 2.534 0.53 0.53 0.53 0.536 1.4678 1.4678 1.2966 0.5628 1.2065 0.5628 1.206 | 0.113
0.091
0.185
0.089
0.185
0.089
0.128
0.089
 | 0.045
0.112
0.112
0.112
0.115
0.115
0.115
0.116
0.116 | 1.855
0.602
0
0
0
0
2.663
0
0
1.5382
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0.2377
0.7552
0.7572
0.0772
0.0772
0.0772
0.0772
0.0772
0.0772
0.0772
0.0772
0.0772
0.0
0.0
0.0
0.0
0.0
0.0
0.0
0.0
0.0
0.
 | 0
0.072
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0.0504
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | | |
| BI LI BI <

 | Note describe Residence - Accessed with some of the
More describe Residence - Accessed with some of the
South describe Residence - Accessed with some of the
South describe Residence - Accessed with some of the
Born describe Residence - Born | 0.113
0.103
0.045
0.116
0.117
0.152
0.225
0.255
0.269
0.211
0.066
0.291
0.053
0.064
0.053
0.064
0.053
0.064
0.053

 | | Moderste
Moderste
Peor
Peor
Peor
Peor
Peor
Geod
Moderste
Geod
Moderste
Geod
Moderste
Geod
Moderste
Geod
Moderste
Geod
Moderste

 | Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
Medum
 | Korrent and a second and a | Labor tao
Labor tao | 1.4956 1.2556 0.7772 0.787 0.7772 1.0027 1.0027 1.445 1.643 1.644 1.6340 0.778 0.8772 1.144 0.758 0.562 1.6572 0.562 0.562 1.6572 1.6545 0.564
0.564 0 | 0.113
0.021
0.021
0.185
0.089
0.089
0.089
0.089
0.020
0.020 | 0.045
0.15
0.12
0.15
0.17
0.18
0.18
0.262
 | 1.855
0.6002
0
0
0
0
0
0
1.9282
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0
0.237
0.7552
1.0032
0.0772
0.0772
0.0772
0.0772
0.0772
0.0772
0.0
0
0
0.2.3892
0
0
0
0.2.3892
0
0
0.0
0
0.0
0
0.0
0
0.0
0.0
0.0
0.0 | 0
0.0772
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
 | 0.5504
0.5504
0
0
0.4455
0
0.4455
0
0.4455
0
0.7529
0.3752
0.3752
0.3752
0.3752
0.3752
0.3752
0.3752
0.3554
0.3554
0.25284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.15284
0.152844
0.152844
0.152844
0.152844
0.152844
0.152844
0.152844
0.15 | | |
| BB 147 IM 149 IM 140

 | Nucle should be a set of the set | 0.113
0.103
0.045
0.116
0.117
0.152
0.225
0.239
0.239
0.211
0.115
0.066
0.230
0.128
0.058
0.231
0.058
0.231
0.054
0.231
0.054
0.231
0.054
0.231
0.054
0.231
0.054
0.231
0.054
0.235
0.054
0.235
0.054
0.235
0.055
0.235
0.055
0.235
0.055
0.235
0.055
0.235
0.055
0.235
0.055
0.235
0.055
0.235
0.055
0.235
0.055
0.235
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.

 | | Nadarasa
Madarasa
Paco
Paco
Paco
Paco
Actor
Actor
Madarasa
Madarasa
Madarasa
Madarasa
Madarasa
Madarasa
Madarasa
Madarasa
Madarasa
Madarasa
Madarasa
Madarasa
Madarasa
Madarasa

 | Nedum
 | Bell Barbary B | Lake for the
Lake for the
La | 1485
1326
1326
0377
0378
0377
1487
1487
1487
1487
1487
1487
1487
1487
1487
1487
1487
1487
1487
1487
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
1486
 | 0.113
0.021
0.021
0.185
0.089
0.089
0.089
0.089
0.089
0.028
0.028
0.028
0.027 | 0.045
0.136
0.137
0.132
0.15
0.179
0.185
0.253 | 1.4555
0.4602
0
0
0
0
0
1.52842
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
 | 0
0.237
0.7524
0.7722
0.969
0
2.3628
0
0
2.3628
0
0
0
2.3627
0
0
0
2.3626
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0.0772
0
0
0
0.0755
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 |
0.5504
0.5504
0
0
0.4455
0
0.4455
0
0.7520
0.7520
0.7520
0.7520
0.7520
0.7520
0.7520
0.7520
0.7528
0.7520
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7528
0.7558
0.7558
0.7558
0.7558
0.7558
0.7558
0.7558
0.7558
0.7558
0.7558
0.7558
0.7558
0.7558
0.7558
0.7558
0.7558
0.7558
0.7558
0.7558
0.7558
0 | | |
| H2 H2

 | Nucle science its hulling wave Account of the hard or dist
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account of the hard or dist.
Nucle science its hulling wave Account dash hard or dist.
Nucle sci | 0.113
0.103
0.045
0.116
0.152
0.255
0.229
0.231
0.155
0.239
0.231
0.155
0.239
0.231
0.056
0.231
0.056
0.056
0.051
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.

 | | Nadarsa
Madarsa
Para
Para
Para
Para
Para
Para
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
Nadarsa
 | Medium
Medium
Medium
Medium
Medium
Medium
Medium
Medium
Medium
Medium
Medium
Medium
Medium
Medium
Medium
Medium
Medium
Medium
Medium
Medium
Medium
 | Korrenze i Korren | Lake to the
Lake to the
Laket | 1485 1336 1337 234 234 234 234 24 24 24 24 24 24 24 24 24 24 24 24 24
 | 0.113
0.021
0.385
0.385
0.089
0.089
0.089
0.089
0.089
0.089
0.089
0.089
0.020
0.021
0.021
0.021
 | 0.045
0.134
0.137
0.132
0.15
0.179
0.179
0.179
0.179
0.179
0.179
0.179 | 1.8956
0.600
0
0
0
0
0
0
1.52364
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0.237
0.7524
0.7722
0.969
0
2.3628
0
0
2.3628
0
0
0
2.3628
0
0
0
0
2.3626
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0.0772
0
0
0
0
0
0.0755
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
 | 0
0.5504
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | | |
| H2 140 140 140 141 140

 | Nucle science Residence - Accessed with some of the
Mark science Residence - Accessed with some of the
Source Source Residence - Ac | 0.113
0.103
0.065
0.116
0.117
0.153
0.255
0.256
0.228
0.228
0.211
0.056
0.228
0.051
0.056
0.051
0.054
0.054
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.

 | |
bbbinsi
Marina
Par
Par
Par
Par
Par
Par
Par
Par
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Asses
Assess
Asses
Assess
Asses
Asses
Asses
Asses
Asses
Asses
Asses
Asses
Asses
Asses
Asses
Asses
Asses
Asses
Asses
Asses
Assess
Asses
Asses
Asses
Asses
Asses
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
Assess
 | Weden Medan
 | Korrent and a second and a | Labor State
Labor | 1485
1326
1327
1328
1329
1329
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1320
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
1300
 | | 0.045
0.117
0.117
0.113
0.117
0.117
0.117
0.117
0.117
0.117
0.117
0.117
0.117
 | 1.8956
0.6957
0
0
0
0
0
1.52364
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0
0.337
0.7664
0.7722
0.0002
0.000
2.3628
0
0
0
2.3628
0
0
0
2.3628
0
0
0
2.3628
0
0
0
0
2.3628
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0.0772
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0.5504
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
 | | |
| APP A APP

 | Next generation in the second | 0.113
0.103
0.065
0.116
0.123
0.225
0.126
0.226
0.228
0.238
0.211
0.058
0.058
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.055
0.054
0.054
0.054
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.

 | 140
140
140
140
140
140
140
140
140
140 |
Noterias
Noterias
Pero
Pero
Pero
Pero
Pero
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noterias
Noter
 | Weden Meden
Meden
Meden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Weden
Wed
 |
 | Lake to the
Lake | 1495
1336
1337
1336
1337
1372
1372
1382
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
1383
13
13
13
13
13
13
13
13
13
13
13
13
13
1 | 6.335
6.345
6.345
6.345
6.345
6.345
6.345
6.345
6.342
6.342
6.342
6.342
6.342
6.342
6.342
 | 0.045
0.134
0.137
0.152
0.155
0.181
0.263
0.263 | 1.8916
0.690
0
0
0
0
0
0
1.6936
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0
0.337
0.7664
0.7722
0.99
0
0
2.3828
0
0
2.3828
0
0
0
2.3828
0
0
0
2.3828
0
0
0
0
2.3828
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
 | 0
0.0572
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0.5504
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | | |
| BD LID

 | Nucle science Residence - Accessed and have of all
Marker sciences Residence - Accessed and have or all
Nucle sciences Res | 0.113
0.103
0.065
0.116
0.152
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.

 | |
Norman
Norman
Pror
Pror
Pror
Pror
Canto
Canto
Canto
Canto
Notarisse
Notarisse
Canto
Notarisse
Notarisse
Canto
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notarisse
Notari
 | Wedan Medan
Medan
Medan
Wedan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan
Medan Medan Meda
 | Kal Care and Car | Labor Table
Labor | 1495
1336
1336
1337
1337
1347
1447
1447
1447
1347
1347
 | 6.13
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03 | 0.045
0.136
0.137
0.132
0.15
0.15
0.15
0.15
0.15
0.15
0.15
0.15
 | 1.8916
0.692
0
0
0
0
0
0
0
1.8936
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0
0,337
0,772
0,772
0,98
0
0
2,3628
0
0
2,3628
0
0
0
0
0,24564
0
0
0,0
0
0,0
0
0,0
0,0
0,0
0,0
0,0
0, | 0
0.0572
0
0
0
0
0
0
0
0
0
0
0
0
0
 | 0
0.5504
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | | |
| 147 147 148 148 149 148 140 148 140 148 141 148 142 148 143 148 144 148 145 148 146 148 147 148 148 148 149

 | Nucle solver Exhibiting with a second | 0.113
0.103
0.055
0.115
0.152
0.225
0.152
0.235
0.235
0.235
0.231
0.066
0.231
0.056
0.231
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.057
0.056
0.057
0.056
0.057
0.056
0.057
0.056
0.057
0.056
0.057
0.056
0.057
0.056
0.057
0.056
0.057
0.056
0.057
0.056
0.057
0.056
0.057
0.056
0.057
0.056
0.057
0.056
0.057
0.056
0.057
0.056
0.057
0.056
0.057
0.056
0.057
0.056
0.056
0.057
0.056
0.056
0.057
0.056
0.056
0.056
0.057
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.

 | | bebener
Honorita
Per
Per
Per
Per
Per
Per
Per
Per

 | Weden Weden
Meden
Meden
Weden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden
Meden Meden Med
 | | Labor Star
Labor |
1495
1396
1397
1396
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1400
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490
1490 | 6 113
6 033
6 033
6 038
6 039
6 | 0.045
0.15
0.15
0.15
0.15
0.15
0.15
0.243
0.243
0.243
0.243
0.243
0.243
0.243
 | 1.8916
0.4600
0
0
0
0
1.5882
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0
0,337
0,752
0,752
0,96
0
2,3638
0
0
2,3638
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0.0272
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 |
0
0.5604
0
0
0
0.555
0
0.552
0
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.752
0.750 | | |
| BP BP BP

 | Nucleonic Residence Accordence in the set of all
Marke lackshift (Margare - Accordence in the set of all
Nucle lackshift (Margare - Accordence in the set of all
Nucle lackshift (Margare - Accordence in the set of all
Nucle lackshift (Margare - Accordence in the set of all
Nucle lackshift (Margare - Accordence in the set of all
Nucle lackshift (Margare - Accordence in the set of all
Nucle lackshift (Margare - Accordence in the set of all
Nucle lackshift (Margare - Accordence in the set of all
Nucle lackshift (Margare - Accordence in the set of all
Nucle lackshift (Margare - Accordence in the set of all
Nucle lackshift) (Margare - Accordence in the set of all
Nucle lackshift) (Margare - Accordence in the set of all
Nucle lackshift) (Margare - Accordence in the set of all
Nucle lackshift) (Margare - Accordence in the set of all
Nucle lackshift) (Margare - Accordence in the set of all
Nucle lackshift) (Margare - Accordence in the set of all
Nucle lackshift) (Margare - Accordence in the set of all
Nucle lackshift) (Margare - Accordence in the set of all
Nucle lackshift) (Margare - Accordence in the set of all
Nucle lackshift) (Margare - Accordence in the set of all
Nucle lackshift) (Margare - Accordence in the set of all
Nucle lackshift) (Margare - Accordence in the set of all
Nucle lackshift) (Margare - Accordence in the set of all
Nucle lackshift) (Margare - Accordence in the set of all
Nucle lackshift) (Margare - Accordence in the set of all
Nucle lackshift) (Margare - Accordence in the set of all
Nucle lackshift) (Margare - Accordence in the set of all
Nucle lackshift) (Margare - Accordence in the set of all
Nucle lackshift) (Margare - Accordence in the set of all
Nucle lackshift) (Margare - Accordence in the set of all
Nucle lackshift) (Margare - Accordence in the set of all
Nucle lackshift) (Margare - Accordence in the set of all
Nucle lackshift) (Margare - Accordence in the set of all
Nucle lackshift) (Margare - Accordence in the set of all
Nucle lackshift) (Margare - Accordence in the set of all
Nucle lacksh | 0.113
0.103
0.055
0.116
0.117
0.122
0.122
0.123
0.123
0.123
0.123
0.123
0.123
0.124
0.123
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.

 | | blobinson
Monitaria
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
Parce
P
 | Weden
 | Korr 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 | Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
Laboratoria
 | 1485
1356
1357
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
1457
14577
14577
14577
14577
14577
14577
14577
14577
14577
145
 | 6.132
6.81
6.85
6.99
6.99
6.99
6.99
6.99
6.99
6.99
6.92
6.92
6.92
6.92
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.93
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94
6.94 | 0.045
0.15
0.17
0.17
0.17
0.17
0.17
0.17
0.17
0.17 | 1.8516
3.4602
0
0
1.662
0
1.5364
0
1.5364
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0
0,227
0,7526
0,7722
0,029
0
0
2,2828
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
 | 0
0.0272
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0.5604
0
0
0
0.555
0
0.552
0
0.752
0.752
0.752
0.752
0.752
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755
0.755 | | |
| D2 AD2 B2 AD2

 | Nucleonic Robinger, Account of histor data
Nucle legisch Robinger, Account data Nucle data
Nucle legisch Robinger, A | 0.113
0.005
0.116
0.117
0.117
0.117
0.117
0.117
0.118
0.229
0.239
0.231
0.118
0.239
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.

 | | blobener
blobener
Per
Per
Per
Per
Per
Per
Per
Per
Per
P

 | Wedan Medan
 | Maria Carabana San Angelan San | Labor 1995
Charlor 1996
Charlor 1997
Charlor | 1495
1395
1395
1395
1395
1495
1495
1495
1495
1495
1495
1495
14
 | | 0.045
0.15
0.17
0.15
0.15
0.15
0.17
0.35
0.35
0.25
0.25
0.25
0.25
0.25
0.25
0.25
0.2 | 1.855
0.4652
0
0
1.5554
0
1.5554
0
0
0
0
0
0
0
0
0
0
0
0
0
 | 0
0
0.332
0.7554
0.752
0.099
0
2.3630
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0.0275
0
0
0.0275
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
 | | |
| 9 9 8 9 8 9 8 9 8 9 8 9 8 9 9 9

 | Nucl seine Exhigen A Anotaerah harar disa
Man dan bench Khingen A Anotaerah harar disa
Nucl seine Khingen A Anotaerah harar disa
Man dan bench Khingen A Anotaerah harar disa
Man dan khingen A Mantaerah harar disa
Man dan khingen A Mantaerah harar disa
Man dan khingen A Anotaerah harar disa
Man dan khingen Anotaerah harar dis | 0.113
0.005
0.116
0.117
0.117
0.117
0.117
0.117
0.118
0.228
0.229
0.218
0.118
0.118
0.118
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.058
0.

 | | biblioms Para

 | Wedne
Madar
Madar
Madar
Madar
Madar
Madar
Madar
Madar
Madar
Madar
Madar
Madar
Madar
Madar
Madar
Madar
Madar
Madar
Madar
Madar
Madar
Madar
 | Bel at servers and a server of a serv | Labor Steel
Labor Steel |
1495
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477
1477 |
6.137
6.83
6.435
6.435
6.435
6.437
6.437
6.437
6.437
6.437
6.437
6.437
6.437
6.437
6.437
6.437
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.438
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.448
6.4488
6.4488
6.4488
6.4488
6.4488
6.4488
6.4488
6.44888
6 | 0.045
0.15
0.15
0.15
0.17
0.18
0.28
0.28
0.28
0.28
0.28
0.28
0.28
0.2 | 1.855
4.4655
4.4555
4.455
4.455
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.4555
4.45556
4.45556
4.45556
4.45556
4.45556
4.45556
4.45566
4.45566
4.45566
4.45566
4.455666
4.4556666666666 | 0
0
0,327
0,7554
0,7722
0,099
0
0
1,2523
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
 | 0
0.0275
0
0.0275
0
0.0275
0
0.0275
0
0.0275
0
0.0275
0
0.026
0
0.0210
0
0.0210
0
0.0210
0
0.0210
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0.5504
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | | |
| 20 3

 | Nucleonic Robinger Account of histor data
Maria lackshiftinger Account of histor data
Nucle lackshiftinger Acc | 0.113
0.005
0.116
0.115
0.152
0.225
0.152
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.

 | |
biolesses
biolesses
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Para
Pa
 | Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Walnes
Wa
 | Korrent and a second a second and a second a | Labor 199 |
1408
1506
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507
1507 |
6137
6838
6.338
6.338
6.338
6.338
6.338
6.339
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.489
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.499
6.49 | 0.065
0.15
0.15
0.15
0.15
0.15
0.262
0.062
0.062 | 1.855
0.4652
0.
0.
0.
0.
0.
0.
0.
0.
0.
0.
0.
0.
0. | 0
0
0
3397
0.7656
0.7202
0.898
0
0
2.3638
0
0
2.3638
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
 | 0
0.0272
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 5
5.5564
6.5564
0
0
0
0
0
0
0
0
0
0
0
0
0 | | |
| 20 4 20 4 20 <td>Nucleonic Richardson Annual An</td> <td>0.113
0.005
0.116
0.115
0.152
0.225
0.152
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.</td> <td></td> <td>blobersen
Blobersen
Person
Person
Person
Person
Person
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
B</td>
<td>Weinn
Weinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinnn
Meinn
Meinnn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Mein</td> <td>Bell Barbary Bell Barbary B</td> <td>Labor Star
Labor Star</td> <td>1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405</td> <td>6.137
6.873
6.875
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.</td> <td>0.045
0.134
0.172
0.152
0.152
0.182
0.182
0.182
0.042
0.042</td> <td>1.855
0.4692
0
0
0
1.5882
0
1.5882
0
1.5884
0
0
0
0
0
0
0
0
0
0
0
0
0</td> <td>0
0
0
337
0.756
0.752
0.752
0
0
2.328
0
0
2.328
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td>
<td>0
0.0272
0
0
0.0275
0
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.0566
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0</td> <td>5
5.5504
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td> <td></td> <td></td> | Nucleonic Richardson Annual An | 0.113
0.005
0.116
0.115
0.152
0.225
0.152
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.225
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.255
0.

 | |
blobersen
Blobersen
Person
Person
Person
Person
Person
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
Blobersen
B
 | Weinn
Weinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinnn
Meinn
Meinnn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Meinn
Mein
 | Bell Barbary B | Labor Star
Labor Star
 | 1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405
1405 |
6.137
6.873
6.875
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.895
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6.995
6. | 0.045
0.134
0.172
0.152
0.152
0.182
0.182
0.182
0.042
0.042 | 1.855
0.4692
0
0
0
1.5882
0
1.5882
0
1.5884
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0
0
337
0.756
0.752
0.752
0
0
2.328
0
0
2.328
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
 | 0
0.0272
0
0
0.0275
0
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0.0566
0.056
0.056
0.056
0.056
0.056
0.056
0.056
0 | 5
5.5504
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | | |
| 20 30 30 <

 | Next selection between the selection of | 0.113
0.103
0.045
0.112
0.152
0.152
0.152
0.152
0.153
0.054
0.054
0.054
0.054
0.054
0.054
0.054
0.055
0.051
0.055
0.051
0.051
0.055
0.051
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.055
0.

 | | blobener
blobener
Per
Per
Per
Per
Per
Per
Per
P

 | Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
Weine
 | Bell attack Bell atta | Labor See
Labor |
1498
1498
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497
1497 |
6113
6033
6048
0.048
0.048
0.048
0.048
0.042
0.042
0.042
0.042
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044
0.044 | 0.065
0.13
0.17
0.17
0.17
0.17
0.18
0.28
0.28
0.28
0.28
0.28
0.28
0.28
0.2 | 1.855
0.4692
0
0
0
0
1.5850
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 |
0
0
0
3.757
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.7552
0.75 | 0
0.0272
0
0
0.0275
0
0
0.025
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | | |
| 30 30 30 <

 | Nucleonic Robinger Account of histor data
Nucle specific Robinger Account of | 0.113
0.103
0.005
0.112
0.125
0.125
0.126
0.128
0.128
0.128
0.128
0.128
0.128
0.059
0.050
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.051
0.

 | | blobener
helener
Per
Per
Per
Per
Per
Per
Per
Per
Per
P

 | Walne Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
Warne
War
 | Korrent and a second a second and a second a sec | Labor State Labor State Labor State | Lange Lange </td <td>6 113
6 013
6 033
0 035
0 039
0 000
0 000
0 000
0 0000
0 0000
0 0000
0 0000
0 0000
0 00</td> <td>0.066
0.136
0.17
0.12
0.12
0.12
0.12
0.12
0.26
0.26
0.26
0.26
0.26
0.26
0.26
0.14</td> <td>1.8854
0.4652
0
0
0
0
1.8840
0
1.8840
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td>
<td>0
0
0
3.757
0.7554
0.7554
0
0
0
1.0022
0.898
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td> <td>0
0.0272
0
0.0275
0
0
0.0275
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td> <td>5
5.5564
5.556
6.557
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752</td> <td></td> <td></td> | 6 113
6 013
6 033
0 035
0 039
0 000
0 000
0 000
0 0000
0 0000
0 0000
0 0000
0 0000
0 00 | 0.066
0.136
0.17
0.12
0.12
0.12
0.12
0.12
0.26
0.26
0.26
0.26
0.26
0.26
0.26
0.14 | 1.8854
0.4652
0
0
0
0
1.8840
0
1.8840
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 |
0
0
0
3.757
0.7554
0.7554
0
0
0
1.0022
0.898
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0.0272
0
0.0275
0
0
0.0275
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 5
5.5564
5.556
6.557
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752
6.752 | | |
| 20 4 20 4 20 <td>Nucl societ. Shallyon with a second societ of all societ and and societ. Shallyon with a second societ of all societ. Shally and societ. Shally an</td> <td>0.113
0.005
0.005
0.112
0.123
0.226
0.228
0.228
0.228
0.230
0.212
0.051
0.051
0.051
0.051
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.</td> <td></td> <td>biblioma biblioma Biblioma Para Pa</td> <td>Walan Walan Walan<!--</td--><td>Bel a target a second a s</td><td>Labor Steel
Labor Steel</td><td>Lange Lange Lange<!--</td--><td>6.137
6.137
6.437
6.437
6.437
6.437
6.437
6.437
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.4577 6.4577
6.</td><td>0.066
0.134
0.17
0.12
0.13
0.15
0.242
0.242
0.052
0.052
0.052</td><td>1.8854
0.6692
0
0
0
0
0
0
1.5884
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td><td>0
0
0
0.337
0.7552
1.0332
0.9772
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td><td>0
0.0272
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td><td>5
5.5564
5.556
5.556
5.557
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575</td><td></td><td></td></td></td>
 | Nucl societ. Shallyon with a second societ of all societ and and societ. Shallyon with a second societ of all societ. Shally and societ. Shally an | 0.113
0.005
0.005
0.112
0.123
0.226
0.228
0.228
0.228
0.230
0.212
0.051
0.051
0.051
0.051
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.151
0.

 | | biblioma biblioma Biblioma Para Pa

 | Walan Walan </td <td>Bel a target a second a s</td> <td>Labor Steel
Labor Steel</td> <td>Lange Lange Lange<!--</td--><td>6.137
6.137
6.437
6.437
6.437
6.437
6.437
6.437
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.4577 6.</td><td>0.066
0.134
0.17
0.12
0.13
0.15
0.242
0.242
0.052
0.052
0.052</td><td>1.8854
0.6692
0
0
0
0
0
0
1.5884
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td><td>0
0
0
0.337
0.7552
1.0332
0.9772
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td><td>0
0.0272
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td><td>5
5.5564
5.556
5.556
5.557
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575</td><td></td><td></td></td> | Bel a target a second a s | Labor Steel
Labor Steel
 | Lange Lange </td <td>6.137
6.137
6.437
6.437
6.437
6.437
6.437
6.437
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.4577 6.</td> <td>0.066
0.134
0.17
0.12
0.13
0.15
0.242
0.242
0.052
0.052
0.052</td> <td>1.8854
0.6692
0
0
0
0
0
0
1.5884
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td> <td>0
0
0
0.337
0.7552
1.0332
0.9772
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td> <td>0
0.0272
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td> <td>5
5.5564
5.556
5.556
5.557
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575</td> <td></td> <td></td> |
6.137
6.137
6.437
6.437
6.437
6.437
6.437
6.437
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.447
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.457
6.4577 6. | 0.066
0.134
0.17
0.12
0.13
0.15
0.242
0.242
0.052
0.052
0.052 | 1.8854
0.6692
0
0
0
0
0
0
1.5884
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0
0
0.337
0.7552
1.0332
0.9772
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
 | 0
0.0272
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 5
5.5564
5.556
5.556
5.557
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575
5.575 | | |
| 20 3

 | Nucleonic Robinger Account of histor data
Nucleonic Robinger Account of | 0.113
0.066
0.116
0.117
0.122
0.223
0.223
0.228
0.228
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.231
0.

 | |
biolemen
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
biolement
bioleme
 | Wadan
 | Action | Labor See 2014
Cabor See 2014 | 1408
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
 | 613
633
644
644
645
645
645
645
645
645 | 0.045
0.115
0.117
0.12
0.12
0.15
0.25
0.25
0.25
0.25
0.25
0.25
0.25
0.2
 | 1.8854
0.6692
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0
0
2327
3.564
0
0
23628
0
0
23628
0
0
24526
0
24526
0
24526
0
0
24526
0
0
24526
0
0
0
24526
0
0
0
24526
0
0
0
0
24526
0
0
0
0
0
24526
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0.0272
0
0
0
0
0
0.055
0
0.054
0
0.054
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 5
5.500
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
 | | |
| 20 30 30 <

 | Barta akan Kanagawa Managawa M | 0.1131 0.0061 0.1142 0.1152 0.1127 0.1127 0.1228 0.238 0.238 0.238 0.238 0.239 0.239 0.239 0.329 0.329

 | | Noticipant Part Part
<td>budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan</td> <td>Bel Barbary Bel Barba</td> <td>Labor Steel
Labor Steel</td> <td>Less
Same
Same
Same
Same
Same
Same
Same
Same</td> <td>6 133
6 035
6 345
6 345
6 345
6 462
6 462
6</td> <td>0065
0135
0137
0137
0137
0138
0138
0138
0138
0138
0138
0138
0138</td> <td>1.6852
G. 6652
G. 6652
G. 6655
G. 6755
G. 7555
G. 7</td> <td>0
0
0
2.327
1.0562
0.0772
2.0002
0
0
2.3628
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td> <td>0
0.0272
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td> <td>5
5.500
5.500
6.500
6.500
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
7.750
6.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.75</td> <td></td> <td></td> |
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
budan
 | Bel Barbary Bel Barba | Labor Steel
Labor Steel | Less
Same
Same
Same
Same
Same
Same
Same
Same
 | 6 133
6 035
6 345
6 345
6 345
6 462
6 | 0065
0135
0137
0137
0137
0138
0138
0138
0138
0138
0138
0138
0138 | 1.6852
G. 6652
G. 6652
G. 6655
G. 6755
G. 7555
G. 7 | 0
0
0
2.327
1.0562
0.0772
2.0002
0
0
2.3628
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0.0272
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
 | 5
5.500
5.500
6.500
6.500
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
6.750
7.750
6.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.750
7.75 | | |
| 20 3

 | Naci saka ka halpan ya waka ana ka ka ana ya waka ana ya | 0.1131 0.1036 0.1147 0.1127 0.1126 0.1127 0.1126 0.1228 0.1229 0.1241 0.1241 0.1241 0.1241 0.1241 0.1241 0.1241

 | | babana
babana
Para
Para
Para
Para
Para
Para
Para
P

 | Madan
Waran
Waran
Waran
Waran
Waran
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
Madan
 | Bell Barbardson and a second a second and a second and a second and a second and a second a | Labor See 2014
Labor See 2014 | 1488 1232 12332 12332
 | 6 333
6 387
6 387
6 387
6 399
6 | 0.065
0.114
0.127
0.151
0.151
0.151
0.052
0.052
0.052
0.052
0.052
0.052 | 1.6002
0.6002
0
0
0
1.6300
0
1.6300
0
1.6300
0
0
0
0
0
0
0
0
0
0
0
0
 | 0
0
0.227
0.762
0.772
0.092
0.2328
0
0
2.3428
0
0
0
0.23428
0
0
0.24726
0
0
0.24726
0
0
0.24726
0
0
0.24726
0
0
0
0.24726
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0.772
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 5
5.500
5.500
0
0
0
0
0
0
0
0
0
0
0
0
 | | |
|

 | Naci saka kata kata ya kata y | 0.113
0.066
0.061
0.112
0.125
0.126
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.226
0.227
0.226
0.227
0.226
0.227
0.226
0.226
0.227
0.226
0.227
0.227
0.226
0.227
0.226
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.227
0.

 | |

 | Walan Waran
 | Martingen and sense and | Lake raise
Lake r | Lans
Lans
Lans
Lans
Lans
Lans
Lans
Lans
 | 6 133
6 393
6 395
6 395
6 499
6 | 0.065
0.135
0.137
0.137
0.137
0.252
0.252
0.252
0.252
0.252
0.252 | 1.6854
0.6652
0
0
0
0
0
1.5860
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 |
0
0
0
0.237
0.752
0.752
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772
0.772 | 0
0.0772
0
0
0
0
0
0.075
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 5
5.500
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | | |
| a a

 | Nucleonic Robinger, Aussister all heart all sear all sear | 0.1131 0.0461 0.1140 0.1141 0.1121 0.122 0.123 0.1241 0.125 0.126 0.127 0.128 0.128 0.128 0.128 0.128 0.128 0.128 0.128 0.128 0.128 0.128 0.128 0.128 0.128 0.129 0.120 0.121 0.126 0.127 0.128 0.129 0.121 0.124 0.134 0.134 0.134 0.134 0.134 0.134 0.134 0.134 0.134 0.134 0.134 0.134 0.134 0.134 0

 | | babana
babana
Para
Para
Para
Para
Para
Para
Para
P

 | budin
Warn
Warn
Warn
Warn
Warn
Warn
Warn
War
 | Bell Barbares and a second secon | Labor See See See See See See See See See Se
 | |
6.13
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03
6.03 | 0.046
0.137
0.137
0.13
0.13
0.242
0.062
0.062
0.062
0.062
0.062
0.062
0.062
0.062
0.062
0.062 | 1.8854
0.6892
0.6992
0.0
0.0
0.0
0.0
0.0
0.0
0.0
0.0
0.0
0. | 0
0
0
2327
0 3.864
0
2.772
2 3.892
0
2 3.892
0
0
2 3.892
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
 | 0
0,272
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 5
5
5
5
5
5
5
5
5
5
5
5
5
5 | | |
| of of a a

 | Nucleonic Robinger Account of histor data
Nucleonic Robinger of histor, Account of histor data
Nucleonic Robingero | 0.1131 0.066 0.116 0.116 0.117 0.128 0.228 0.228 0.316 0.317 0.116 0.117 0.128 0.318 0.311 0.116 0.311 0.312 0.312 0.312 0.312 0.324 0.312 0.312 0.324 0.312 0.325 0.326 0.327 0.328 0.329 0.329 0.329 0.329 0.329 0.329 0.329 0.329 0.329 0.329 0.329 0.329 0.329 0.329 0.329 0.329 0.329 0.329 0.329<

 | | Notesta Res

 | badan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
 | And a programme a | Labor See See See See See See See See See Se | Lange Lange </td
<td>6.13
0.93
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94</td> <td>0.066
0.134
0.137
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318</td> <td>1.4814
0.4692
0
0
0
1.4814
0
1.4814
0
0
0
1.4814
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td> <td>0
0
0,237
0,752
1,032
0,99
0
0
2,382
0
0
2,382
0
0
2,382
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td> <td>0
0,272
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td> <td></td> <td></td> <td></td> | 6.13
0.93
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94
0.94 |
0.066
0.134
0.137
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318
0.318 | 1.4814
0.4692
0
0
0
1.4814
0
1.4814
0
0
0
1.4814
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0
0,237
0,752
1,032
0,99
0
0
2,382
0
0
2,382
0
0
2,382
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0,272
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
 | | | |
| a a

 | Nari quos habitagen Annotagen bastar als a | 0.113 0.041 0.116 0.116 0.116 0.116 0.116 0.116 0.122 0.152 0.123 0.152 0.152 0.152 0.152 0.060 0.061 0.062 0.161 0.162 0.163 0.164 0.170 0.161 0.171 </td <td></td> <td>babana
babana
Para
Para
Para
Para
Para
Para
Para
P</td> <td>budin
Warn
Warn
Warn
Warn
Warn
Warn
Warn
War</td> <td>Bell Barbary Bell Barbary B</td> <td>Lake trais
Lake trais</td> <td>Lens
Autor
Lans
Lans
Lans
Lans
Lans
Lans
Lans
Lans</td> <td></td> <td>0.066
0.130
0.137
0.137
0.15
0.15
0.242
0.252
0.252
0.252
0.252
0.252
0.252</td> <td>1.8030
0.6092
0.000
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0</td> <td>0
0
0
0.9772
1.0032
0.7526
0
0
1.3430
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td>
<td>0
0,272
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td> <td>5
5
5
5
5
5
5
5
5
5
5
5
5
5</td> <td></td> <td></td>
 | | babana
babana
Para
Para
Para
Para
Para
Para
Para
P

 | budin
Warn
Warn
Warn
Warn
Warn
Warn
Warn
War
 | Bell Barbary B | Lake trais
Lake trais | Lens
Autor
Lans
Lans
Lans
Lans
Lans
Lans
Lans
Lans
 | | 0.066
0.130
0.137
0.137
0.15
0.15
0.242
0.252
0.252
0.252
0.252
0.252
0.252 |
1.8030
0.6092
0.000
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0.00
0 | 0
0
0
0.9772
1.0032
0.7526
0
0
1.3430
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0,272
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 5
5
5
5
5
5
5
5
5
5
5
5
5
5
 | | |
|

 | Naci socio Absilgen Absilgen Absolute State S | 0.111 0.000 0.112 0.112 0.112 0.122 0.122 0.122 0.122 0.122 0.122 0.122 0.122 0.122 0.122 0.123 0.131 0.124 0.132 0.131 0.132 0.132 0.132 0.131 0.132 0.132 0.131 0.132 0.131 0.132 0.131 0.132 0.131 0.132 0.131 0.132 0.131 0.132 0.132 0.133 0.134 0.135 0.134 0.135 0.134 0.135 0.134 0.135 0.135 </td <td></td> <td> National Antipation and antipation antite antipation antipation antite antipation antipation antipat</td> <td>budin
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water</td> <td>Bel Barbares Bel Barbares B</td> <td>Lake to see
Lake to see
Lake</td> <td>Lang
Lang
Lang
Lang
Lang
Lang
Lang
Lang</td> <td>6 133
0 893
0 999
0 999
0</td>
<td>0.066
0.136
0.137
0.137
0.137
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.2420
0.2420
0.2420
0.2420
0.2420000000000</td> <td>1.8050
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.9000
0.9000
0.9000
0.9000
0.9000
0.9000
0.9000
0.9000
0.9000</td> <td>0
0
0.277
0.752
0.752
0.752
0.072
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td> <td>0
0,0272
0,0272
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,</td> <td></td> <td></td> <td></td> | |
National Antipation and antipation antite antipation antipation antite antipation antipation antipat
 | budin
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
 | Bel Barbares B | Lake to see
Lake | Lang
Lang
Lang
Lang
Lang
Lang
Lang
Lang
 | 6 133
0 893
0 999
0 | 0.066
0.136
0.137
0.137
0.137
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.2420
0.2420
0.2420
0.2420
0.2420000000000 |
1.8050
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.900
0.9000
0.9000
0.9000
0.9000
0.9000
0.9000
0.9000
0.9000
0.9000 | 0
0
0.277
0.752
0.752
0.752
0.072
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0,0272
0,0272
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0,005
0, |
 | | |
|

 | Barta basch Rahigen, Auszahraf hartar das das das das das das das das das das | 0.111 0.000 0.112 0.112 0.112 0.112 0.112 0.112 0.112 0.112 0.112 0.112 0.112 0.112 0.112 0.112 0.112 0.112 0.112 0.112 0.112 0.114

 | | Jobaina
Jobaina
Par
Par
Par
Par
Par
Par
Par
Par
Par
Pa
 |
Audion
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wardin
Wa
 | | Lake rase
Lake rase | Lans
Lans
Lans
Lans
Lans
Lans
Lans
Lans
 | 6 133
0 393
0 395
0 395
0 395
0 395
0 395
0 395
0 395
0 499
0 | 0.066
0.130
0.130
0.130
0.131
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242
0.242 | 1.8854
0.8892
0.8992
0.90
0.00
0.00
0.00
0.00
0.00
0.00
0 | 0
0
0.3721
0.7524
0.7524
0.7529
0
0
2.3529
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
 | 0
0,0272
0
0
0,0275
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | | | |
|

 | Nari saka Kasagawa Kasagaw | 0.111 0.006 0.112 0.113 0.114 0.115 0.112 0.122 0.123 0.124 0.125 0.126 0.127 0.128 0.121 0.126 0.127 0.128 0.129 0.120 0.121 0.121 0.121 0.121 0.121 0.121 0.121 0.121 0.121 0.121 0.121 0.121 0.121 0.121 0.122 0.121 0.122 0.121 0.121 0.122 0.121 0.121 0.122 0.123 0.124 0.125 0.125 0.121

 | | Noticitation Particitation Particitation <

 | budin
Warn
Warn
Warn
Warn
Warn
Warn
Warn
War
 | |
 | Lang
Lang
Lang
Lang
Lang
Lang
Lang
Lang | 6 133
6 837
6 937
6 | 0.066
0.132
0.132
0.13
0.15
0.15
0.12
0.22
0.22
0.22
0.22
0.22
0.22
0.23
0.23
 | 1.8854
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0
0
0,772
1,030
0
0
0
1,030
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 6
6
6
6
7
6
7
7
7
7
7
7
7
7
7
7
7
7
7
 | | | |
|

 | Nari savich Schöger, Auszahl sakar sören sören | 2) 2) 3) 2) 3) 3) 4) 4) 4) 4) 4) 4) 4) 4) 4) 4) 4) 4) 4) 4) 4) 4) 4) 4) 4) 4) 4) <

 | | Notifier Res

 | Wadan
Watan
Watan
Wadan
Wadan
Wadan
Madan
Madan
Madan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
Wadan
 | An Array Andre And | |
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
1400
140 |
 | 0.046
0.158
0.179
0.381
0.382
0.382
0.382
0.383
0.384
0.384
0.384
0.384
0.384
0.384
0.384
0.384
0.385
0.385
0.385
0.385
0.385
0.385 | ۱. Rester ۵. Comparison ۵. Comparison ۵. Comparison 1. State 0. Comparison 1. State 0. Comparison 0. Comparison | 0
0
0,772
1,072
1,072
1,072
1,072
0
0
1,772
0
1,778
0
0
0
1,2785
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 6
6
6
6
7
6
7
7
7
7
7
7
7
7
7
7
7
7
7
 | | | |
|

 | Normal and a second | 2).11 2).21 2).21 2).22 2).23 2).24 </td <td></td> <td>babener
bester
Per
Per
Per
Per
Per
Per
Per
Per
Per
P</td> <td>budin
Warn
Warn
Warn
Warn
Warn
Warn
Warn
War</td> <td>Net a fait of a second second</td> <td>Labor Sou
Labor Sou
Labor</td> <td>Laba Laba <tdlaba< td=""> <tdlaba< td=""></tdlaba<></tdlaba<></td> <td>6 133
6 213
6 214
6 4 244
6 4 244
6 4 244
6 4 244
6 4 247
6 4 247
7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5</td> <td>0.065
0.15
0.15
0.15
0.20
0.20
0.20
0.20
0.20
0.20
0.20
0.2</td> <td>1.8854
0.8892
0.8992
0.9
0.9
0.9
0.9
0.9
0.9
0.9
0.9
0.9
0.9</td> <td>0
0
0,772
1,603
0,772
1,603
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td> <td>6
6
6
6
7
6
7
7
7
7
7
7
7
7
7
7
7
7
7</td> <td></td> <td></td> <td></td>

 | | babener
bester
Per
Per
Per
Per
Per
Per
Per
Per
Per
P
 | budin
Warn
Warn
Warn
Warn
Warn
Warn
Warn
War
 | Net a fait of a second | Labor Sou
Labor
 | Laba Laba <tdlaba< td=""> <tdlaba< td=""></tdlaba<></tdlaba<> | 6 133
6 213
6 214
6 4 244
6 4 244
6 4 244
6 4 244
6 4 247
6 4 247
7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 | 0.065
0.15
0.15
0.15
0.20
0.20
0.20
0.20
0.20
0.20
0.20
0.2
 | 1.8854
0.8892
0.8992
0.9
0.9
0.9
0.9
0.9
0.9
0.9
0.9
0.9
0.9 | 0
0
0,772
1,603
0,772
1,603
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 6
6
6
6
7
6
7
7
7
7
7
7
7
7
7
7
7
7
7
 | | | |
|

 | Nari quori Absigne, Ausociard has been deline del nel del | 3.11 3.11 2.12 3.12 2.12 3.12 2.12 3.12 2.12 3.12 2.12 3.12 2.12 3.12 2.12 3.12 3.12 <td></td> <td></td> <td>Madan
Watan
Watan
Watan
Watan
Watan
Madan
Madan
Madan
Madan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan</td> <td></td> <td>Laber See See See See See See See See See S</td> <td>Lang
Lang
Lang
Lang
Lang
Lang
Lang
Lang</td> <td>6 133
6 838
6 989
6 989
6</td>
<td>0.046
0.15
0.15
0.24
0.24
0.24
0.24
0.24
0.24
0.24
0.24</td> <td> ۱. 8050 ۵. 8062 ۵. 8062</td> <td>0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td> <td>0 0</td> <td></td> <td>Additional areas badges 13.8.4.4
Additional areas badges 13.8.4.4
Additional areas badges 13.8.4.4</td> <td></td>
 | |

 | Madan
Watan
Watan
Watan
Watan
Watan
Madan
Madan
Madan
Madan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
Watan
 | | Laber See See See See See See See See See S | Lang
Lang
Lang
Lang
Lang
Lang
Lang
Lang
 | 6 133
6 838
6 989
6 | 0.046
0.15
0.15
0.24
0.24
0.24
0.24
0.24
0.24
0.24
0.24 | ۱. 8050 ۵. 8062 ۵. 8062
 | 0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0 | | Additional areas badges 13.8.4.4
Additional areas badges 13.8.4.4
Additional areas badges 13.8.4.4 | |
|

 | Net of solver a so | 3.11 3.11 2.12 3.12 2.12 3.12 2.12 3.12 2.12 3.12 2.12 3.12 2.12 3.12 2.12 3.12 3.12 <td></td> <td>Notesting Partial <</td> <td>budin
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water</td> <td>An An De Annue a de An</td>
<td>Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Lab</td> <td>Lang
Lang
Lang
Lang
Lang
Lang
Lang
Lang</td> <td>6 133
0 893
0 395
0 395
0 395
0 395
0 395
0 499
0 499
0</td> <td></td> <td>1. Kettic0.<</td> <td>0
0
0
0,772
1,000
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td> <td>0
0422
0422
0
0422
0
0
0425
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td> <td></td> <td>Additional and Marger 13 & 8 -
Additional and Marger 13 & 8 -</td> <td></td>
 | | Notesting Partial <
 |
budin
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
Water
 | An An De Annue a de An | Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Laberson
Lab | Lang
Lang
Lang
Lang
Lang
Lang
Lang
Lang
 | 6 133
0 893
0 395
0 395
0 395
0 395
0 395
0 499
0 | | 1. Kettic0.< | 0
0
0
0,772
1,000
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
 | 0
0422
0422
0
0422
0
0
0425
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | | Additional and Marger 13 & 8 -
Additional and Marger 13 & 8 - | |
|

 | Norse in characterization of a second sec |

 | | Notester Participation

 | badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
 | Bel at a set a se | Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
Labora
La | Lang
Lang
Lang
Lang
Lang
Lang
Lang
Lang
 |
 | 0.056
0.15
0.17
0.27
0.27
0.27
0.27
0.27
0.27
0.27
0.2 | 1.8850 0 0 0 0 0 0 1 2 0 0 1 0 1 0 1 0 1 0 <td>0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td> <td>0
04272
04272
04274
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
044747
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
0447474
0447474
0447474
0447474
0447474
0447474
0447474
0447474
0447474
0447474
0447474
0447474
0447474
0447474
0447474
0447474
044747474
0447474747</td> <td></td> <td></td> <td></td> | 0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 |
0
04272
04272
04274
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
044747
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
04474
0447474
0447474
0447474
0447474
0447474
0447474
0447474
0447474
0447474
0447474
0447474
0447474
0447474
0447474
0447474
0447474
044747474
0447474747 | | | |
|

 | Baria Baria Baraga Ang a |

 | | Notester Res
 | budin
Warn
Warn
Warn
Warn
Warn
Warn
Warn
War
 |
 | Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
Labera
La | Lange Lange <tdlange< td=""> <tdlange< t<="" td=""><td>6 133
6 837
6 987
6 989
6 989
6</td><td>8866
915
915
915
915
915
915
915
915</td><td> ۱. Restored ۵. Restored ۵. Restored ۵. Restored ۵. Restored 1. Restored </td><td>0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td><td>0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0</td><td></td><td>Additional antis Malage 7.3 & 4.
Additional antis Malage 7.3 & 4.
Additional antis Malage 7.3 & 4.</td><td></td></tdlange<></tdlange<> | 6 133
6 837
6 987
6 989
6 | 8866
915
915
915
915
915
915
915
915
 | ۱. Restored ۵. Restored ۵. Restored ۵. Restored ۵. Restored 1. Restored | 0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 |
 | Additional antis Malage 7.3 & 4.
Additional antis Malage 7.3 & 4.
Additional antis Malage 7.3 & 4. | |
|

 | Nario and San |

 | |
 |
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
badan
 | | | | | |
 | 6 132
0 887
0 887
0 488
0 | |
 | 0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | 0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 | | | |
|

 | Nari akai kata yan kata akai akai akai akai akai akai akai |

 | | Notesting Particular Par
 | Badan Watay Watay<
 | National And Source Sour |
 | | 6 132
6 838
6 938
6 939
6 | |
 | | 0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0
0 |
 | | Addited on Idea 184 1 | |

		Langarth Garden Village	1											
B-2 Si	te Hed	ge Creation	1											
Ĺ	Condense,	/ Show Columns Condense / Show Rows	j											
1	Ma	in Menu Instructions					Multipliers							
							Spatial quality							
		Proposed habitats		Habitat Habitat distinctiveness condition		Ecological connectivity	Strategic significance	Temporal multiplier	Herize units	Comments				
Baseline ref	New hedge number	Habitat type	Length km	Distinctiveness	Condition	Ecological connectivity	Strategic significance	Time to target condition/years	delivered	Assessor comments	Reviewer comments			
1		Native Species Rich Hedgerow - Associated with bank or ditch	1.46	High	Moderate	Medium	Area/compensation not in local strategy/ no local strategy	5	10.81	Green lane in west of site				
2		Native Hedgerow - Associated with bank or ditch	8.05	Medium	Poor	Low	Area/compensation not in local strategy/ no local strategy	1	20.82	Hedges within residential plots				
3		Native Hedgerow - Associated with bank or ditch	0.665	Medium	Poor	Low	Location ecologically desirable but not in local strategy	1	1.89	Hedges within residential plots				
4		Native Hedgerow - Associated with bank or ditch	3.756	Medium	Poor	Low	Area/compensation not in local strategy/ no local strategy	1	9.71	Hedges within residential plots				
5		Native Hedgerow	8.05	Low	Poor	Low	Within area formally identified in local strategy	1	17.87	Hedges within residential plots				
6		Native Hedgerow	0.665	Low	Poor	Low	Location ecologically desirable but not in local strategy	1	1.41	Hedges within residential plots				
7		Native Hedgerow	3.756	Low	Poor	Low	Area/compensation not in local strategy/ no local strategy	1	7.25	Hedges within residential plots				
8		Native Species Rich Hedgerow with trees - Associated with bank or ditch	1.75	High	Moderate	Medium	Area/compensation not in local strategy/ no local strategy	10	10.84	Hedge along northern side of A390				
9														
11														
12														
13			-											
14			-											
16														
17														
18			<u> </u>											
19			-											
20														
22														
23														
24														
25			1											

B-3 Site	Hedge Enhancement													
Condense / Show Columns Condense / Show Rows														
N	tain Menu Instructions			Post development/ post intervention	ASSESS									
	Baseline Habitats		Change in distincition	veness and condition	Length		-	Ecological	Strategic significance	Temporal multiplier	Difficulty Multipliers	Hedge units	Come	nents
Baseline ref	Baseline habitat	Proposed	Distinctiveness movement	Condition movement	юм	Distinctiveness	Condition	connectivity	Strategic significance	Time to target condition/years	Difficulty of enhancement Cotestory	delivered	Assessor comments	Reviewer comments
6	Native Hedgerow	Native Species Rich Hedgerow	Low - Medium	Lower Distinctiveness Habitat - Moderate	0.063	Medium	Moderate	Low	Area/compensation not in local strategy/	5	Medum	0.34		
7	Native Hedgerow	Native Species Rich Hedgerow	Low - Medium	Lower Distinctiveness Habitat - Moderate	0.058	Medium	Moderate	Low	Area/compensation not in local strategy/ no local strategy	5	Medium	0.41		
9	Native Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or dtch	Medium - High	Lower Distinctiveness Habitat - Moderate	0.235	High	Moderate	Medium	Area/compensation not in local strategy/ no local strategy	5	Medium	2.19		
11	Native Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or ditch	Medium - High	Lower Distinctiveness Habitat - Moderate	0.109	High	Moderate	Medium	Area/compensation not in local strategy no local strategy	5	Medium	1.02		
12	Native Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or disch	Medium - High	Lower Distinctiveness Habitat - Moderate	0.05	High	Moderate	Medium	no local strategy Area/compensation not in local strategy/	5	Medium	0.56		
15	Native Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or ditch	Medium - High	Lower Distinctiveness Habitat - Moderate	0.2	High	Moderate	Medium	no local strategy Location ecologically desirable but not in		Medum	2.05		
15	Native Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or dtch	Medium - High	Lower Distinctiveness Habitat - Moderate	0.099	High	Moderate	Medium	Area/compensation not in local strategy/	5	Medum	1.12		
17	Native Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or ditch	Medium - High	Lower Distinctiveness Habitat - Good	0.088	High	Good	Medium	Area/compensation not in local strategy/ no local strategy	30	Medium	1.02		
18	Native Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or ditch	Medium - High	Lower Distinctiveness Habitat - Moderate	0.09	High	Moderate	Medium	Area/compensation not in local strategy/ no local strategy	5	Medium	0.94		
19	Native Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or ditch	Medium - High	Lower Distinctiveness Habitat - Moderate	0.08	High	Moderate	Medium	Area/compensation not in local strategy no local strategy Area/compensation not in local strategy	5	Medium	0.75		
21	Native Hedgerow - Associated with bank or disch	Native Species Rich Hedgerow - Associated with bank or dich	Medium - High	Lower Distinctiveness Habitat - Moderate	0.088	High	Moderate	Medium	no local strategy Area/compensation not in local strategy/	5	Medium	1.16		
27	Native Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or dtch	Medium - High	Lower Distinctiveness Habitat - Good	0.066	High	Good	Medium	no local strategy Area/compensation not in local strategy/	10	Medum	0.92		
30	Native Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or ditch	Medium - High	Lower Distinctiveness Habitat - Good	0.084	High	Good	Medium	Area/compensation not in local strategy/ no local strategy	30	Medium	1.17		
31	Native Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or dtch	Medium - High	Lower Distinctiveness Habitat - Good	0.058	High	Good	Medium	Area/compensation not in local strategy/ no local strategy	30	Medium	0.81		
38	Native Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or ditch	Medium - High	Lower Distinctiveness Habitat - Good	0.115	High	Good	Medium	Area/compensation not in local strategy no local strategy Area/compensation not in local strategy	30	Medium	1.34		
40	Native Hedgerow - Associated with Bank or disch	Native Species Ach Hedgerow - Associated with bank of dich	Medium - High	Lower Distinctiveness Habitat - Moderate	0.03	High	Moderate	Medium	no local strateev Area/compensation not in local strategy/	•	Medum	0.20		
42	Native Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or dtch	Medium - High	Lower Distinctiveness Habitat - Good	0.16	Hgh	Good	Medium	no local strategy Area/compensation not in local strategy/ no local strategy	30	Medium	1.86		
45	Native Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or ditch	Medium - High	Lower Distinctiveness Habitat - Moderate	0.205	High	Moderate	Medium	Location ecologically desirable but not in local strateev	5	Medium	2.54		
45	Native Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or ditch	Medium - High	Lower Distinctiveness Habitat - Moderate	0.121	High	Moderate	Medium	Area/compensation not in local strategy/ no local strategy	5	Medium	1.13		
47	Native Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or ditch	Medium - High	Lower Distinctiveness Habitat - Good	0.047	High	Good	Medium	no.local strateev Location ecologically desirable but not in	10	Medium	0.55		
50	Native Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or ditch	Medium - High	Lower Distinctiveness Habitat - Moderate	0.164	High	Moderate	Medium	local strategy Location ecologically desirable but not in	5	Medum	2.04		
51	Native Species Rich Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or dtch	High - High	Moderate - Good	0.149	High	Good	Medium	Area/compensation not in local strategy/ no local strategy	30	Medium	2.43		
52	Native Species Rich Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or dtch	High - High	Moderate - Good	0.089	Hgh	Good	Medium	Area/compensation not in local strategy/ no local strategy	10	Medum	1.45		
54	Native Species Rich Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or ditch	High - High	Moderate - Good	0.039	High	Good	Medium	Area/compensation not in local strategy/ no local strategy	10	Medum	0.64		
55	native Species Rich Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or ditch	High - High	Moderate - Good	0.121	High	Good	Medium	no local strategy/ Area/compensation not in local strategy/	10	Medum	1.97		
57	Native Species Rich Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or dtch	High - High	Moderate - Good	0.084	High	Good	Medium	no local strategy Area/compensation not in local strategy/	10	Medum	1.37		
58	Native Species Rich Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or dtch	High - High	Moderate - Good	0.12	High	Good	Medium	Area/compensation not in local strategy/ no local strategy	10	Medum	1.96		
59	Native Species Rich Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or ditch	High - High	Moderate - Good	0.08	High	Good	Medium	Area/compensation not in local strategy/ no local strategy	30	Medum	1.30		
62	Native Hedgerow with trees - Associated with bank or dtch	Native Species Rich Hedgerow - Associated with bank or dtch	Medium - High	Lower Distinctiveness Habitat - Moderate	0.188	High	Moderate	Medium	Area/compensation not in local strategy/ no local strategy	5	Medum	2.48		
64	Native Hedgerow with trees - Associated with bank or dtch	Native Species Rich Hedgerow - Associated with bank or ditch	Medium - High	Lower Distinctiveness Habitat - Moderate	0.066	High	Moderate	Medium	no local strategy Location ecologically desirable but Proving	5	Medum	0.97		
65	Narve Hedgerow with trees - Associated with bank or dtch Narve Hedgerow with trees - Associated with bank or dw-h	Native Species forn Hedgerow - Associated with bank or ditch	Medium - High Medium - High	Lower Distinctiveness Habitat - Moderate	0.138	High	Moderate Moderate	Medium	local strategy Area/compensation not in local strategy/	5	Medum	1.72		
69	Native Hedgerow with trees - Associated with bank or disch	Native Species Rich Hedgerow - Associated with bank or dtch	Medium - High	Lower Distinctiveness Habitat - Moderate	0.15	High	Moderate	Medium	no local strateev Area/compensation not in local strategy/ no local strateev	5	Medum	1.98		
71	Native Hedgerow with trees - Associated with bank or disch	Native Species Rich Hedgerow - Associated with bank or dtch	Medium - High	Lower Distinctiveness Habitat - Moderate	0.079	High	Moderate	Medium	Area/compensation not in local strategy/ no local strategy	s	Medum	0.89		
123	Native Species Rich Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or ditch	High - High	Moderate - Good	0.112	High	Good	Medium	Area/compensation not in local strategy/ no local strategy	30	Medum	1.83		
125	Native Species Rich Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or ditch	High - High	Moderate - Good	0.116	High	Good	Medium	 wea/compensation not in local strategy/ no local strategy Area/compensation not in local strategy 	10	Medum	1.89		
130	Native Species Rich Hedgerow - Associated with bank or disch	Native species rech Hedgerow - Associated with bank or ditch Native Species Rich Hedgerow - Associated with bank or dit ^{-h}	High - High High - High	Moderate - Good	0.112	High	Good	Medium	no local strateev Location ecologically desirable but not in	10	Medum	2.02		
131	Native Species Rich Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or ditch	High - High	Moderate - Good	0.289	Hgh	Good	Medium	local strategy Area/compensation not in local strategy/ molocal strategy/	10	Medum	4.71		
134	Native Species Rich Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or ditch	High - High	Moderate - Good	0.1	High	Good	Medium	Area/compensation not in local strategy/ no local strategy/	10	Medum	1.63		
139	Native Species Rich Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or ditch	High - High	Moderate - Good	0.066	High	Good	Medium	Area/compensation not in local strategy/ no local strategy	10	Medium	1.08		
150	Native Species Rich Hedgerow with trees - Associated with bank or disch Native Spacies Rich Liedenma with trees - Americated with bank or	Native Species Rich Hedgerow with trees - Associated with bank or ditch	High - High	Moderate - Good	0.1	High	Good	Medium	Area/compensation not in local strategy/ no local strategy	30	Medium	1.54		
155	dtch	Native Species Rich Hedgerow with trees - Associated with bank or ditch	High - High	Moderate - Good	0.121	High	Good	Medium	local strateev Area/compensation not in local strategy/	20	Medium	2.05		
190	Native Species Rich Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or ditch	High - High	Poor - Moderate Poor - Moderate	0.045	High	Moderate	Medium	no local strategy Area/compensation not in local strategy/	5	Medum	0.46		
191	Native Species Rich Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or dtch	High - High	Poor - Moderate	0.117	High	Moderate	Medium	Area/compensation not in local strategy/	5	Medum	1.21		
192	Native Species Rich Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or ditch	High - High	Poor - Moderate	0.152	High	Moderate	Medium	Area/compensation not in local strategy/ no local strategy	5	Medium	1.57		
193	Native Species Rich Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or dtch	High - High	Poor - Moderate	0.15	High	Moderate	Medium	Area/compensation not in local strategy/ no local strategy	5	Medum	1.55		
195	Native Species Rich Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or ditch	High - High	Moderate - Good	0.179	High	Good	Medium	Area/compensation not in local strategy/ no local strategy	30	Medium	2.92		
201	Native Species Rich Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or ditch	High - High	Moderate - Good	0.181	High	Good	Medium	no local strategy Area/compensation not in local strategy/	30	Medium	2.95		
204	Native Species Rich Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or dtch	High - High	Moderate - Good	0.053	High	Good	Medium	no local strateev Location ecologically desirable but not in	10	Medum	0.95		
209	Native Species Rich Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or dtch	High - High	Moderate - Good	0.032	High	Good	Medium	Location ecologically desirable but not in local strategy	10	Medium	0.57		
214	Native Species Rich Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or ditch	High - High	Moderate - Good	0.101	High	Good	Medium	Location ecologically desirable but not in local strategy	30	Medium	1.81		
215	Native Species Rich Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or dtch	High - High	Moderate - Good	0.154	High	Good	Medium	Location ecologically desirable but not in local strateev	30	Medium	2.76		
224	disch Native Species Rich Hedgerow with trees - Associated with bank or	Native Species Rich Hedgerow with trees - Associated with bank or ditch	High - High	Moderate - Good	0.1	High	Good	Medium	no local strategy Location ecologically desirable but not in	20	Medum	1.54		
228	dtch Native Species Rich Hedgerow with trees - Associated with bank or	Native Species Rich Hedgerow with trees - Associated with bank or ditch	High - High	Moderate - Good	0.177	High	Good	Medium	local strateev Location ecologically desirable but not in	20	Medum	2.00		
229	disch Native Species Rich Hedgerow with trees - Associated with bark or disch	Native Species Rich Hedgerow with trees - Associated with bank or ditch	High - High	Moderate - Good	0.185	High	Good	Medium	Local strateey Location ecologically desirable but not in local strateey	20	Medium	3.13		
230	Native Species Rich Hedgerow with trees - Associated with bank or dtch	Native Species Rich Hedgerow with trees - Associated with bank or ditch	High - High	Moderate - Good	0.229	High	Good	Medium	Location ecologically desirable but not in local strategy	20	Medium	2.87		
234	Native Species Rich Hedgerow with trees - Associated with bank or disch Native Spacies Rich Liedenma with trees - Americated with bank or	Native Species Rich Hedgerow with trees - Associated with bank or ditch	High - High	Moderate - Good	0.101	High	Good	Medium	Area/compensation not in local strategy/ no local strategy	30	Medium	1.55		
235	dtch Native Species Rich Hedgerow with trees - Associated with bank or	Native Species Rich Hedgerow with trees - Associated with bank or ditch	High - High	Moderate - Good	0.173	High	Good	Medium	no local strateev Area/compensation not in local strategy/	20	Medium	2.66		
230	dtch Native Species Rich Hedeerow - Associated with bank or dtch	Native Species Rich Hedgenow with trees - Associated with bank or ditch Native Species Rich Hedgerow - Associated with bank or ditch	High - High	Moderate - Good	0.033	High	Good	Medium	no local strategy Area/compensation not in local strategy/	20	Medum	0.51		
245	Native Species Rich Hedgerow - Associated with bank or ditch	Native Species Rich Hedgerow - Associated with bank or dtch	High - High	Moderate - Good	0.038	High	Good	Medium	no local strategy Area/compensation not in local strategy/ no local strategy	10	Medium	0.62		
									1					
								_						
								_						
									<u> </u>					
									-					
	<u></u>							_						
								_						
		· · · · · · · · · · · · · · · · · · ·												
								_						
								_						
								_						
								_						
							<u> </u>		_					

D-2 Off Site Habitat Creation			-																	
Candence / Thise Columns Con	năm (ə / Urina	- Rines	1																	
Main Menu	waruction	4																		
							Post developmenty post intervention hobits Boological cannectivity		Xi Megis oʻge	ificance	Name	Temporal	ultiplier	Difficulty	nultiples Difference	Tpatal fits waitiplier		Indefant weiter	Cana	unds
Proposed kabitat	Area ha	ChillingSymposi	haire	Condition	haare	Ecological connectivity	Connectivity	Connectivity multiplier	Strategic significance	Musiegic ognificance	position multiplier	Time to target candition/years	Time to taget multiplier	creation category	creation multiplier	Spatial 4ck category	Spatial risk multiplier	delivered	Assessor comments	Reviewer comments
			-								_									
	-		-																	
	-		-																	
			-																	
	-																			
	<u> </u>																			
	1																			
	-																			
	1																			
	-																			
			-																	
			-																	
			-	_																
								_												
			-																	
			-																	
			-																	
			-								_									
			-																	
	1																			
	<u> </u>																			
	1																			
	+																			
	-																			
	1																			
	1																			
	1																			
	+																			
	1																			
	<u> </u>																			
	+																			
	-																			
	-																			
	1																			
	-																			
	+																			
	-																			